# Care to College An Exploratory Study on Care-Experienced Students' Educational Journeys

Merike Darmody





empowering people in care

# CARE TO COLLEGE: AN EXPLORATORY STUDY ON CARE-EXPERIENCED STUDENTS' EDUCATIONAL JOURNEYS

Merike Darmody

May 2025

# JOINTLY-PUBLISHED REPORT

Economic and Social Research Institute EPIC, Empowering People in Care

Available to download from www.esri.ie and www.epiconline.ie

https://doi.org/10.26504/jr13

© 2025 Copyright is held jointly by the Economic and Social Research Institute and EPIC, Empowering People in Care. EPIC, Empowering People in Care, 7 Red Cow Lane, Smithfield, Dublin 7. The Economic and Social Research Institute, Whitaker Square, Sir John Rogerson's Quay, Dublin 2.



This Open Access work is licensed under a Creative Commons Attribution 4.0 International License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly credited.

# **ABOUT THE ESRI**

The Economic and Social Research Institute (ESRI) advances evidence-based policymaking that supports economic sustainability and social progress in Ireland. ESRI researchers apply the highest standards of academic excellence to challenges facing policymakers, focusing on ten areas of critical importance to 21st Century Ireland.

The Institute was founded in 1960 by a group of senior civil servants led by Dr T.K. Whitaker, who identified the need for independent and in-depth research analysis. Since then, the Institute has remained committed to independent research and its work is free of any expressed ideology or political position. The Institute publishes all research reaching the appropriate academic standard, irrespective of its findings or who funds the research.

The ESRI is a company limited by guarantee, answerable to its members and governed by a Council, comprising up to 14 representatives drawn from a cross-section of ESRI members from academia, civil services, state agencies, businesses and civil society. Funding for the ESRI comes from research programmes supported by government departments and agencies, public bodies, competitive research programmes, membership fees, and an annual grant-in-aid from the Department of Public Expenditure NDP Delivery and Reform.

Further information is available at www.esri.ie.

# **ABOUT EMPOWERING PEOPLE IN CARE (EPIC)**

Empowering People in Care (EPIC) is a rights-based organisation advocating with and for children in care, young care leavers, and anyone with experience of the care system up to the age of 26. EPIC operates the only National Advocacy Service for care-experienced youth, which works to ensure that their voices are represented and considered in decisions that affect their lives.

Research and policy at EPIC are informed by the data generated by the Advocacy Service caseload and participative research projects – both of which provide an evidence base to support national decision-making on alternative care. EPIC also delivers a programme of youth engagement and participation that provides spaces for connection, fun and growth, as well as opportunities for action and learning by working on youth-led campaigns and projects.

Further information is available at www.epiconline.ie.

# ABOUT TU DUBLIN AND THE PATH 3 PROGRAMME

The Programme for Access to Higher Education (PATH 3) is a partnership between TU Dublin and community organisations to support access to higher education for under-represented groups.

The partnership brings together different sectors, including statutory education providers and community organisations to share knowledge and devise strategies that support equity of access to higher education. It particularly focuses on strategies to support people from communities and groups where there has not been a strong tradition of participation in higher education.

PATH 3 is funded by the Higher Education Authority under the National Plan for Equity of Access to Higher Education. The project forms part of TU Dublin's commitment to extending community partnerships, engagement activities and creating pathways for all.

# **THE AUTHORS**

Dr Merike Darmody is a Research Officer at the Economic and Social Research Institute (ESRI) and an Adjunct Assistant Professor at Trinity College Dublin (TCD).

# ACKNOWLEDGEMENTS

This research was commissioned by EPIC (Empowering People in Care). EPIC is a rights-based organisation advocating with and for children in care, young care leavers, and anyone with experience of the care system up to the age of 26. Funding for the report is from the Programme for Access to Higher Education (PATH 3) partnership between TU Dublin and community organisations to support access to higher education for under-represented groups. PATH 3 is funded by the Higher Education Authority under the National Plan for Equity of Access to Higher Education. The author is very grateful to Conor Stitt and John Murphy from EPIC for their support throughout the study. Particular thanks go to the care-experienced young people participating in this study. The author is also grateful to the key stakeholders participating in the consultation event, whose expertise and views greatly assisted in gaining better understanding of challenges care-experienced young people face once leaving school. Thanks are due to the editor, Helen Russell, and the three reviewers for their very helpful comments.

This report has been accepted for publication by the Institute, which does not itself take institutional policy positions. All ESRI Research Series reports are peer reviewed prior to publication. The author(s) are solely responsible for the content and the views expressed.

# FOREWORD

This research was made possible by a grant from TU Dublin Access and Outreach under PATH 3 funding, which supports access to higher education for underrepresented groups as part of the Programme for Access to Higher Education under the National Access Plan.

The ESRI and Dr Merike Darmody have produced an important study. In her exploration of the subject, we can see clearly the value of placing young people's experiences at the centre of research about them. The voices of the young people give us a rich and nuanced insight into both the academic and social challenges of navigating school, the transition to post-school education and beyond. It shows us that to address challenges in relation to post-school education, we need to understand how attitudes to education start in the early years and are influenced by care placement changes and school changes, learning difficulties, support needs around schoolwork, and mental health issues.

EPIC's commissioning of this study is a response to the direction of our Youth Council, to highlight barriers faced by care-experienced young people in the education system. A key recommendation emerging from our work is the need for greater awareness and understanding of the school lives and educational journeys of children in care and young adults with experience of the care system.

This is an important piece of research for deepening our understanding in this regard. Any impact that it will have will be due in no small part to the commitment of the author and the ESRI to excellence in the production of independent, high-quality research.

This research is relevant to schools, education stakeholders, foster carers, social workers, children's rights organisations and policymakers to enhance our understanding of school life and education through the lens of children in care.

EPIC sees this research as an important part of our broader programme of work in this area with the aim of creating educational environments where care-experienced children and young people can thrive.

We hope our ongoing work with others will create a synergy between the different sectors and stakeholders and ultimately lead to practical supports for children and young people as they move through care and education.

Wayne Stanley,

CEO, EPIC, Empowering People in Care

# **Table of contents**

EXECU	TIVE	SUMMA	ARY	IX			
	Вас	kground	l to the study	ix			
	School experiences of care-experienced children and their post-school transition to third-level education – results of the study						
	Dire	ections f	or policy development and further research	xii			
СНАРТ	ER 1:	INTRO	DUCTION	1			
	1.1	Background to the study					
	1.2	Care-experienced young people and education					
		1.2.1	Academic sphere and care experience	2			
		1.2.2	Social sphere and care experience	3			
		1.2.3	Leaving care	5			
		1.2.4	Aspirations, expectations and entry into third-level education	6			
	1.3	Care-experienced young people in Ireland					
	1.4	Aim of the research and research questions1					
	1.5	Methodology					
		1.5.1	Study design	12			
		1.5.2	Sample selection	13			
		1.5.3	Ethics and confidentiality	15			
		1.5.4	Limitations of the study	15			
	1.6	Outline	e of the report	16			
			TION OF CARE-EXPERIENCED YOUNG PEOPLE INTO HIGHER EDUCATION: ID BEST PRACTICES	17			
			Introduction				
	2.2	International interventions and practices supporting the education and wellbeing of care-experienced young people					
		2.2.1	Educational support for care-experienced young people in schools	18			
		2.2.2	Support for young people leaving care	19			
		2.2.3	Higher and further education	20			
		2.2.4	Support for care-experienced youth accessing higher education in Ireland	22			
	2.3	Conclu	sions	23			
СНАРТ	ER 3:	RESULT	S OF THE STUDY: CARE PLACEMENTS AND SCHOOL EXPERIENCES	25			
	3.1 Introduction						
	3.2	Coping	with changes – care placements and schools	25			
	3.3	Lack of	f communication between schools and social workers	29			
	3.4	Coping	with mental health issues	30			
	3.5	Care p	lanning, relationships with foster parents and care professionals	31			

3.6	Conclusions	
CHAPTER 4:	SOCIAL EXPERIENCES AT SCHOOL	
4.1	Introduction	
4.2	Identification of care-experienced children and young people	
4.3	Relationships with teachers	
4.4	Relationships with foster parents and care professionals	42
4.5	Relationships with peers	44
	Support in the social sphere	
4.7	Conclusion	
CHAPTER 5:	ACADEMIC EXPERIENCES AT SCHOOL	50
5.1	Introduction	50
5.2	Attitudes towards school and attendance	50
5.3	Expectations, programmes and academic outcomes	52
	5.3.1 Expectations	52
	5.3.2 Programmes	53
	5.3.3 Academic outcomes	54
	5.3.4 Additional help with learning	57
	5.3.5 Outcomes of state exams	58
5.4	Conclusions	61
CHAPTER 6:	DECISION-MAKING AND ADVICE ABOUT POST-SCHOOL PATHWAYS	63
	Introduction	
6.2	Planning for post-school pathways	63
	6.2.1 Teachers and other staff members	64
	6.2.2 Support from social workers	68
	6.2.3 Support from foster parents and care staff	68
6.3	Conclusion	
	TRANSITION FROM CARE TO AFTERCARE AND THIRD-LEVEL EDUCATION	
	Transition into aftercare	
	From care to college	
7.5	7.3.1 Considering post-school options	
	7.3.2 Challenges related to transition to college/higher education	79
	7.3.2.1 Stigma attached to care experience	_
	7.3.2.1     Stigma attached to care experience       7.3.2.2     Social and academic sphere	
	7.3.2.3 Mental health issues and support with emotional wellbeing	
	7.3.2.4 Financial issues	
	7.3.2.5 Accommodation	

7.4	7.4 The need for flexibility in supporting young care leavers					
7.5	Conclu	sion	.90			
CHAPTER 8:	CONCL	USIONS AND POLICY IMPLICATIONS	.92			
8.1	Introdu	iction	.92			
8.2	Main fi	Vain findings9				
	8.2.1	Care-experienced children and young people at school	94			
	8.2.2	Leaving school and leaving care	95			
	8.2.3	Challenges faced by care-experienced young people in college/higher educa	tion 97			
8.3	Direction	ons for policy development and further research	.98			
REFERENCE	S		L <b>01</b>			
APPENDIX I	: INTERV	IEW SCHEDULE1	14			
APPENDIX I	I: INTER	/IEW CONSENT FORM	17			

# LIST OF TABLES

# **ABBREVIATIONS**

BPD	Borderline Personality Disorder
CAO	Central Applications Office
CES	Care-Experienced Students
CPD	Continuous Professional Development
DARE	Disability Access Route to Education
EET	Education, Employment or Training
EPIC	Empowering People in Care
GCSE	General Certificate of Secondary Education
HE	Higher Education
HEA	Higher Education Authority
HEAR	Higher Education Access Route to Education
ITE	Initial Teacher Education
LCA	Leaving Certificate Applied
LCVP	Leaving Certificate Vocational Programme
PLC	Post Leaving Certificate
QQI	Quality and Qualifications Ireland
SCP	School Completion Programme
SUSI	The Student Grant Scheme

#### **BACKGROUND TO THE STUDY**

Some children and young people do not reside with their birth families but are, instead, being cared for by the state. Systematic reviews have indicated that while not a homogenous group regarding their personal circumstances, many care-experienced children and young people tend to face some similar challenges when growing up and moving through the education system (O'Higgins et al., 2015; Evans et al., 2017; Haddow et al., 2021; Townsend et al., 2020). On average, care-experienced young people tend to have lower levels of educational attainment in comparison to their peers who have not been in care (Cotton et al., 2014), although their attainment levels can increase later on (Harrison, 2017). The implications of their school and care experiences are likely to extend beyond second-level schooling since the educational outcomes of children and young people in care are strongly linked to their post-school outcomes across a number of fields, such as employment, entry into third-level education and training (Harrison, 2020; Herbers et al., 2013; McNamara et al., 2017). This makes exploring their experiences in the education system an important focus for research and policymaking.

The aim of this small-scale exploratory study is to gain a better understanding of the factors that may influence post-school decision-making of care-experienced young people regarding entry into third-level education. With this in mind, the study starts with exploring retrospective views of a small group of careexperienced young people about their school experiences, as such experiences are likely to influence young people's school engagement, aspirations and decisionmaking about their post-school pathways. The study also investigates their transition from school and the care system into post-school education and independent living, focusing on the challenges they encountered. It considers the experiences of students in third-level education, but also of those young people who had interrupted third-level education experience, i.e. who had dropped out. Drawing on the insights of a small group of care-experienced young people and a group of key stakeholders representing various organisations, this study seeks to inform policy development in supporting post-school transition of careexperienced students and identifies areas for future research. Reflections on young people's own perspectives highlight various challenges they face in the care system and at school, particularly valuable for policymaking in this area. Furthermore, the study will provide insights for school practice and support services for careexperienced young people.

Although small in scale, the study contributes to the international and national literature by highlighting the factors that shape the decision-making processes of care-experienced young people regarding their post-school pathways and their experiences in third-level education. The study addresses the following research questions:

- 1. How does care experience impact on young people's experiences in school?
- 2. How does care experience impact on young people's interaction with friends, teachers and other school staff?
- 3. What are the main factors that shape the decision-making process of care-experienced young people regarding their post-school pathways?
- 4. What are the main challenges care-experienced young people face after leaving school and entering third-level education?

To answer these questions, the study draws on data from focus groups with careexperienced young people who have progressed to third-level education, and those whose participation in third-level education has been interrupted. The findings also incorporate insights from a group of key stakeholders who attended a consultation event.

# SCHOOL EXPERIENCES OF CARE-EXPERIENCED CHILDREN AND THEIR POST-SCHOOL TRANSITION TO THIRD-LEVEL EDUCATION – RESULTS OF THE STUDY

International and Irish studies have shown that many care-experienced young people encounter multiple challenges when moving through the education system (O'Higgins et al., 2015; Bakketeig et al., 2020; Darmody et al., 2013; Brady and Gilligan, 2019). While some succeed, despite the adversity they have encountered in their lives and fare well in terms of their educational outcomes and accessing third-level education, others tend to be more affected by structural (e.g. education and care system) and personal factors (e.g. care experience, mental health issues, learning difficulties) and do less well at school and beyond (Haddow et al., 2021, Hammond et al., 2020).

In line with previous studies from Ireland and elsewhere, the findings of this study indicate that young people with multiple care and school placements tend to be more likely to find their experiences at school more challenging compared to their peers who are in more stable and nurturing placements and have had fewer school changes. Young people's accounts have shown that high mobility in terms of care placements and school changes translates into interrupted learning and social ties, echoing findings of previous research from Ireland and abroad (Darmody et al., 2013; Schwartz-Mette et al., 2020). Furthermore, the findings of this study showed that frequent school engagement. In line with international research (Hammond et al., 2020), those young people in the study who experienced mental health issues and learning difficulties tended to have additional challenges in academic and social spheres.

In line with broader research on children who have experienced adversities (Frederick et al., 2023), this study highlights the importance of having a positive

and supportive relationship with at least one key adult, whether at school or in the care system. Teachers, foster families and social workers can play a key role in facilitating school engagement of care-experienced young people, support their learning and personal development, and help build up their aspirations. However, less supportive relationships at school and low expectations regarding educational progression of care-experienced young people and perceiving them as 'troublemakers' can have an adverse impact on these students. Early intervention that starts when the child enters the education system and support are key, as young people develop their attitudes, aspirations and dispositions towards education early on in their educational career (Greenaway and Treton, 2017; Gale et al., 2010). This study also showed that foster parents and social workers can be an important support in supporting education of care-experienced young people and enhancing their aspirations.

Previous research from Ireland and elsewhere have highlighted the challenges associated with leaving secondary school and choosing a post-school pathway (McCoy et al., 2014; The Smith Family, 2022). This juncture may be particularly challenging for care-experienced young people as they lack supporting family structure (Storø, 2018). Many young people in this study struggled with the 'double transition' – leaving school and foster/residential care. Preparing for state exams while preparing to move into post-school education and independent living caused considerable stress for some young people, sometimes adversely affecting their wellbeing. Some young people were critical about the guidance they received at school regarding their post-school course options or had their third-level education options criticised by the guidance counsellor. Yet others did not want to enter third-level education straight after secondary school or wanted to do this later. Having their aftercare support tied to attending full-time accredited education or training course was criticised by several young people in this study. Overall, the results of the study show that care-experienced young people wish to exercise control over their post-school pathways; however, the current structure of support available is rather rigid and embedded in the assessment of need and direct link between support and post-school education/training. While for many young people, continuing their education at third level was seen as a natural step, others felt that they would have liked to choose this pathway on their own terms, rather than being directed to take this step in order to access aftercare allowance and supports.

Various studies have highlighted difficulties young people experience when moving into third-level education, largely centring around finances, accommodation, course regret and preparedness for study in post-school education (Hauschildt et al., 2024; Fenech, 2025; McCoy et al., 2014). The situation is likely to be more challenging for care-experienced young people who often lack the supportive family structure, as highlighted by international literature (Storø, 2018; Paulsen and Berg, 2016). While all care-experienced students in this study had accessed third-level education, many experienced various challenges. These ranged from practical arrangements involving moving to a new location on their own, difficulties around finances and accommodation, getting used to different teaching and learning approaches to not being certain about their interest in the course they had enrolled in. While some navigated these challenges with the help of their aftercare workers, others had limited support from their aftercare worker and felt that they had to 'figure things out the hard way', to quote one young person. In total, 16 care-experienced young people participated in this study. Although 12 young people in this study were attending their course or had completed it, four young people had dropped out, highlighting factors such as not wanting to enter into third level straight after secondary school, having mental health challenges and learning difficulties as some of the challenges that contributed to their dropping out of college.

#### DIRECTIONS FOR POLICY DEVELOPMENT AND FURTHER RESEARCH

This study demonstrates that the educational experiences of young people in care are often influenced by structural and personal factors. Of particular importance in this respect are care placement quality and consistency and school stability, in line with existing research (Atkinson and Hyde, 2019; Ellis and Johnston, 2022). Furthermore, support from a dedicated person emerged as an important factor in helping care-experienced young people to negotiate their care and school placements, helping to develop their aspirations and planning for their next steps after finishing school.

At school level, teachers in conjunction with social workers need to prioritise the needs of care-experienced young people to support their school engagement and ensure they have the best possible opportunity to continue their education after finishing school. Teachers and other staff members need to be sufficiently informed about the challenges faced by many care-experienced children and young people, something that could be addressed by initial teacher education and continuous professional development. Furthermore a whole school approach enables schools to create a supportive environment where these young people can thrive. The School Completion Programme (SCP) is one way to provide additional support for these students. However, SCP is only available in DEIS schools and thus not accessible to students attending non-DEIS schools. Closer collaboration is also needed between schools and social workers in supporting care-experienced children and young people. As social workers are sometimes the only constant in the lives of care-experienced children and young people when they are still at school, more attention should be directed to retention of the individuals and funding of the sector, to prevent a turnover of social workers. Future research could explore how different stakeholders (schools, social workers and other agencies) interact in providing a wrap-around support to young people as they move through the care and education system. Furthermore, early intervention is important. Targeted supports (including consistent key contact, access to mental health/trauma support) need to be available to young people early on and be provided consistently throughout their primary and secondary schooling.

The results of the study show that entry into third-level education is associated with a number of challenges for care-experienced young people. This suggests that care-experienced young people who have accessed third-level education may need additional support. The accounts of young people have revealed that more guidance is required at school level, as this group of young people are particularly dependent on this type of support. Care-experienced young people may also benefit from other types of tailored support prior to leaving school and entering third-level education, something that could be explored in future research.

Financial support for young people, including care leavers, accessing third-level education has not kept pace with the cost of living in Ireland. There have been recent calls for such support to be increased for all students, to reflect the actual living needs. A Standardised National Aftercare Allowance was introduced by Tusla in 2015.<sup>1</sup> This allowance can be up to €300 per week. As this allowance has not been increased thereafter, it represents a cut in support to the care-experienced young people in real terms. Further research is needed to establish the financial needs of care-experienced young people attending full-time education and training. Considering the current housing crisis in Ireland, public debate has focused on lack of accommodation for students. In some cases, on-campus housing is only available during term time. This is likely to pose particular challenges to care-experienced young people who would need accommodation for the full calendar year and for the duration for their studies. With this in mind, careexperienced students should be prioritised in the allocation of on-campus accommodation. Where year-long accommodation is not possible, third-level education providers could consider linking the student with a local family who could host them during the holidays, an approach utilised by some of the higher education institutions in the United Kingdom.

If eligible, Irish young people in third-level education or training can avail of an aftercare service by Tusla up until the age of 21. This can be extended to the age of 23 if a young person is still in full-time accredited education or training. Some young people highlighted limited support received when in college, despite having been allocated an aftercare worker. At present, Tusla employs 71 aftercare workers; this suggests that these individuals may experience a notable workload, as they generally monitor the situation of several young people. Future research is needed to explore the workload of aftercare workers and associated challenges.

Limited support at third level could also be explained by the fact that some young people did not want to be identified as 'care-experienced', thus making it potentially more difficult for third-level institutions to put in place targeted supports. In order to provide maximum support for care-experienced students, including a question on care experience in the CAO form could be explored, while

<sup>&</sup>lt;sup>1</sup> This payment is available for young people who have been in care for 12 months on their 16th birthday or have been in care for 12 months in a row before their 18th birthday. Eligible young people may also be entitled to grants as a care leaver from Care Leavers Ireland.

clearly outlining incentives associated with doing so. This would also enable linking up care leavers with a named contact at third-level institutions.

#### **CHAPTER 1**

#### Introduction

#### 1.1 BACKGROUND TO THE STUDY

A wealth of research has highlighted the importance of educational attainment to the future life chances of young people (Gillborn and Mirza, 2000; Jones, 2002). There is now a growing body of international and Irish studies that have focused on educational experiences of children and young people who have lived in alternative care settings – such as foster care or residential care (Bakketeig et al., 2020; Brännström et al., 2020; Baker et al., 2023; Brady and Gilligan, 2019; Brady et al., 2019a, 2019b; Darmody et al., 2013). Research from different jurisdictions shows that the challenging life circumstances of these individuals, often multiple care and school placements, allied with their experiences at school often result in difficulties and disruptions in their educational careers that may have a negative impact on their academic outcomes, thus possibly limiting their post-school pathway options, including access to third-level education (Darmody et al., 2013; Brady and Gilligan, 2019; Bakketeig et al., 2020; Baker et al., 2020; Baker et al., 2020; Child and Marvell, 2023).

It is important to note that care-experienced young people are not a homogenous group, as their personal circumstances such as foster family characteristics, economic circumstances and challenges associated with changing care settings, changing schools, experience of additional needs, peer and social networks vary. However, as demonstrated by systematic reviews (see Gypen et al., 2017 and Townsend et al., 2020) as well as other studies from different jurisdictions (Cotton et al., 2014; Vinnerljung and Sallnas, 2008; Boddy et al., 2019) on average, these young people tend to have poorer educational outcomes compared to their peers who have not been in state care. In Ireland, there is a lacuna of detailed data on educational outcomes of care-experienced children and young people (Ó Súilleabháin et al., 2024). Data that is available shows that care-experienced children are somewhat more likely to leave school without their Leaving Certificate than their peers (Central Statistics Office (CSO), 2024a). Care-experienced young people are less likely to progress to third-level education, and in case of pursuing this option, tend to do it later as shown by international and Irish research (Williams et al., 2020; Colvin and Knight, 2021; Brady et al., 2019a). Evidence from the UK shows that encountering various challenges, some care-experienced young people leave third-level education before completing their degree (Office for Students, 2021), and by doing so, may lose support from the state. International research has also shown that for some young people, their care experience intersects with other forms of disadvantage, such as minority ethnic or migrant background, belonging to lower socio-economic groups, and belonging to an LGBTQ+ group (Cheruvallil and Anand, 2023; NICE, 2021; López López et al., 2024; Suárez-Orozco et al., 2011; Gypen et al., 2017) that can have an impact on their experiences in the education system.

Much of the discussion internationally focuses on the outcomes of careexperienced young people (Gypen et al., 2017; Parsons et al., 2023), with less focus on structural and systemic factors, thus placing the onus on young people themselves (Liebenberg et al., 2015). However, such factors, including support from the care and education system, play a key role in the experiences of these young people as they move through the care and education systems.

Internationally, various terms are being used for children and young people who do not live with their birth families. For the purposes of this report, the term 'care-experienced' will be used to describe an individual who has experienced any form of formal care, at any point in their life, and for any duration. According to the data from Tusla, there were 5,801 children (0–17 years) in care at the end of February 2025 (Tusla, 2025). Drawing on the lived experiences of a small group of care-experienced young people, this study explores how they fared in the education system and what factors impacted on their post-school pathway to third-level education.

#### 1.2 CARE-EXPERIENCED YOUNG PEOPLE AND EDUCATION

Not all children and young people have equal opportunities when growing up. Much of existing research over several decades has focused on the disparity in educational opportunities and outcomes between different groups of students. This disparity is often reinforced by structural factors, including the education system. In Ireland, some groups of students are more likely to face challenges in the education system, including those from lower socio-economic backgrounds, students with special educational needs (SEN), migrant-origin young people (Devine and Cockburn, 2018; Smyth and Hannan, 2000; Cosgrove et al., 2018; O'Higgins et al., 2015). International and Irish evidence has shown that challenges are also faced by care-experienced young people (Townsend et al., 2020; Gypen et al., 2017; Darmody et al., 2013). Lower levels of educational attainment have been shown to notably influence the future life chances of young people in terms of access to further and higher education, as well as high prestige jobs (Parsons et al., 2023; Parsons et al., 2024).

#### 1.2.1 Academic sphere and care experience

Some care-experienced children and young people grow up in a relatively stable environment, staying with one family throughout their time in care and do not experience many school changes; others undergo several care placements and school changes that may contribute to their educational disadvantage (Clemens et al., 2018; Parsons et al., 2023; Bakketeig et al., 2020; Ellis and Johnston, 2022; McSherry et al., 2013). Several international studies<sup>2</sup> have shown that in general, care-experienced children tend to have poorer educational outcomes compared to non-care experienced young people (Berger et al., 2015; O'Higgins et al., 2015; Cameron et al., 2012, 2018; McSherry et al., 2013) and experience lower educational expectations from their teachers (Mannay et al., 2017a; Harvey et al., 2015). Poorer outcomes can also be associated with a range of other factors, such as quality of support received when in care, placement breakdowns, neighbourhood and community characteristics, and experiences of trauma before entering the care system (Walsh et al., 2018; Taylor et al., 2021). Frequent school changes may also have an adverse effect on mental health, wellbeing and behavioural issues (Townsend et al., 2020). Having a special educational need and/or disability can further impact on the educational outcomes of care-experienced children and young people (De La Fosse et al., 2023). In some cases, age matters – when children enter into care at a younger age, they tend to have better outcomes (Walsh et al., 2018).

On the other hand, research from Ireland and abroad indicates that careexperienced children and young people can perform well at school, especially when receiving sufficient and consistent wrap-around supports, including support with their learning, as well as having more stable placements and good-quality care (Gilligan, 2007; McSherry et al., 2013; Brady and Gilligan, 2019; Moran et al., 2016; Darmody et al., 2013; Walsh et al., 2018; Mannay et al., 2017b). Recent research evidence from a small group of care-experienced young people also shows that despite their challenges in the education and care system, many have similar aspirations as their peers in terms of entry into higher education (Ó Súilleabháin et al., 2024). The authors also highlight the importance of further education in university entry (ibid.).

#### **1.2.2** Social sphere and care experience

Several international studies have highlighted the challenges experienced by young people in the social sphere. Teachers' attitudes, expectations, perceptions and classroom interactions play an important role in how young people experience school and their school engagement (Johnston et al., 2021). In the same vein, care-experienced young people who have a positive and supportive relationship with their teachers and other staff members are more likely to be more engaged with school, see school as a safe environment and fare well in both social and academic spheres (Townsend et al., 2020). However, sometimes teachers may lack sufficient understanding of what it means to be in formal care, at times having stereotypical views of these young people and having low expectations for them because of their care background (ibid.). The situation may be more difficult for care-experienced children and young people in cases when schools have not appointed a designated person to support them, as tends to be the case in Ireland. Acknowledging this,

<sup>&</sup>lt;sup>2</sup> Sebba et al. (2015) highlight the need to utilise a comparator group in research on care-experienced children and young people.

some jurisdictions have put in place special support for these young people. For example, in England, the maintained schools<sup>3</sup> have a statutory duty to designate a staff member with a responsibility of supporting and promoting educational achievement of care-experienced children (Department of Education, 2018). This staff member is tasked with collaborating with key stakeholders (e.g. social workers, foster carers) and relevant agencies, is the main point of contact for the Careers and Enterprise Company (CEC) in schools and a source of information for families who wish to access appropriate services (ibid.).

Friendships and positive peer relationships are important elements in young people's lives as this can promote their mental health and wellbeing (Schwartz-Mette et al., 2020). However, this sphere can often be challenging for careexperienced children and young people, especially for those experiencing multiple school changes that involves moving away from their friends and having to establish new friendships in a new school. These disrupted peer attachments can have a negative impact on care-experienced young people in terms of their confidence and developing an enduring support network. However, a peer group can have either a positive or negative effect on a young person. There is a lacuna of research in Ireland on care-experienced children and their peer relationships. There is some evidence that having good friends is important for care-experienced children, although some find it difficult to establish friendships because of negative stereotyping (Darmody et al., 2013). A review of research by Haddow et al. (2021) on young people in alternative care has shown that a positive and supportive peer group can enhance students' experiences at school while the peer groups that exhibited more negative traits tended to be a strong predictor of problem behaviour. Furthermore, research from the UK and Ireland shows that some young people may experience bullying because of their care background (Dansey et al., 2019; Darmody et al., 2013). Longitudinal data from England has shown that among this group, bullying increases with age, with many being continuously bullied between the ages of 14-16 years (Green et al., 2010). Research from the UK highlights the importance of friendships for care-experienced children and young people, particularly regarding improving their life satisfaction (Roesch-Marsh and Emond, 2020).

Another area of importance is the relationship between care-experienced young people and their care workers and social workers. While Gilligan et al. (2022) in Ireland highlight the importance of continuity and quality of such relationships regarding student outcomes, studies from elsewhere show many young people experience disrupted relationships or do not receive sufficient support from their care workers (Munford, 2022). The lives of many care-experienced children and young people are filled with uncertainty, whether due to care placements or school changes (Mølholt et al., 2022). Thus, not receiving sufficient support from care and social workers can have a detrimental effect on their outcomes. The relationship

<sup>&</sup>lt;sup>3</sup> Most schools in the UK are state schools. There are four types of maintained schools: community schools, foundation and trust schools, voluntary aided (VA) schools and voluntary controlled (VC) schools.

between the foster family and a young person in their care is equally important. Research from Ireland indicates that moving into a stable foster care environment had a positive impact on young people's school experiences (Darmody et al., 2013). International research has shown that good, supportive and warm relationships have been found to enhance a young person's self-esteem, and emotional and behavioural development (Sprecher et al., 2021). The authors note that developing a trusting relationship with foster parents can take time as young people need to overcome their experiences of previous relationships in their lives. This considered, multiple care placement changes can be particularly disruptive and disorienting for a young person.

Certain measures have been used in order to provide these young people with additional support from a key adult. For example, as indicated by research from Ireland, mentoring programmes have been found to have a positive impact on care-experienced young people in developing their confidence, supporting their social and emotional wellbeing and educational engagement (Brady et al., 2019a, 2019b). As shown by research from Australia, mentoring can take several forms, including formal mentoring programmes, a naturally developed supportive and trusting relationship with an adult, a supportive relationship with another young person close to the care-experienced person's age, peer mentoring (close age and shared life experiences) and therapeutic mentoring that is based on formal therapeutic intervention (Lohmeyer et al., 2024).

#### 1.2.3 Leaving care

Leaving school is considered a major transition point for all young people (McCoy et al., 2014). International and Irish research has shown that care-experienced children and youth have often experienced multiple placement transitions during their educational career that often coincide with the key transitions in the education system (Bayfield, 2023; Darmody et al., 2013). In Ireland, young people leave care at 18 years of age<sup>4</sup>. Leaving the care system can be seen as an interruption of an existing support network and may result in young people needing to rely mostly on themselves (Atkinson and Hyde, 2019; Ellis and Johnston, 2022). Irish research has highlighted challenges associated with access to an aftercare worker before a young person leaves care (Mullan, 2022). Research from Australia indicates that in many cases these young people do not have sufficient information about the possibilities and opportunities beyond secondary school and may have difficulty accessing existing services and supports (Muir and Hand, 2018). In line with similar studies, research from Norway shows that compared to their non-care peers, these young people often lack the supportive family structure to offer support and guidance (Storø, 2018). Hence access to both formal and informal supports when leaving care is particularly important for this group of young people (Paulsen and Berg, 2016). Having at least one good adult, people that

<sup>&</sup>lt;sup>4</sup> Young people leaving care at the age of 18 are entitled to aftercare services (including in-person support, an individual plan and financial assistance) based on their assessed needs.

these young people can turn to, offers an invaluable support for care-experienced young people (Storø, 2018). Aftercare services in Ireland generally include support from a designated support person, whose support and guidance a young care leaver can draw on (Glynn and Mayock, 2018). However, there seem to be delays in assigning an aftercare worker 19–24 months before leaving care, as recommended by national guidelines, and some young people do not have an aftercare worker assigned before they turn 18 years of age (Mullan, 2022). Leaving care can be understood as an 'accelerated' transition for many, considering the faster rate of entry into adulthood and independent living, compared to young people who have not been in care of the state (Palmer et al., 2022; Taylor et al., 2021; Dixon et al., 2018), experiencing this transition and associated challenges sooner than their peers (Stein, 2012). The outcomes of these young people are generally associated with the availability and quality of aftercare support (Dixon et al., 2018; Daly, 2012; Mullan, 2022).

#### 1.2.4 Aspirations, expectations and entry into third-level education

Much of international and Irish research has explored post-school aspirations and decision-making of young people (Billet et al., 2024; Belando-Montoro et al., 2024; Cuervo et al., 2019; McCoy et al., 2014; Faas et al., 2019). In general, young people from more advantaged backgrounds tend to have higher post-school aspirations compared to their less advantaged peers (Schoon and Eccles, 2014; Berrington et al., 2016). Reflecting the expansion of higher education in developed countries, more young people now decide to select this option as their future pathway. However, their aspirations are formed and developed throughout one's schooling career, depend on a number of personal (e.g. experience in care, mental health, learning difficulties) and structural (e.g. education and care system) factors (Sellar and Gale, 2011; Khattab, 2014; Gorard et al., 2012; Furlong and Biggart, 1999) and necessitate access to support and advice at different key junctions throughout the compulsory schooling (Osterman, 2000; Nelis et al., 2021; Boguslaw, 2017).

There is also a growing interest in post-school pathways of care-experienced young people. Some international (Henderson et al., 2016) and Irish studies (Ó Súilleabháin et al., 2024) have shown that post-school aspirations and goals, including entry to third-level education, of these young people may not differ notably from their non-care peers. This can be explained, in part, by care-experienced young people having the same career goals as their peers (Mannay et al., 2017a). However, according to this study from Wales, the aspirations tend to be lower for an older cohort of young people, which could be associated with their educational career being disrupted more often (ibid.). It is important to note that the findings regarding aspirations and ambitions of care-experienced young people are mixed. For example, contrary to the findings of Henderson and colleagues in England (2016), another English study has shown that care-experienced young people have lower expectations of moving on to higher education in school year 9 (age 13/14) compared to their same age peers who have not been in care, even when controlling for special educational needs (SEN) status, history of school

exclusions and family benefit levels (Williams et al., 2020). These authors argue that the aspirations and expectations of care-experienced young people are associated with the support offered by the schools, the extent to which their social workers valued education, and support received from their foster parents (ibid.).

Access to effective support systems is a central theme running through the literature on aspirations and ambitions of care-experienced young people (Mannay et al., 2017a). In some cases, not picking the right subjects at the right time reduces the opportunities available to young people later on. Hence the provision of guidance and support should start early in their schooling career and should involve support from care workers and foster families (Creed et al., 2011; Crawford et al., 2011; Munro et al., 2011). The interventions should also consider personal circumstances of young people, including their experiences of trauma, as their experiences can shape aspirations and post-school decision-making (Colvin and Knight, 2021; Bayfield, 2023). In Ireland, there has been a call for emotional counselling and therapeutic supports to be provided in all primary and second-level schools (Downes, 2022). It is important to note that post-school decision-making often coincides with preparation to move out of the care system, and additional stress associated with this (Colvin and Knight, 2021; Mullan, 2022).

Previous research from Ireland has highlighted the importance of guidance counsellors in advising students about options available to them after finishing school (McCoy et al., 2014). Such advice is often invaluable for more disadvantaged groups, including care-experienced young people (Collins et al., 2023) and supports the development and realisation of their career aspirations (Colvin and Knight, 2021). However, guidance received at school, which may often be of variable quality (McCoy et al., 2006), may leave many young people undecided and unsupported. In Ireland, guidance in second-level schools is integrated into both Junior and Senior Cycle. The service is provided by guidance counsellors who are qualified teachers and hold the relevant recognised qualification for schoolguidance work. Despite the importance of the role, there is relative lack of research evidence on the impact of guidance counselling on students (Hearne and Galvin, 2014), with disparate levels of this service identified by earlier research (McCoy et al., 2006). More recent research has highlighted challenges encountered by guidance counsellors and students with regard to increased demands for the service and constraints on time for the service, especially for more personalised, one-to-one discussion (McCoy et al., 2014; Healy, 2019).

Existing international research has also shown that care-experienced young people are less likely to enter directly into higher education after finishing school, compared to their peers (see Harvey et al., 2015 for Australia; Bayfield, 2023 for Wales; Ellis and Johnston, 2022 for England). Yet, many do so later in life (Harrison, 2020; Abbott-Chapman, 2011), following a non-linear pathway to higher education (Ellis and Johnston, 2022). In Ireland, an exploratory study by Brady and Gilligan (2019) based on 18 participants has also shown that while some care-experienced young people have disrupted educational careers, others progress to higher or further education. However, for some this progression took place later in life. Data from Ireland shows that while the proportion of care-experienced young adults enrolled in further and higher education in 2022 was similar to all young people (52% vs. 49%) compared with all children aged 18 to 23, a larger proportion of care-experienced young people attended further education courses (36%) than all young peers (10%). The figures for attending higher education courses were 16% and 39%, respectively. Some in further education may have subsequently progressed to higher education (CSO, 2024a).

Progression to higher education, or third-level education more generally, can be challenging for young people. Existing studies have highlighted the difference between school and higher education. Research from Ireland has indicated that adjusting to the new learning environment can be challenging for all students, although after an initial adjustment period, students get used to different approaches to teaching and learning (Denny, 2020). International research has shown, however, higher education tends to be aimed at a 'typified imagined student' who accesses and moves through higher education in a linear process (Child and Marvell, 2023). This situation is often different for care-experienced youth as their educational trajectories can be more complex, shaped by their previous experiences when still at school, as demonstrated by research from the UK and Ireland (Harrison, 2017; Brady et al., 2019a). Furthermore, complex personal circumstances may also impact on their ability to access and sustain college and university courses (Bayfield, 2023). These challenges may be compounded by moving residence, lack of or limited support networks, social isolation, emotional and mental health issues, workload, financial difficulties, and interpersonal issues (O'Neill et al., 2019; Bayfield, 2023; Child and Marvell, 2023). For some care-experienced students, moving accommodation may also trigger their trauma related to moving between different care placements (O'Neill et al., 2019). Furthermore, some accommodation arrangements can be problematic during the holidays, as care-experienced young people lack an alternative place to spend this time (Child and Marvell, 2023).

In order to sustain and support their education, care-experienced young people need to have access to reliable, consistent relationships with a trusted member of staff or another adult with the necessary knowledge about how to access practical, emotional and financial support (O'Neill et al., 2019; Bayfield, 2023; Ellis and Johnston, 2022). Insufficient support may make some students to consider dropping out of post-school education (Ellis and Johnston, 2022; O'Neill et al., 2019). Research in Ireland has highlighted the challenges associated with identifying students with care experience who have applied to study at third level (Ó Súilleabháin et al., 2024). At present, such students can disclose their care status on their CAO application if they apply to the Higher Education Access Route (HEAR) scheme, although not all students may wish to do so.

Considering the various challenges care-experienced young people face when making a transition from school to third-level education (Ó Súilleabháin et al., 2024; Brady et al., 2019a), additional support interventions may be necessary. For example, in Wales, summer schools or taster days are available, and some are

tailored specifically for care-experienced young people (Bayfield, 2023). Other supports that have been recommended include welcome packages, mentoring and named sources of support, as well as emotional or wellbeing support (ibid.). It is also recommended that third-level education institutions work together with care-experienced young people in deciding what supports are most appropriate, considering that these students often have diverse needs (Child and Marvell, 2023).

In order to develop targeted supports for care-experienced young people who have entered third-level education, these institutions need to be aware of the care background of the students. However, some young people may not want to reveal their background, having previously experienced stigma and being 'treated differently' (Ellis and Johnston, 2022; Townsend et al., 2020). The research findings on whether young people want to be identified as being in care vary, however. A study on school-age children and young people in Ireland found that they did not want their peers to know they were in care (Darmody et al., 2013). However, a study in Scotland indicated that most young people declared their care experience when applying to college or university and found this experience to be positive (O'Neill et al., 2019). It could be that care-experienced young people may be more comfortable in declaring their care background at an older age. Furthermore, they may be more willing to do so, if this may result in access to additional support from the third-level institution they are attending.

Some narratives portray young people who progress into college and higher education as 'successful' or 'resilient', thus placing the onus on these young people regarding their outcomes in the education system, rather than considering systemic and other factors that shape their experiences and outcomes (Bayfield, 2023). The 'resilient student' label is often juxtaposed with discourses around low levels of educational attainment and other issues, thus potentially contributing to stigma attached to being in care (Bakketeig et al., 2020). Yet, much of the existing research on educational disadvantage has highlighted the importance of the 'distance travelled' – that is the progress of the young person relative to their starting point when assessing students' outcomes (Thomas et al., 2017). Hence, studies need to focus on what it means to be successful for different groups of students.

#### 1.3 CARE-EXPERIENCED YOUNG PEOPLE IN IRELAND

Recent figures for Ireland show that 5,801 children (0–17 years) were in care at the end of February 2025 (Tusla, 2025). An overall majority (88%) tend to be in foster care (including general and relative foster care) (CSO, 2024a). International research has highlighted the negative impact of high mobility of care and school placements on young people. The figures for Ireland show that many care-experienced young people are in a stable care placement, with half of the children (50%) having a single care placement (CSO, 2024a). However, a small percentage of the children (9%) in this group had more than five placements (ibid.). The data

also show that compared to all children, a higher proportion of care-experienced children had experienced more than one primary and secondary school change (CSO, 2024a)<sup>5</sup>.

Regular school attendance is often associated with better educational outcomes of children (Klein et al., 2022). Frequent school and care changes may have an adverse effect on school attendance of young people and their educational outcomes. In Ireland, students who miss more than 20 days of school are referred to a branch of the Child and Family Agency, Tusla Education Support Service (TESS). School attendance is generally shaped by a combination of institutional and individual factors (Thornton et al., 2013) and can be problematic for care-experienced children due to their personal circumstances (Mercieca et al., 2021). In terms of school attendance in Ireland, high absenteeism (those absent from school for 20 days or more) in school year 2022/23 was slightly higher for care-experienced children (17%) than for all other children (15%)<sup>6</sup>, with the most common reasons for absenteeism being 'unexplained' (52%) and illness (23%) (CSO, 2024a). Being in foster care rather than in other care types was associated with fewer cases of serious absenteeism (ibid.).

International research has shown that in general, care-experienced children tend to have poorer educational outcomes. In Ireland, only a small percentage of care-experienced children (6%) attending primary or second-level schools repeated at least one year of schooling (CSO, 2024a). However, care-experienced children were more likely to leave school without their Leaving Certificate than their peers (28% vs 8%) (ibid.). Being in foster care with only a single placement versus other types of care and more placements seemed to act as a protective factor – those in foster care and with a single placement while in care tended to fare better in terms of completing their Leaving Certificate (ibid.).

The population in Ireland is highly educated, with over half of 25–64-year-olds having a tertiary qualification (CSO, 2024b). Individuals with higher education also have higher rates of employment and better life chances (Edgerton et al., 2012). Recognising the need to assist young people leaving care and support their post-school education, Tusla carries out a needs assessment<sup>7</sup> and provides support in aftercare planning, which addresses a number of areas including education, training and employment, but also more practical elements, such as finances and budgeting, accommodation, health and wellbeing, personal and social

<sup>&</sup>lt;sup>5</sup> Data on enrolments into primary school over the years (from 2015/16 to 2022/23) shows that care-experienced children were more likely to be enrolled at more than one primary school, compared to all children (32% vs 14%). The corresponding figures for secondary schools in the period 2012/13 to 2022/23 were 20 per cent and six percent accordingly (CSO, 2024a).

<sup>&</sup>lt;sup>6</sup> The figures are based on the proportions of children absent 20 days or more amongst all those aged 7–17 years in January 2024 (CSO, 2024a).

<sup>&</sup>lt;sup>7</sup> The process of leaving care is supported by needs assessment when a young person is 16/17 years of age. A written form is filled in in collaboration with the young person, aftercare worker, and any other adults who provide support to the young person. This needs assessment forms a subsequent aftercare plan (from six months before a young person turns 18 up to 21 years old), the aim of which is to teach young person independent living skills. A young person is linked up with an aftercare worker for ongoing support (for further information see www.tusla.ie/services/alternative-care/after-care/tell-me-more-about-how-aftercare-works).

development. The process is supported by aftercare workers<sup>8</sup> from Tusla, as once a young person turns 18 years of age they are no longer supported by a social worker. In order to access aftercare services, a person needs a referral from a social worker, a social work department, family/foster family member, or provide a selfreferral at any of the drop-in service clinics. A monthly allowance is available to care leavers up to age 23 if a young person is still engaged in full-time accredited training or education at this point. The figures from Tusla show that 75 per cent of care leavers (18–22 years) in receipt of aftercare services are in education/training (Tusla, 2023).

Although care-experienced young people are a distinct group, regarding their life experiences, they have not been included as a target group in the earlier National Access Plans. However, the *Strategic Action Plan for Equity of Access, Participation and Success in Higher Education 2022–2028* has recognised the needs of specific groups of young people seeking to access higher education, including those who have experience of the care system.

While previous Irish research has explored the pathways of children and young people when leaving school more generally (McCoy et al., 2014), post-school transitions from state care to higher education has received limited policy attention in Ireland (Brady et al., 2019a). This study addresses a gap in research in this area.

#### 1.4 AIM OF THE RESEARCH AND RESEARCH QUESTIONS

The aim of this small-scale exploratory study is to gain a better understanding of the main factors that influence the post-school decision-making of careexperienced students regarding entry to third-level education and the challenges associated with this transition. The study examines the experiences of those young people who progressed into third-level education as well as those whose thirdlevel education was disrupted (i.e. who enrolled in a third-level course and subsequently dropped out) and explores the views of a group of key stakeholders. Drawing on students' lived experiences, the study seeks to inform policy development in supporting care-experienced children when still at school, their post-school transitions as well as offering suggestions as to how best to support these young people. The use of young people's own perspectives highlights various challenges they face in school and beyond, particularly valuable for policymaking in this area. Furthermore, the study provides insights for institutional practices and support services for young people.

The study contributes to the international and national literature on the factors that shape the decision-making processes of care-experienced young people regarding their post-school pathways, an area that is relatively under-explored to

<sup>&</sup>lt;sup>8</sup> At present, there are 71 aftercare workers employed by Tusla, holding degrees in areas such as social work and related areas (personal communication from Tusla, 2 April 2025).

date. The study addresses the following research questions:

- 1. How does care experience impact on young people's experiences in school?
- 2. How does care experience impact on young people's interaction with friends, teachers and other school staff?
- 3. What are the main factors that shape the decision-making process of care-experienced young people regarding their post-school pathways?
- 4. What are the main challenges care-experienced young people face after leaving school and entering third-level education?

#### 1.5 METHODOLOGY

This small-scale qualitative exploratory study seeks to provide an insight into the challenges faced by care-experienced young people at school and beyond – namely those who have pursued third-level education, and those whose third-level education was interrupted.

#### 1.5.1 Study design

Over the years, policy debate and academic research has grown to recognise the importance of young people's own perspectives and lived experiences. This approach can provide useful insights into the issues that are important for them and helps to inform policy development in an area under discussion. This considered, the present study sought insights from young people with care experience as they progressed through school into third-level education. The young people were chosen based on predefined characteristics (see below), and the questions were designed to shed light on the main factors that influence the post-school decision-making of care-experienced students regarding entry into third-level education.

The study comprised distinct stages:

**Stage 1** involved desk-based research on international best practices, regarding state interventions, that have been found successful in supporting care-experienced students' access and retention in higher education settings.

**Stage 2** involved primary data collection from care-experienced young people who have pursued third-level education, including higher education. In addition, the study included the perspectives of young people who have dropped out of their course, thus providing an additional and valuable dimension to the study. As the study was not seeking sensitive information related to the reasons for being in care, focus group discussion was considered an appropriate approach for the study. This approach has successfully been used in international research on care-experienced young people (see Ogunshakin et al., 2021; Pinkney and Walker, 2020). These online focus groups were organised with the care-experienced young people from various locations in Ireland. The interviews were digitally recorded

and transcribed and were later analysed by utilising a thematic approach. Thematic analysis enables the researcher to get familiar with the data by coding the content, and to search for patterns or themes across transcripts (Naeem et al., 2023). This is a commonly used technique in analysing interview and focus group data (Xu and Zammit, 2020). In this study, this approach enabled identification of respondents' convergent or divergent views across a number of areas discussed during the interviews and helped to identify issues that the interviewees considered important. The interviewees were assigned pseudonyms to protect their identities (see Table 1.1).

**Stage 3** involved consultation with the main stakeholders and professionals (e.g. social workers, aftercare managers, foster care association members, university Access programme coordinators and admission officers, guidance counsellors, school completion programme coordinators). The online consultation forum involved presentation of preliminary results from the focus groups, followed by random allocation of the participants into discussion groups. Each group was asked to discuss three topics and the feedback from each group was presented to a larger audience by a self-appointed rapporteur. The results of the forum were incorporated into the report to further shed light on the topic and to include their views in policy implications on how support, access, retention, and success rates for care-experienced students in third-level education in Ireland could be enhanced.

#### 1.5.2 Sample selection

In order to achieve the aim of the study and answer the research questions, input was sought from care-experienced young people who have progressed into third-level education<sup>9</sup> and whose participation in third-level education has been interrupted.<sup>10</sup> The focus group participants were identified and recruited by an advocacy group Empowering People in Care (EPIC). Five online focus groups were conducted, with a maximum of six individuals in each group. In total, 16 young people were interviewed, four males and 12 females, from Munster, Leinster and Connacht. The study targeted young people who had left second-level education and were 18 years of age or older. The age range of the participants was 18–25. Regarding ethnicities, an overall majority were White Irish, some Black African, other Black background, mixed ethnicities and Traveller background.

<sup>&</sup>lt;sup>9</sup> In Ireland, third-level education refers to all post-secondary education, including higher education in universities and colleges and further education on Post Leaving Certificate (PLC) and other courses.

<sup>&</sup>lt;sup>10</sup> See section on limitations of the study.

Pseudonym	Gender	Situation at the time of the interview
Mateo	М	In third-level education (PLC)
Theodore	Μ	In third-level education (university)
Isabella	F	In third-level education (college)
Ava	F	In third-level education (college)
Luna	F	In third-level education (university)
Camila	F	In third-level education (PLC)
Hazel	F	In third-level education (college)
Nora	F	In third-level education (university)
Mila	F	In third-level education (college)
Levi	М	In third-level education (university)
May	F	In third-level education (college)
Lily	F	In third-level education (college)
Ezra	Μ	Interrupted third-level education (college)
Sofia	F	Interrupted third-level education (college)
Eliana	F	Interrupted third-level education (college)
Willow	F	Interrupted third-level education (college)

#### TABLE 1.1 DESCRIPTIVE DETAILS RELATING TO THE YOUNG PEOPLE WHO PARTICIPATED IN INTERVIEWS

*Note:* All names have been anonymised to protect the identity of participants. Third-level education refers to all education after second-level education, including higher education in universities and colleges as well as further education on Post Leaving Certificate (PLC) and other courses.

The topics (see Appendix for interview schedules) covered in the focus group discussion with care-experienced young people included the following:

- Care placements and settling in at school
- Social experiences at school
- Academic experiences at school
- Decision-making and progression
- Leaving school
- General reflections

The focus groups with care-experienced young people were complemented by an online consultation event including a range of key stakeholders (e.g. social workers, aftercare managers, foster care association members, university Access programme coordinators and admission officers, guidance counsellors, school completion programme coordinators, policymakers) to explore the main issues highlighted by the interviews in greater detail. The event, held on 23 May 2024, was attended by 19 individuals. The stakeholders were identified by their online profile and invited to participate in the event. The participants were randomly assigned into three discussion groups in 'virtual rooms'. The feedback from the groups was recorded, transcribed and thematically analysed.

#### 1.5.3 Ethics and confidentiality

The research design and instrument were approved by a Tusla independent Research Ethics Committee (REC). The proposal to the REC followed specific guidelines. These guidelines apply and an REC review is required where the research relates to the following areas of the Agency's responsibility:

- Potential research participants identified from, or because of their past or present use of services provided by the Agency (including services provided under contract with the private, voluntary or community sectors).
- Potential research participants identified because of their status as relatives or carers of past or present users of the Agency's services.
- Potential research participants identified because of their status as providers of the Agency's services.

The procedures ensure the maximum benefit of the research while minimising the risk of actual or potential harm; and seek to protect, as far as possible, all groups involved in the research, including participants, researchers and research teams throughout the lifecycle of the research<sup>11</sup>.

The study also followed standard ethical guidelines applicable to vulnerable groups involving obtaining informed written consent to participate in the study<sup>12</sup>. Furthermore, the interviewer was vetted by An Garda Síochána in accordance with their procedures. As all interviewees were older than 18 years of age, consent was sought directly from them (see Appendix II). The interviewees were assured that the information gathered would be anonymised and the following issues were explained to them:

- No individual will be identified.
- The aim is to identify general themes arising from the interviews.
- The participant can refuse answering questions and is free to terminate the interview at any point.
- The information will be used for research purposes only.
- The data gathered will be stored on a secure network in the offices of the ESRI and will not be taken outside the premises.
- Access to the data is limited only to the researcher working on the project.

#### 1.5.4 Limitations of the study

The study has certain limitations, particularly in terms being a small-scale

<sup>&</sup>lt;sup>11</sup> For further information see www.tusla.ie/research/tusla-research-office/research-ethics-committee/guidelines-on-the-functioning-of-tuslas-research-ethics-committee.

<sup>&</sup>lt;sup>12</sup> This is in line with Guidelines on Ethical Research by the British Sociological Society.

exploratory study with a relatively small sample size of 16 care-experienced young people and 19 key stakeholders. Furthermore, in the selection and recruitment of young people for focus groups, the researcher relied on EPIC to identify and invite care-experienced young people to participate in the study who fell into the following categories: 'in tertiary education', 'interrupted tertiary education' for the study. There were also more females than males in this study, possibly indicating that more females choose to enrol in third-level education. Despite the small number of young people, the participants represented a range of experiences in the education system and enabled this exploratory study to identify factors that are likely to impact on the post-school pathways of this group of young people, particularly regarding their entry to third-level education. However, the findings cannot be generalised to the population of care-experienced young people. Any future research aiming to provide a more comprehensive picture would benefit from a larger number participants, also incorporating the views of these young people who have not progressed to tertiary education, for whatever reason. Regarding stakeholder consultation, there was a good range of individuals representing different organisations and the discussion was exploratory in nature. Future research could explore the perspectives of different groups of key stakeholders (social workers, aftercare workers, guidance counsellors and others) regarding their engagement with care-experienced young people in greater detail. Despite the limitations, the information collected complements earlier research while addressing the gap in research on factors that impact on decision-making about post-school pathways and challenges associated with continuing their education at third level.

#### 1.6 OUTLINE OF THE REPORT

Following this introductory chapter, Chapter 2 provides information on international best practices that have been considered successful in supporting care-experienced students' education, leaving care and accessing post-school education. Chapter 3 discusses care placements and settling in at school. It looks at coping with challenges regarding these moves, communications between schools and social workers, coping with mental health issues, relationships between foster parents and care professionals. Chapter 4 explores social experiences at schools. It discusses the intersection between care experience and school, sources of support, and making friends. Chapter 5 focuses on academic experiences at school. The topics discussed in this chapter include school attendance, programmes and grades, help with learning, expectations of teachers and other adults. Chapter 6 discusses decision-making of care-experienced young people regarding entry into third-level education, while Chapter 7 focuses on the transition from care to aftercare and third-level education. Chapter 8 summarises the main findings of the study and discusses their implications for the policy development as to how access, retention, and success rates for care-experienced students in Ireland can be enhanced.

### **CHAPTER 2**

# Transition of care-experienced young people into higher education: Interventions and best practices

#### 2.1 INTRODUCTION

Considering the importance of educational attainment for further life chances, policies across Europe and beyond have focused on improving access to post-secondary education for young people. However, despite the widening of participation in higher education, gaps still exist between students from disadvantaged backgrounds and their more privileged peers (Robinson and Salvestrini, 2020). Much of the existing research to date has indicated that care-experienced children and young people encounter challenges regarding their transition to independent living and adulthood. These challenges apply to a number of domains including employment, education and training, wellbeing and health, finance, accommodation, to name but a few (Abbott-Chapman, 2011; Munford, 2022; Alderson et al., 2023; Baker et al., 2023). As shown in the previous chapter of this report, many care-experienced young people have encountered various disruptions and other challenges when still at school and these experiences often shape their post-school aspirations and pathways.

International research has indicated that care-experienced young people who aim to continue their education in further or higher education need continuous and consistent support starting from school and continuing throughout their education (Children's Commissioner, 2024; Gazeley and Hinton-Smith, 2018). This chapter focuses specifically on education of care-experienced young people, exploring interventions that have been successful in supporting care-experienced students' educational career. It is important to note, however, that while lessons can be learned from other countries in terms of successful policies and practices, caution should be applied in 'policy borrowing' (Raffe, 2011) because of contextual differences between countries (O'Donnell et al., 2020). Rather than offering suggestions for 'blueprint solutions', this chapter looks at various international interventions put in place to support care-experienced young people moving through the education system and beyond, to suggest ideas for 'policy learning' in Ireland. While the focus of this chapter is on accessing third-level education, this transition point needs to be considered in the context of earlier educational experiences of young people who have been in care of the state, while taking into account other factors that may have an impact on their educational experiences. The chapter starts with discussing interventions and practices in schools, moving then to discussing leaving care. Section 2.2.3 focuses on supports available for care-experienced young people in higher education. Section 2.3 discusses the supports available for these young people in Ireland. Section 2.4 presents conclusions.

# 2.2 INTERNATIONAL INTERVENTIONS AND PRACTICES SUPPORTING THE EDUCATION AND WELLBEING OF CARE-EXPERIENCED YOUNG PEOPLE

As discussed in Chapter 1, existing research has shown that care-experienced children and young people tend to have lower levels of educational achievement and attainment, compared to their peers who have not been in care. Their experiences in primary and secondary education often impact on access to college and higher education as some young people may not have the required grades or have low post-school aspirations. In order to support young people, the interventions should start early, in primary school. To improve educational aspirations and attainment, across different jurisdictions, various initiatives have been set up to support young people.

#### 2.2.1 Educational support for care-experienced young people in schools

In Canada, Teach Your Children Well is an educational intervention aimed at children and young people between six and 13 years of age. The programme includes two components – direct teaching and behaviour management (Evans et al., 2017). The programme can be delivered either at individual level or in small groups, with the duration either 25 or 30 weeks. At individual level, the intervention is usually delivered by foster parents to the child/children in their care and involves three hours of tuition a week in reading and mathematics. In case of small groups, this intervention is delivered by volunteer university students, with care-experienced children receiving two hours of tuition per week. To explore how well this programme has functioned in practice, a randomised control trial at individual level was undertaken (Flynn et al., 2011, 2012). The results showed that the intervention had a positive effect on sentence comprehension and mathematics computation. However, no significant effects were found on reading and spelling. Implementation of the intervention had encountered some challenges, mostly concerning time needed by the caregiver to provide additional tuition, but also the caregiver's health issues, placement changes, carers' views that no additional tuition is needed, or tensions between the child and the caregiver (Flynn et al., 2012). Where additional tuition was provided in small group settings, the programme had some effect on reading, spelling and mathematics computation. No difficulties regarding implementation of the programme were reported by the authors regarding implementing the intervention in a group setting (Flynn et al., 2012).

Delivered by University of Manchester in the United Kingdom, **Success4Life**, launched in 2012, is a programme for care-experienced children and young people 11–16 years of age who are in local authority care with an aim to widen participation and assist in aspiration-building. For ten weeks, the participants are involved in a range of group work activities and themed sessions to develop their confidence and transferrable lifelong skills, as well as raising their awareness of higher education. In order to not interrupt the school day, the sessions take place on campus from 4–6pm on a weekly basis. To provide maximum support, the

programme is delivered to a small group of targeted students with two staff members for each student. The students are nominated either by their teacher at school or their social worker. To facilitate participation, paid transport is provided for the students. The results of the programme evaluations have shown that an overall majority of students who participated in the programme stayed on in education after the age of 16, following GCSE exams and intended to continue their studies in higher education. They also tended to have a better understanding of the higher education sector and courses available therein. Lewis et al. (2015)<sup>13</sup> note that feedback from carers, social workers and students who had availed of the programme was very positive, with the majority of participants reporting their intention to stay at school, apply to enter higher education as well as increased self-confidence.

#### 2.2.2 Support for young people leaving care

Various existing studies have highlighted the challenges care-experienced young people face when leaving care across various dimensions including employment, education, finances and accommodation, among others (O'Donnell et al., 2020; Taylor et al., 2021). While still at school, these young people tend to have multiple sources of support, but less support tends to be available for them once they leave the care system. As young care leavers face multiple challenges that often overlap, various interventions have been utilised that provide the necessary scaffolding when entering into adulthood, which for many also means continuing their education. The interventions are all the more important if a young person has complex needs.

Berry Street (Child and Family Services) in Victoria, Australia utilised the Stand by Me programme in part of Melbourne, aimed at care leavers with multiple and complex needs. Once the young people turned 16, they could avail of a wraparound pre-exit support, ensuring that each person had a leaving care plan. These young people worked with their personal advisers - a consistent point of support, who provided them with relevant information, implemented and reviewed care leaving plans and acted as a contact person and advocate, facilitating communication between the young person and different support services (e.g. substance abuse, mental health, disability, accommodation) until the young person turned 21 years of age (Purtell et al., 2016). The evaluation of the programme established that the intervention improved outcomes for the care leavers and alleviated 'leaving care anxiety' often experienced by this group (ibid.). When comparing outcomes of young people who had availed of the programme with those who did not, the evaluation showed that as the intervention provided a more holistic support for these young people, it proved to be a particularly important support for those young people who were at the highest risk of homelessness and other adverse outcomes (ibid.). This approach resembles the

<sup>&</sup>lt;sup>13</sup> While the evaluation did not involve a control group, the participants filled in pre-project and post-project questionnaires.

aftercare support model available to care-experienced young people in Ireland. Care-experienced children and young care leavers in Ireland are also supported by a rights-based organisation, Empowering People in Care (EPIC), who advocate for their rights<sup>14</sup>.

In the United Kingdom, Care Leavers' Social Impact Bond (SIB) is an umbrella programme aimed at supporting young care leavers across various domains: 1) improving their outcomes accessing education, employment or training (EET); 2) exploring new approaches to supporting care leavers; and 3) developing the capacity of local authorities (LAs) to support care leavers. In terms of the first domain - access to education and training, an evaluation of three programmes -Reboot West, i-Aspire and Apollo was undertaken between 2018–2023<sup>15</sup>. The evaluation incorporated a number of components: a) longitudinal qualitative research with project stakeholders and care leavers; b) monitoring information (MI) data collection and analysis; c) a quasi-experimental impact assessment; and d) a value for money assessment (Davey et al., 2023). All three projects were aimed at care leavers (average age 19.5) who had high-end need, although some differences were highlighted across the interventions in how young people were targeted for the programmes. The results of the evaluation showed that the stakeholders had developed a good understanding of the challenges young care leavers experience when moving to EET. It was acknowledged that for many care leavers, transition into EET is a gradual and staged process, rather than a linear one and that many young people face various barriers while moving into EET. All three projects were considered helpful in supporting young people's post-care transition (Davey et al., 2023). In terms of care leavers themselves, the responses showed that most of them were satisfied with the support they got by participating in the programmes, especially if the support was consistent and young people were able to develop trust and a bond with their support person (Davey et al., 2023). The evaluation showed that support needs to be flexible, as the circumstances and needs of young people tend to change over time. Interagency collaboration in supporting young people in the programme was considered to be a particular strength of the programmes (Davey et al., 2023).

#### 2.2.3 Higher and further education

As discussed in the previous chapter in this report, across Europe, and elsewhere, care-experienced young people tend to be underrepresented in higher education for a number of reasons, including low self-belief and post-school aspirations. Data from the UK indicates that care-experienced young people who have entered higher education can have different experiences – while some find the experience positive, others may face various challenges, may not continue beyond their first year, may spend longer finishing their degree, or achieve lower scores (not 1st or

<sup>&</sup>lt;sup>14</sup> For more information about EPIC, see www.epiconline.ie/about-epic.

<sup>&</sup>lt;sup>15</sup> i-Aspire (South East London) had the largest care leaver population across their three LAs (n = 634); Reboot West (West of England) had the second largest population of 590 care leavers across the four LA areas; Apollo (Sheffield) had the smallest population with 231 care leavers in that LA. It included young people aged 19–21 years.

#### 2:1) (Office for Students, 2021).

With a view to supporting care-experienced young people in accessing higher and further education, the Springboard programme, aimed at 16-18-year-olds, was developed in the United Kingdom. The programme aims to develop young people's aspirations regarding their post-school pathway into higher education (Sanders et al., 2021). In order to make this pathway more relatable, a personalised letter is sent from care leavers already in further and higher education to young people still in the care system. Such an approach was seen to encourage young people who were still uncertain about their post-school options to consider continuing their education (Sanders et al., 2021). Furthermore, information was shared about initiatives such as the Propel website, where care leavers can find useful information (e.g. applying, funding options, accommodation, rights and entitlements) about entry into higher end further education. As young care leavers often lack role models regarding their post-secondary education, the website included stories and experiences of care leavers already in further and higher education with a view to develop aspirations among care-experienced young people still at school. Furthermore, the website also offers useful information for teachers, parents and carers, as well as universities and colleges. Young people can use this website for contacting the Care Advice Line, run by Become (the national charity for children in care and young care leavers). In addition to the personalised letter, the Springboard programme also involved the recruitment of a further education/higher education advice and support officer for a year, who offers targeted support and advice to students and professionals about various issues concerning higher and further education. Assessment of the programme indicated that the intervention had a positive impact in encouraging care-experienced young people to access higher and further education (Sanders et al., 2021).

In some cases, initiatives are set up by higher education (HE) providers. For example, Go Higher West Yorkshire (GHWY) in the UK is a partnership of 13 higher education providers, working in collaboration in supporting access to, success in and progression from higher education of groups underrepresented in HE. The initiative involves teachers, advisers, carers and employers, to make higher education a viable option for young people. Under the umbrella of GHWY, Care to Go Higher offers a nationwide CPD-certified training programme, consisting of six two-hour sessions, to individuals (e.g. foster parents, personal advisers, etc.) working closely with care-experienced young people in supporting them in making informed decisions about entry into higher education and covering topics such as barriers to entering HE, application to HE, support for care-experienced young people, among others. Evaluation of the programme has shown that the participants feel better equipped to assist care-experienced young people in various issues concerning higher education (West Yorkshire, 2024). The University of York has put the following pre-entry support in place for care-experienced students (CES): access to a dedicated contact who supports a young person from the point of application throughout to their graduation. This contact person is responsible for providing information about study and courses, is a link between

the young person and the local authority, assists a visit to the university involving free return travel, accommodation and food for the young person and a guest; and is there to offer emotional and practical support throughout the studies<sup>16</sup>.

The Home Comforts programme was set up by the Sussex Learning Network to support care-experienced students, particularly first-year students, who lacked wider family support. The programme ran from August 2020 to December 2020. The programme collaborated with higher education providers, local authorities, foster carers and care leaving teams to link care leavers with a local family during their time at higher education, thus providing them with an additional support network, particularly over holiday periods, which can be a lonely and difficult time for care-experienced students. Young people were identified by the university student support team and students who were interested in entering the programme were referred to the Home Comforts project leader, who assisted them in filling out a form that helped match them with a family and organised meetings between the young person and the foster family. The process was overseen by the Brighton & Hove Leaving Care Team. Evaluation of the pilot programme has shown that the initiative provided young people with valuable additional support while at university. An interagency support was also seen to develop close links between different organisations: the university, fostering services, and local authority in supporting care-experienced young people<sup>17</sup>.

# 2.2.4 Support for care-experienced youth accessing higher education in Ireland

In Ireland, the number of students studying at higher education institutions has steadily increased over time, including those with lower socio-economic backgrounds and disabilities (Erskine and Harmon, 2023). However, according to the information from the Higher Education Authority (HEA), the progression rates for new entrants following into the academic year show that non-progression rates are lower amongst disadvantaged<sup>18</sup>.

Over the years, there has been a drive to encourage young people from disadvantaged and underrepresented backgrounds to access higher education. These young people can apply via the Higher Education Access Route (HEAR), even when they don't have sufficient Leaving Certificate points to access the course of their choice. Students who have been accepted have access to various college-based supports including information about grants and scholarships, study skills and exam preparation, social supports and mentorships, and an orientation programme. Applications can also be made through the Disability Access Route to

<sup>&</sup>lt;sup>16</sup> For further information, see www.york.ac.uk/students/support/care-experienced.

<sup>&</sup>lt;sup>17</sup> For further information, see www.sussexlearningnetwork.org.uk/uploads/case-studies/21%2004%2028%20home-comforts.pdf.

<sup>&</sup>lt;sup>18</sup> For further information, see https://hea.ie/2025/04/09/latest-student-progression-data-reveals-vast-majority-of-new-entrants-progress-to-the-following-academic-

year/#:~:text=Almost%2090%25%20of%202022%2F23%20new%20entrants%20progress%20to,by%20the%20Higher%20Ed ucation%20Authority%20a.

Education (DARE) programme, which is meant for young people with disabilities.

Until recently, only limited data has been available on how many care-experienced young people access higher education in Ireland (Brady et al., 2019a). Some information has been available via HEAR, although this is based on self-identification and may not reveal the actual numbers of these young people. In recent years, data from the Central Statistics Office (CSO) provide insights into educational attendance and attainment of children in care, reflecting practices in data collection in other jurisdictions such as the UK. The data from the CSO shows that the total proportion of young people 18–23 years of age enrolled in further and higher education was broadly similar for care-experienced young people and their non-care-experienced peers. However, the former were more likely to attend further education courses than higher education courses than the latter. In order to support care leavers in enrolling in higher education, they are now included in the National Access Plan for Higher Education for the 2022–2026.

Financial aftercare support by Tusla is based on assessment of need. Eligible care leavers may receive an aftercare allowance of €300 per week. This is available to care leavers up to the age of 23 if they are enrolled in accredited education or training programmes. The aftercare allowance may include the aftercare grant, which is a once off payment. While Tusla also covers PLC course fees, the government contribution fees are covered by the medical card (Tusla, 2017).<sup>19</sup> In 2023, additional funding was put in place to pay for the accommodation costs for care leavers and members of the Traveller and Roma communities to access higher education. Some funding is also available through the Tusla Bursary Scheme, which is aimed at assisting with higher education course fees. The Student Grant Scheme (SUSI) enables care-experienced students to apply for funding under different categories that are used to decide income considered when calculating income for grant purposes: Dependent Student, Mature Dependent Student or an Independent Student.<sup>20</sup> Overseas courses may also be considered, if no similar course is available in Ireland. However, the cost may not exceed the aftercare allowance available when studying in Ireland (Tusla, 2017).

## 2.3 CONCLUSIONS

This chapter has looked at international initiatives and programmes devised to support care-experienced children and young people. As outlined in Chapter 1, this group has been found to have lower levels of educational achievement compared to their peers who have never been in care. Much of this can be explained by disrupted care placements, personal circumstances, mental health issues, frequent changing of schools and insufficient support and guidance. Internationally and in Ireland, there has been growing policy focus on disadvantaged or

<sup>&</sup>lt;sup>19</sup> Tusla (2017). *Financial support in aftercare*, Dublin: Tusla.

<sup>&</sup>lt;sup>20</sup> It is now recognised that not all care leavers wish to progress into higher education. Considering this, an Apprenticeship bursary scheme is now also available for this group of young people.

underrepresented students, including care-experienced students, accessing higher education. However, the aspirations to enter higher education and progress through a degree programme may be influenced by their experiences and outcomes while still at school. This chapter has shown that additional support at school (Teach Your Children Well, Success4Life) can improve educational outcomes of care-experienced children and young people and develop their aspirations (Springboard programme). For example, children in long-term foster care who participated in the Teach Your Children Well programme experienced a significant increase in standard scores on reading, decoding and spelling skills. In the UK, care-experienced young people who participated in the Success4Life programme showed an increased desire stay in education post-16 and to apply to university. Availability of relevant information and guidance throughout compulsory education is important in encouraging care-experienced students to consider post-school education. Furthermore, the support needs to be holistic, as careexperienced children and young people are not a homogenous group and can experience different challenges, and some may have complex needs. The recent Take Care resource and accompanying guide developed for Irish schools (see EPIC, 2025) play an important role in informing teachers and principals about the challenges faced by care-experienced children and countering negative stereotypes.

It is important to note that after finishing second-level schooling, care-experienced young people experience a double transition – leaving the care system and entering adulthood with some moving into post-school education. To support young people at this junction in their lives, consistent pre-care exit supports are important (Stand by Me). For example, care-experienced young people in Australia participating in the Stand by Me programme engaged more fully in their transitional and leaving care planning. It is equally important to improve stakeholder awareness of challenges involved in leaving care and enhance interagency collaboration in supporting care leavers (Care Leavers' Social Impact Bond).

Supporting entry into higher education can take many forms – it can involve initiatives and support developed by individual universities or collaboration between various higher education providers and other agencies. An important element in these initiatives is access to a dedicated person that a young care leaver can contact and who can support them throughout their studies with advice and information, and act as a link between the young person and relevant agencies. Furthermore, having access to an additional support network during holidays (Home Comforts programme) could enhance a young person's experiences in third-level education.

## **CHAPTER 3**

# **Results of the study: Care placements and school experiences**

### 3.1 INTRODUCTION

As discussed in Chapter 1 of this report, some children and young people are not in the care of their parents. The reasons for being in care are many and varied, depending on young person's circumstances and needs. While some young people have continuous and stable care placements, others may experience multiple placements during the time they are in the care system. Changes in care placements can sometimes also mean changing schools and getting used to new teachers and peers. Changes in the environment can be challenging for any young person, but particularly for those in the care system who can feel unsettled or dislocated due to multiple changes in their lives. These challenges can be compounded by poor mental health and wellbeing of some care-experienced children and young people (Parsons et al., 2024). How these and other disadvantaged young people settle in and fare in the new environment can shape their school engagement and post-school trajectories (Frostick et al., 2015).

This chapter starts by discussing the adjustment at school and how young people's care experience influenced this process. Section 3.3 explores communication between schools and social workers and the impact of this on young people. Section 3.4 explores the impact of mental health issues on school experiences, while Section 3.5 discusses the relationship between foster parents and care professionals. Section 3.6 provides conclusions.

#### 3.2 COPING WITH CHANGES – CARE PLACEMENTS AND SCHOOLS

In Ireland, children can be placed in different care environments: foster care (general or relative foster care) – which is the case for an overall majority (88%) of the cases; residential care (general or special care), or other care placements. Half of all young people in care in January 2024 had a single care placement, with 9 per cent having more than five placements (see Chapter 1). The number of care placements of the young people participating in this study varied. While some had just one care placement change when at school, others had several. It was evident from the interviews that different care placements and moving into new communities and schools a number of times can leave young people feeling disorientated and isolated.

So, it was a bit tiring to adjust to different places and meeting different people like all the time. And then on top of that, to new surroundings as well. It just would be a lot easier to try to find a placement that would kind of work for the same school ... I'd be really recommending that like I would have been kept within the same **community or within the same area**. So that I wouldn't have to actually go to **a completely different new school** altogether, because it's hard enough to meet a new family and get used to their rules and their ways. (Sofia, FG5, interrupted third-level education (college))

Moving from (an urban centre) was a big change in itself and I just felt like **in the middle of nowhere**, all you seen was fields around it and it was like isolated. Like you **didn't know anyone**, the whole culture was completely different. And I found it really, especially going into sixth class when it's the very end, when a lot of them have been together for years. And like you know, it does have an impact – like you don't enjoy it as much and you don't really know anyone around and it's **very hard to go into the clicks** that they already have. (May, FG2, in thirdlevel education (college))

The **move in schools was a real sudden shift**. You're going from people you're now somewhat comfortable with, a school you've been in since you're a child to now basically living with strangers, being in a strange place in a strange school. (Theodore, FG1, in third-level education (university))

I was fostered far away from school. I was in (a rural environment) and then I was put in (an urban environment) so it was a bit of **a drastic change** ... So, like even moving to different foster homes, it was very hard to kind of manage things and there was a breakdown in one of my second foster placements. (Mila, FG2, in third-level education (college))

The participants in the consultation event with key stakeholders unveiled the unsettling nature of constant change for some young people that can be a result of placement shortages in the system, reflecting the findings of the recent report by the Child Law Project (CLP) (Corbett and Coulter, 2024) which referred to an acute shortage of appropriate care placements for children. This can mean that if there is a breakdown in the placement, it may not be possible for the young person to stay in the same school. Furthermore, finding another foster family can also prove to be a challenge. The stakeholders felt that as the quality of care varies, this can also have an impact on a young person.

**Placement shortages** is a huge issue in the system at the minute. If foster placement breaks down for a child for whatever reason, trying to get another foster placement **close to the school that they're in would be practically impossible**. Never mind trying to get a foster placement. ... So, in my view, there's a huge plethora of challenges. (Stakeholder consultation group)

The care-experienced young people in this study highlighted the importance of the quality of support from foster parents. While many had positive experiences, one person noted that while the foster family endeavoured to support his education,

the situation changed once the family had to address their own issues, with less time and attention given to the young person in their care.

Even though in my personal opinion **I was not in a great situation** where I was living, umm, the family had their own issues. You know that would affect the way ... I suppose as a foster child, I wouldn't have got the support that you needed maybe. They tried their best to help me with my education, but once their own personal problem started raising, I was kind of pushed out of it. There was **less support, there** was less time. (Ezra, FG4, interrupted third-level education (college))

The quality of placements emerged as an additional challenge in the consultation event with stakeholders. It was felt that care-experienced young people do not always receive the quality of care that they should, in terms of supporting their education and aspirations.

And the **quality of care** that they get, the supports that they might get in terms of supporting their education could be quite different and maybe not up to speed. (Stakeholder consultation group)

The consistency of placements, that's one thing. But actually, **the quality of placements is really important**. ... So, [the young people] don't have the knowledge if the family [has] that cultural or social capital that other kids have. So, **their aspirations, their expectations aren't even triggered in the first place**. (Stakeholder consultation group)

**Changes in placements**, obviously that's a huge one. **Changes in school**, I mean, if they can't be maintained in the same school while the placement is changing, that's gonna be a huge challenge. Changes in carers, **changes in the quality of care**, enormous challenges, I think. (Stakeholder consultation group)

In some cases, there can be a breakdown of a care placement, and a subsequent need to move into a new school. Having to change foster families can be stressful for a young person and can also impact on their level of connectedness and engagement at the new school.

I had to meet a new family and that was stressful enough, and that stress would have impacted how I would have met new people in school and [I would be] **worrying about meeting new people all the time** in my new schools. (Sofia, FG5, interrupted third-level education (college))

For some of these young adults, having experienced multiple placements can have an enduring impact on their lives and transition into independent living and college.

So, even like **moving to different foster homes, it was very hard to kind of manage things** and there was a breakdown in one of my foster placements. So that had a big impact on me coming into college. So, by the time I reached college, I would have been in like five foster homes already. (Mila, FG2, in third-level education (college))

Similar sentiments were expressed by Ezra, who had also experienced various moves during the time he was at school. Constant change had an adverse impact on his adjustment to routine and added to difficulty in making friends. Mental health issues further magnified the challenges of this young person.

I was in two different secondary schools, and I was in maybe four different primary schools in my time in foster care, and even before that, my primary schools before I was in foster care were always changing. So, I found it **very difficult to adapt to school** and to routine. I found it **hard to make friends**. And a lot of it did boil down to **mental health motivation**. And confusion, not knowing where to go in life, let alone after secondary school. (Ezra, FG4, interrupted third-level education (college))

According to some interviewees, age mattered, with the participants finding the adjustment process easier at a younger age.

So, in terms of the settling in process, while it was difficult, I also do acknowledge **it's good that it happened while young**. Because I think if that had happened to me in the teenage years, I would have found that a lot more difficult ... like I would have been a bit more set in my ways. (Theodore, FG1, in third-level education (university))

Another factor that impacted on the settling in process was the timing of the change of schools. For example, one of the school changes that Hazel experienced coincided with starting school with everybody else and hence not being perceived as a 'new' student, an experience she had in primary schools.

So, when I went into secondary school, I was kinda lucky because I started as everyone else was starting, so the primary school placements were quite difficult because I'd have to go in and the whole year had already been starting and I'd have to make new friends (Hazel, FG2, in third-level education (college))

In some cases, efforts were made to keep a young person in the same school, following a change in care placement. However, the distance between the new foster family and school the student attended resulted in the young person having to opt for remote learning.

For me, I had kind of a unique experience because I moved at the very beginning of my sixth year in secondary school. So, I actually didn't end up moving schools. I actually **ended up having to do online schooling**, which was kind of difficult because I just moved to a completely different area. It was like a two-and-half-hour drive to come to school. (Ava, FG1, in third-level education (college))

In other cases, care placement changes occurred very suddenly, causing a 'jarring effect' for the young person.

I was **informed of being moved like the Friday beforehand and I was moved on Monday**, so as soon as I got home from school we just took all our stuff and then went to the new place and I feel like that can be quite, I don't know, jarring to not know anything about the new people that you're gonna go and live with. (Luna, FG1, in third-level education (university))

The interviews highlighted the precarious nature of being in care as in many cases young people need to adapt to new care and school environments multiple times, thus making it challenging for young people to settle into both contexts.

# 3.3 LACK OF COMMUNICATION BETWEEN SCHOOLS AND SOCIAL WORKERS

Difficulties in settling in at school were also associated with limited or lack of communication between the social workers responsible for supporting the young person, and the schools. When Luna had to change schools, the situation was not handled very well by the target school, thus making the young person feel uncomfortable.

And she [deputy principal] was saying to me how she was very angry with the social workers, that she hadn't been informed that I'd be moving placement ... but I remember feeling at that time kind of conflicted because I really liked my social worker ... I think everybody who's in the child's life needs to be informed about the big changes that are happening. (Luna, FG1, in third-level education (university))

At times, lack of communication can also be due to abrupt changes in care placements, leaving little space for the social workers to inform the schools about the arrival of a new student. However, as seen in the case of Ava, not informing the student's old school about the placement change weeks after the student had left indicates that there is room for improvement in sharing information about the changes in students' lives.

In my secondary school **my teachers weren't informed**, and it ... was literally the previous Friday I had got told. And by the Tuesday I had moved, and it was a case that the teachers hadn't seen me, like **the teachers didn't even know where I was** ... I basically disappeared for like three weeks and then the school got a phone call from the social workers because they needed I think an address or something like that. ... There is definitely a need to inform the schools. (Ava, FG1, in thirdlevel education (college))

### 3.4 COPING WITH MENTAL HEALTH ISSUES

Some young people in this study spoke about dealing with mental health issues, which they considered added further challenges to their experiences at school and beyond. For Ava (FG1), having ADHD made it difficult to keep focused while at school, especially with limited support. Other young people also felt that more support was needed to assist young people with mental health issues, as lack of such support may result in them withdrawing from education.

I know a lot of people that are in care do suffer from **mental health** and stuff ... You know, like there just needs to be more support there, ... they **don't have that one person that they could go to** and ask them for ... So, like **a lot of them do drop out of education because of that**, and I wouldn't blame them for a drop. (May, FG2, in third-level education (college))

A lot of it did come down to a lot of moving, not a lot of guidance, not a lot of the support for my academics and my mental health there. There was pretty much **no help for mental health**. (Ezra, FG4, interrupted third-level education (college))

Stress, associated with turning 18 and having to make choices regarding continuing their education as well as having personal issues, changes in care settings and schools allied with lack of guidance and support with mental health were all seen to exacerbate mental health issues for some young people.

And when I turned 18, I got an aftercare worker, and I still did not know what I wanted to do. And I was pressured into going to college, even though I didn't want to. And I didn't know what I wanted to do, and it was **affecting my mental health** at the time, and it was making my anxiety really bad and other things were really bad at the time. (Willow, FG5, interrupted third-level education (college))

I had **personal issues. Depression**. You know, I felt **very unmotivated** for anything at school. I started slacking with homework. I started slacking with schoolwork. I was falling asleep in classes, and I just wasn't where I need to be. (Ezra, FG4, interrupted third-level education (college))

The consultation event with the stakeholders revealed issues around support in mental health, learning difficulties and disabilities for care-experienced young people. It was generally felt that difficulties accessing assessments, delayed diagnosis, and missed assessments due to placement changes can impact the experiences of these young people in the education system.

The **mental health** side of things. That, and the **learning difficulties**. You can't underestimate how many students need extra support with that stuff. And there are often waiting lists and there are often difficulties accessing educational assessments, et cetera, et cetera. And that can have a ripple effect then, going forward in their education system ... I would say the mental health and the learning support piece is under-resourced. (Stakeholder consultation group)

Disabilities, whether that would be specific learning difficulties, mild learning, intellectual, would all play a part in **challenges for young people who are in the education system**. (Stakeholder consultation group)

The mental health issues, the disabilities, the learning difficulties – all of those **can be under-resourced, underserved**. There can be difficulties in getting assessments. The, you know, assessments and issues can be missed because people are moving placements. [Young people are] not engaging, they're not attending. So, there's massive scope for **needs not being met**, shall we say. (Stakeholder consultation group)

The interviews and a consultation group discussion demonstrated that more support should be available for young people who struggle with their mental health. The views of the participants echo previous Irish research, which found that multiple placements tend to interfere with schooling and contribute to ongoing psychological difficulties of young people (McElvaney et al., 2013).

# 3.5 CARE PLANNING, RELATIONSHIPS WITH FOSTER PARENTS AND CARE PROFESSIONALS

Young people in care are required to have a care plan that details the actions needed to support their needs and development. Interviews with young people and the consultation discussion highlighted the importance of care planning, which was considered 'probably very hard to plan' but 'necessary' (Theodore, FG1, in third-level education (university)) The consultation discussion revealed that more could be done in the care planning phase to support young people. In particular, more emphasis could be placed on the personal and social development aspect that would support young people as they leave the care system.

I've been part of the journey of a number of young people through their care journey and in their care planning. And there's always an emphasis on, you know, the health education and the care planning. But I don't think there's enough [of] an emphasis on the personal social development, self-efficacy and confidence of young people in care as well. So that when they are coming to college, they feel like they belong there. That they can achieve, they will achieve, that they expect more and want more. And unfortunately, I don't think that selfefficacy or belief is always inherent in the young person with care experience, that they don't feel like they can achieve or should achieve. So, can we include that in their care planning? (Stakeholder consultation group) As discussed in Chapter 1, the relationship between the young person and their foster family is important for the outcomes of care-experienced children and young people. Young people in this study generally had positive experiences with their foster families. Those with supportive foster care placement seemed to cope better with settling in the school and having aspirations for the future. In some cases, foster parents were also a consistent support for the young person in their care, particularly in case of frequent changes of the social workers.

For me, my foster mom, she was on the parents committee team, so she kind of **knew all the teachers and communicated with them on a regular basis** about what is happening and what I was facing and I think that was really good. So, I had a really positive experience in that. Yeah. Always pushing to just do something with your life. Have like a plan, have the vision and go for it. Like whether that be doing like an apprenticeship, going to work or getting education, just do something that would benefit your life (Eliana, FG5, interrupted third-level education (college))

I had **a good relationship with my foster parents**, so it was kinda like they were pushing me to do well in school, but then I had **a different social worker after nearly every six months**, and they didn't understand my struggles with school and my learning disabilities. So, they were kind of like pushing me beyond what I could actually do. My foster parents were trying to explain to them, but then my social worker would change and it would be the whole thing again. So, they kind of just got fed up. (Willow, FG5, interrupted third-level education (college))

However, it was acknowledged by some young people that not all foster families were supportive of their educational participation and engagement.

Again, different families were different. They had different attitudes. **Some didn't really care if I came home and finished the homework**. I did make sure I had it done, kind of towards the end of my placement years and education was key. (Sofia, FG5, interrupted third-level education (college))

The interviews also showed that good communication between the social worker and school principal supported a smooth transition for the young person. In cases where a young person experienced a number of care placements, the social worker was the one constant adult in their lives.

Well, when I went to the different primary schools, my **social worker had already kind of sorted it out with the principal**. They knew that they were gonna kind of do it slowly, so that I wasn't just thrown into the school the first day. Like I was prepared. It was good that way, that they did that transition for me. They helped me move in more, but then kind of as you get older, it's different. (Hazel, FG2, in third-level education (college)) I mean, I've had loads of different families, you know, and they all work different in every kind of way. I had **my support worker**. She was a great support for me and that **one person that I needed all the time**. She was there. (Sofia, FG5, interrupted third-level education (college))

However, not all young people had positive experiences with their social workers. Some did not find their social workers very helpful or had minimal contact with them.

So, like **they weren't very helpful** and it was kind of like a whole social work situation, because I just felt like I had no one to talk to, that would actually listen. (Mila, FG2, in third-level education (college))

Yeah, maybe the last three-and-a-bit years of my time in foster care, **I** wasn't getting a lot of visits from my social worker. I wasn't getting a lot of, you know, support from the outside. (Ezra, FG4, interrupted third-level education (college))

Willow (FG5, interrupted third-level education (college)), who had learning difficulties, felt that having one constant social worker would have benefitted her more: 'Like growing up, to have a social worker that they can connect with, not to change all the time.'

Young people noted that it was preferable to have support from the social worker 'behind the scene'. One young person noted that having social workers coming to school is likely to draw unnecessary attention to them, thus making it more difficult to adjust to the school and peer group.

I don't think social workers should be going there, to school, unless it's like, really serious, because automatically you're being talked about. They'll see someone walking. And if you're right beside them, they're obviously gonna put two and two together and know about it. So, I just think **they should refrain from going into the schools as much as they can**. (May, FG3, in third-level education (college))

The turnover of social workers meant that it was difficult to develop a bond with them.

Within the last ten years of me being in care, from age eight to 18, I had about I think **one social worker every year**. So, **I wasn't really attached to them.** I was actually blessed to have very committed foster parents who I was with the whole entire time. (Eliana, FG5, interrupted third-level education (college))

The young people in this study had received help from various people at school, such as special needs assistants (SNAs), guidance counsellors and teachers. Getting to know the young person, they were able to provide appropriate support.

I had **an SNA** in school, so that was very beneficial for me, and I was able to kind of say: oh can I leave the class for a bit to take a break if I needed because I would get very ... I had a lot of anxiety as a child, so I have to, you know, step out and leave the classroom. And they were OK with that. (Hazel, FG2, in third-level education (college))

I think they were kind of guidance counsellors or, you know, just **school support**. They did help me. They gave advice. I started going to the lunches after school, before, after school studies, I started eating there in the morning. Because I was just very unmotivated with a lot of stuff ... I do remember there were at least three teachers who I did feel comfortable talking to about, maybe not exactly what was going on, but that I was in a difficult situation. (Ezra, FG4, interrupted third-level education (college))

Usually, if you try or even though it's hard to do so, you may find a very good adult to support you. So, for example, that could be your guidance counsellor, your SNA. And for me, that would have been certain **teachers**, of subjects that I would have had been very good at and very interested in. Naturally, you may feel more inclined to speak with those teachers, you know. You have some sort of connection to fall back on. (Levi, FG2, in third-level education (university))

In some cases, young people who had developed a good relationship with a person at school, continue communication even after finishing school.

And I had **a guidance counsellor** in school. She was very helpful. It's funny because I still talk to her today and she always used to, like, help me out with stuff. Especially when it came to, let's say, problems at home. She was trying to help calm me down at times. So, I kind of felt a bit more like warm and comfortable when I was speaking to her. (Mila, FG2, in third-level education (college))

Another source of support was an EPIC advocate.

In terms of the adults who supported me while in care, I suppose, firstly, I think obviously my EPIC advocate would have been a huge support for my education throughout, which I think was very important because it just backs you up when you're speaking to social workers and it's a huge support. (Levi, FG2, in third-level education (university))

However, in some cases, schools may handle a young person's personal situation in a very insensitive way, as is demonstrated by Lily.

Like I know I didn't have the social worker coming in, but my **year head** knew [about a sensitive issue concerning the child]. Somebody had told her, and she went and said it to the whole lot, to school like, the whole year. She called everyone in from the year and spoke to everyone before she spoke to me. So, everybody ended up knowing through that. (Lily, FG3, in third-level education (college))

### 3.6 CONCLUSIONS

This chapter explored the factors that influence care-experienced young people's school adjustment. In line with international research on the topic, the results of this study indicate that young people's school engagement appears to be related to the number of care place moves that sometimes dovetail with the change of schools. Change of schools may also reflect care placement shortages in Ireland, whereby it may prove challenging to find another foster family near the school that the child attends, in case of placement breakdown. Multiple moves may have an adverse impact on a young person's ability to adjust to new routines and rules, their educational outcomes, ability to form interpersonal relationships and their wellbeing. While consistency of placements is important, so is the quality of the placements. Proactive and supportive foster parents play a key role in supporting the education of young people in their care and helping them in developing aspirations for their future.

In addition to the foster parents, social workers can play an important role in supporting care-experienced young people, including help with adjusting to a new school. High turnover of social workers that some young people in this study referred to may reflect the fact that the work of social workers is challenging, especially when they may be supporting several young people. This may contribute to communication between social workers and school personnel being sometimes insufficient. This is particularly problematic when a care-experienced young person has multiple and complex needs. Ineffective communication between schools, foster families and social workers could also reflect the turnover of the latter.

International research has highlighted the importance of one key adult in the lives of children growing up in adverse circumstances. At school level, various staff members were seen as a source of help and support, including teachers, guidance counsellors and SNAs. However, the results of this study suggest that mental health and learning support may currently be under-resourced, as shown in other recent research with second-level school students (Carroll et al., 2024).

In terms of the experiences of young people who were pursuing third-level education and those whose third-level education was interrupted, the findings show that there were no notable differences between the two groups. Individuals in both groups found different care placements and moving into new communities and schools challenging. Experience of mental health issues and personal challenges seem to have contributed to interrupted education for some young people, something that could be explored more in detail in future research. Other factors that may also have contributed to interrupted education included not having sufficient guidance and support at school and feeling reluctant to pursue third-level education. Future research could focus on a group of care-experienced young people who had not decided to pursue third-level education.

## **CHAPTER 4**

## Social experiences at school

#### 4.1 INTRODUCTION

Chapter 3 demonstrated that for children in care, their care placements and experiences at school are closely entwined. As young people spend much of their daily lives at school, their experiences are shaped by everyday interactions with teachers and their peers. For children and young people in care, their journeys through primary and second-level schools are often disrupted by multiple care placement and school changes. Such experiences are likely to have an impact on their social interactions at school and shape their attitudes towards school and school engagement. Some young people may be particularly sensitive to how they are being treated as a consequence of their care experience. Existing research has shown that good relationships with teachers, foster parents, carers and peers tends to make their school experiences more positive.

This chapter explores which factors impact on the social sphere of careexperienced young people. The chapter begins with a discussion about the need for sensitivity when providing support for care-experienced young people in schools. Section 4.3 focuses on student-teacher relationships and the impact of the nature of this relationship on young people. Section 4.4 discusses the quality of relationships with foster parents and care professionals, while Section 4.5 highlights the challenges many care-experienced young people face when making friends at school. Section 4.6 highlights some of the measures that young people have found helpful in settling in at school. Section 4.7 provides conclusions.

# 4.2 IDENTIFICATION OF CARE-EXPERIENCED CHILDREN AND YOUNG PEOPLE

For many young people in this study, liking or disliking school was associated with their experience of being in care and the attitudes and practices of staff and peers. While the interviewees acknowledged that it was good to know that the school staff cared about them, additional attention from teachers sometimes made them feel 'different' rather than a 'normal' student.

I didn't want to have any [additional attention]. I just **wanted to be normal as everyone else**. I didn't want them to try and say I couldn't do this or I couldn't do that because of the situation I had been in ... I didn't want to be 'the one' – 'Oh yeah, she's the one that's in foster care'. I wasn't ashamed about it, but it wasn't something I needed everyone to know. (Hazel, FG2, in third-level education (college))

While Hazel was not ashamed of her care status, other young people felt selfconscious, or even embarrassed about their circumstances. Levi felt that being a young teenager and self-conscious, being in care can influence social interactions at school and education in general.

You want to be **a normal child**. But I think you want to have a boundary, a healthy boundary, or maybe just kind of ... some sort of division between your care status and all of that kind of traumatic stuff that has happened to you, and then being a student and that normal part of life that every teenager goes through, going through education. I myself, while I was in care, **I was very embarrassed** of my care status. Of course, that isn't the case anymore, but I think when you're a young teenager and you're self-conscious and stuff like that and what has been said and how professionals react to you, I think that has a very profound impact on your experience and education. (Levi, FG2, in third-level education (university))

Similar points were made by stakeholders in the consultation group who felt that schools need to show more sensitivity around the identity of care-experienced young people at school.

Some school procedures can draw attention to the young people and may be embarrassing for them. In some cases, it may be that the staff at school may not be sufficiently trained in how to approach situations that may arise with careexperienced students.

An experience that I would have, trying to sign out of school, I found very embarrassing ... I had to go through three people to sign out of school, which, you know, was very stressful at the time. Having to stand for 40 minutes in the corridor trying to get X, Y and Z to line up together. So, you know, you want to be a normal child, and I think what it highlights is maybe the need for training of professionals that are working with children. (Levi, FG2, in third-level education (university))

While the participants in the stakeholder consultation agreed that more sensitivity is needed regarding the identity of the care-experienced child or young person and how schools should address this, the discussion also revealed the complexities involved, depending on the level of support the child needs. While some children may be in stable situations and experience very few issues, if any, others may have high-end needs.

There can be, I suppose, **issues around disclosure** whether a child or young person discloses that they're in care, particularly in third level, but I suppose in second level as well. This could be communicated to a principal, but maybe the whole staff mightn't be aware of the needs of the child. So that issue around the difficulty around knowing who's affected ... to be able to offer support and the level of support needed. And that's quite mixed, I suppose, from children in care, because you can have children that are in very stable environments and children that are in an absolute crisis. (Stakeholder consultation group) When asked whether teachers at school should be aware of the care background of young people, the interviewees' opinions varied. On the one hand, some young people thought that if school staff are made aware that a young person was in care, their attitude to them may change, they will be treated differently, and this will then be noticed by their peers. On the other hand, it was also felt that teachers should be aware of the challenges some children experience, in order to provide support for the young person.

After they are told, they kind of get a bit ... they **treat you differently**. **You don't want that**. You want to be treated like everybody else, in a sense, because when you're starting to be treated differently then it's noticed by your peers. It's noticed by your friends ... and then once that's noticed, that's when you start to get singled out ... I don't think [making the care background known is] necessary, unless there's issues with attending school, or there's issues with transport and getting to school and getting to different places. Unless there's issues like that, then I don't think that it's necessary whatsoever. (Lily, FG3, in third-level education (college)

Of course, [teachers] should maybe have a knowledge of the challenges and adversity you face as a care-experienced young person. But I would be careful of kind of [showing support] to an extent that will **lead to maybe special treatment**. (Levi, FG2, in third-level education (university))

As noted in Chapter 1, care-experienced children and young people are a heterogenous group and have different circumstances and views. While many young people wanted to separate their care experience from their school lives, others did not see awareness among teachers about their care background as an issue. However, there seemed to be a consensus among the interviewees that care-experienced young people should not be treated differently, so that it will be noticed by their peers.

## 4.3 RELATIONSHIPS WITH TEACHERS

Existing research has highlighted the important role teachers play in the experiences of young people and their school engagement (Johnston et al., 2021). The interviews with care-experienced young people revealed that overall, such relationships were good.

You can find teachers who, for example, if they know that you're in care, they will **really try their best to bring you up** and to help you catch up. (Levi, FG2, in third-level education (university))

I actually **got on with the teachers**, so I did, but there was always, you know, one or two that were a bit difficult. Well, you know, the majority of the teachers were nice and really helpful. (Isabella, FG1, in third-level education (college))

Young people felt that it was easier to connect with some teachers more than others. The interviewees noted that it was easier to form a closer relationship with those teachers whose classes they enjoyed. Receiving help and support from the teachers was greatly appreciated by the young people.

And **the majority of [teachers] were fantastic**. There was one or two teachers ... and I suppose that goes for everybody. Like there's always one or two, you know ... But like all I ever had to do was just ask for help ... So, they provided that for me, not a bother. (Sofia, FG5, interrupted third-level education (college))

Several young people highlighted the importance of teachers understanding their circumstances and challenges and being understanding and supportive. Some teachers had made every effort to keep the student motivated and addressed their difficulties with schoolwork.

They were **very understanding** and that **really made a massive impact on my education**. (Eliana, FG5, interrupted third-level education (college))

The main teachers, **they understood**. And they wouldn't exactly be hard on me. But they would say, at least pay attention, you know, don't be falling asleep during classes and take down the notes from the board, so that you can see what was going on. If you can't pay attention during class, even just simple things like doing my assignments online so I'm not writing that much. **They tried to give a small bit of a push**. (Ezra, FG4, interrupted third-level education (college))

In some cases, the interviewees had maintained contact with their favourite teacher even after finishing school.

Yeah, I also had a good teacher that used to be **very friendly with the students**, you know? And used to help us a lot. I also **have like a good bond with them**. And up to now, I have a contact with him. From time to time, [I'd let him] know what I'm doing [and he'd] help me. Like, how's college right now? He asked a lot about my life. You know, he was a good teacher. (Mateo, FG1, in third-level education (PLC))

Two of the interviewees had participated in the School Completion Programme (SCP) and they found the staff of SCP very supportive and encouraging.

It was just one teacher who basically, like **encouraged me to stay in school** because I was getting so sick and tired of sixth year and everything like that. I was at risk of becoming an early school leaver. She'd just take me out of class once a week and kinda encouraged me really. To do my best and all that. So yeah, **that programme [SCP] was a great support when I was in school.** (Sofia, FG5, interrupted third-level education (college)) I did a School Completion Programme, and **I had a teacher who I was pretty close with**. I had like one session with her every week, and we talked about things going on at home, things going on at school. If I needed any extra assistance and stuff like that, when I brought it up to her, she would just kind of notice the attendance was dropping and I'd be like, yeah, it's becoming difficult to attend school. (Camila, FG1, in third-level education (PLC))

Negative interaction with a teacher, particularly when happening in front of the whole class, sometimes resulted in a young person not wanting to go to school.

And with this teacher in particular, like – **she did pick on me** ... making very, you know, smart comments and just making it difficult for me and, sort of **making fun of me in front of the whole class** ... So, because of that like, I just didn't really want to go to school. You know, I felt embarrassed. I **dreaded going to her classes** every single day ... I ended up dropping the whole module altogether, because of how horrible she was, and I just didn't want to do it anymore. (Isabella, FG1, in third-level education (college))

Some young people felt that not all school staff were sensitive to their care background. Luna noted that curiosity of some teachers regarding the reasons for being in care was presented as an offer to support the young person, leaving her feeling that the teacher may have had an ulterior motive in offering support.

I just want to mention that I had really lovely teachers mostly ... So, when I actually went to school, first [the staff member] pulled me out of a few classes. Kind of sporadically, kind of saying that they wanted to support me, but it wasn't. It was really that **they just wanted to know why I was in care** ... and she **asked very intrusive questions** ... she was basically pulling that all out of me ... So, yeah, I think just more training for teachers in that sense would be good. (Luna, FG1, in thirdlevel education (university))

Luna argued that teacher training and continuous professional development needs to provide future teachers with necessary skills regarding how to interact with care-experienced children and young people.

Some staff members were also seen as having negative stereotypical views on children and young people in care. The principal in Isabella's school seemed to assume that all care-experienced young people can be 'trouble'. Isabella felt that the way she was spoken to was very insensitive and showed a lack of understanding and care. Like Luna, Isabella argued that teachers need to be more aware as to how to communicate with care-experienced young people and how to support them.

I'll never forget going to that school. I went into school like that Monday or something, and the principal came around to me and said: 'Well, now all that is in the past. All the bad stuff that has happened. Now I expect you to behave now in school'. You know, just being very like insensitive with the knowledge, as she knew what occurred ... You know, so definitely [teachers] need to be more informed ... Teachers need to know and just learn how to be very sensitive and how to approach children in care and just not label them all as troublemakers ... So, they need to be trained, they need to be more aware, they need to be a lot more sensitive. (Isabella, FG1, in third-level education (college))

Similar sentiment was expressed by Levi, who also felt that some teachers lack sufficient understanding of the challenges experienced by children in care and how these experiences, which are often out of their control, can impact on their school engagement and attendance. Subsequently, teachers may have low expectations of them as they assume that these young people have no interest in education.

... When you have young people who are care-experienced and they have experienced trauma, so maybe their level of attendance at school is very poor ... **that doesn't reflect their dedication at all to education**. I think it's just the chaotic environment that they are in at that moment. I would be concerned that maybe some teachers may put care-experienced students into that group of students that ... aren't interested in education. (Levi, FG2, in third-level education (university))

In addition to being better informed about what it means to grow up in care, and how this may impact on a young person's school attendance and school engagement, it was also felt that teachers need to be more trauma-informed, reflecting the experiences of some children and young people in care. 'I believe that being more trauma informed in the teaching profession will never be to your detriment.' (Theodore, FG1, in third-level education (university))

Similar points were made by stakeholders in the consultation group. The stakeholders argued that teachers in primary and secondary schools need to be aware of the challenges some care-experienced young people encounter. Some have experienced traumatic events that may have had an adverse impact on the young person. By knowing what approach to take with the young person who has had such an experience, teachers would be able to put in place appropriate support for the young person, to provide them with additional 'scaffolding'.

[Better preparation is needed] in primary and secondary schools where **teachers are not trauma informed**. I suppose the teacher at the top of the room may not always know that or know what they should do to support those children. The impact of trauma on children and young people, we know that for a child to be in care, there would most likely be numerous traumas in their life and they will **impact their future outcomes**. So, you know, that means that there is **that additional level of support that's required for those children or young people** because they're facing [different] challenges as everyone else. *It is much harder. So there needs to be more scaffold. (Stakeholder consultation group)* 

The teachers need to be aware that while having individual life challenges, careexperienced young people are 'a unique group' of people who also encounter similar difficulties moving through the education system. Dealing with these situations and supporting care-experienced young people needs to be done sensitively and with compassion.

[Teachers need to] know we're a unique group of people. Not everybody experiences life in care and a lot of people will not be able to relate to it or understand or, you know, have that feeling of being moved and you know, new people and new rules and new cousins, new aunts and just all that type of stuff ... They **need to be more informed and a lot more sensitive and understanding**. (Isabella, FG1, in thirdlevel education (college))

Previous research in Ireland and abroad has highlighted the importance of the teacher-student relationship in student engagement. Overall, the interviewees in this study had positive relationships with their teachers and found most of them helpful and supportive. However, some young people felt that some teachers picked on them and perceived them as potential 'troublemakers'. While young people felt that teachers should be more aware of the challenges experienced by children in care and have the necessary knowledge and skills to support them, some of the interviewees did not wish to be identified as care-experienced – thus revealing a tension in this area.

# 4.4 RELATIONSHIPS WITH FOSTER PARENTS AND CARE PROFESSIONALS

In addition to support from teachers and other staff at school, many young people also mentioned the importance of having supportive and caring foster parents, social workers and other care professionals in assisting them at school and beyond. For Ava, (FG1, in third-level education (college)) who was not in school and had lost a purpose in life, it was her foster parents who got her a job. Getting over a difficult period, Ava was thereafter able to pursue a third-level education course. In the same vein, Theodore's (FG1, in third-level education (university)) foster parents supported his education and made sure he had support from a social worker who was instrumental in ensuring his move between schools.

Levi (FG2, in third-level education (university)) identified EPIC advocates and mentors from sport and volunteering as an additional source of support. Some young people also got support from their relatives. 'He [a relative] has always been there to help and kind of push me towards, you know, future success.' (Ezra, FG4, interrupted third-level education (college))

I think **my sister** was my biggest support because [she would] always like push me to do things when I'd say that I couldn't because she knew that I could. She would always help me with my homework or subjects I was struggling with. (Willow, FG5, interrupted third-level education (college))

Young people participating in the study had varied experiences with their social workers. Some had the same dedicated social worker while moving through the education system, who was also their advocate. 'I had a support worker from the age of 13 or 12 up to 18, and the same one. She was really good. She gave me a voice.' (Sofia, FG5, interrupted third-level education (college)) For another young person, their social worker facilitated his move from one school to another. 'She helped the transition by not having a gap between the schools – even though it is a massive distance, and the social worker had to travel 27 kilometres to and from every morning. She did this till I started a new school which helped the consistency, which again meant that settling in process was a lot easier. (Theodore, FG1, in third-level education (university))

However, others felt that they had no trust in their social worker, especially if these individuals were also dealing with other people that the young person knew. 'I just had no trust [in her] whatsoever.' (May, FG3, in third-level education (college)) Some young people like Lily felt that some social workers are not particularly helpful and do not always provide the support the young person needs.

Those **jobs are meant to be for people who want to help kids like us** who are struggling. Not so that you can fill out paperwork and get your pay at the end of the week. That's not what the job is for. The job is for people who care, because if you don't care, how are you meant to work in a care facility for kids who need that care because they're not getting it elsewhere? (Lily, FG3, in third-level education (college))

The stakeholder consultation discussion highlighted the challenges involved in resources invested in the social work sector, often resulting in a frequent turnover of social workers and inability of the child to form a bond of trust with the person. Changes in social workers may also mean that there is a limited understanding of the circumstances of children they are assigned to. Furthermore, shortage of social workers may leave a care-experienced child without additional support.

... the **shortage in social workers** ... she [a participant] knew of one child recently who's age seven and has had seven different social workers. So, you know, the ripple effect of that. (Stakeholder consultation group)

There's a lot of change with social work I found when kids were under 18. So that **continuity was really a key**. (Stakeholder consultation group)

And then of course, the difficulty now is that **some children in care don't even have a social worker**, and there's just such a shortage in the system. And **retention is an enormous issue**. So, I think that's bound to have a ripple effect that somebody hasn't got an overview, particularly if there are changes in placement. If the child doesn't have a consistent social worker, how can their educational needs, never mind any other needs, be fully met and fully addressed? (Stakeholder consultation group)

High turnover and low retention of social workers may reflect the fact that the work of social workers is challenging, especially when supporting young people with a higher level of need. The social workers may also lack necessary knowledge about how to support the more complex cases. Furthermore, these individuals often need to fight for resources for the young people in their care.

If you're working in trauma spaces and very emotionally charged spaces, like **it's a hard role**. And they **don't always have the tools to do their job** either. Like, you know, the therapeutic placements, the residential placements, foster placements, the counselling services. And if you don't have the tools to do your job, I think that can be quite hard or difficult to be in that space, constantly fighting for resources or priorities. (Stakeholder consultation group)

The interviews highlighted the importance of support from foster parents and care professionals, such as social workers. Consistent support from them helps young people to maintain a routine in learning, and supports their engagement with education. While most young people had positive experiences with their foster parents and care professionals, the discussion with the stakeholders revealed that there is room for improvement, particularly regarding the number of social workers available and their skill levels.

### 4.5 RELATIONSHIPS WITH PEERS

International research has highlighted the importance of friendships for children and young people, including those in care. Many young people in this study found it difficult to make new friends, mostly because of a number of school changes that often coincided with new care placements. Many missed the friends they had left behind. Furthermore, the process can be more difficult for a more introverted person.

I dreaded going into school every day because I'm not a person who likes change or new things, so it was like, I have to learn new rules about the school or find my way around the school, new classrooms, new subjects, new people ... So, like **everyone had already, you know, made their clicks, made their friends from first year**, they've already known each other, already had their little groups, and then there was this like new girl coming in towards the end of the year that nobody knew. (Camila, FG1, in third-level education (PLC))

*I would miss my friends* from the city. Like the friends I made in primary school, and I knew that I had to change to a new school. But I was sad about it. (Nora, FG2, in third-level education (university))

For Mateo, making new friends was even more challenging as he had moved to Ireland from another country. Mateo also had to adjust to the new environment, as he had not been at school in his country of origin.

It was like **a little bit hard for me to make friends** and you know at the beginning. And it was like a new thing to me. Like I never have been in school also in my country before. (Mateo, FG1, in third-level education (PLC))

Some interviewees felt that the timing of school changes was important in settling in and making new friends. If changing schools coincided with the start of secondary school, which means a change of school in general, the careexperienced young people share their adjustment experience with their peers. Further difficulty may arise from the fact that the young person may not be certain how long they will stay in the new school. 'I didn't know how long I was going to stay in that placement, so I didn't even know if these friendships would last.' (Hazel, FG2, in third-level education (college))

The age at which people move schools can also be important with regards to the ease at which children connect with their peers. Nora felt that making friends was easier at a younger age, as the peers are more accepting. Whereas at an older age, peer relationships can be more complex. In Nora's case, being in care was not something that was talked about.

So, for me, **when I was young, making friends was fine** because as you're young, you're kind of oblivious to things ... but when you go into secondary school, then it's like people are aware, but they won't really acknowledge it – what they think. It was strange in that kind of way, and so I did make friends, but it was very obvious that they knew that something was not the norm. They would be my friend, and we'd have a ball and all of that, but we would never acknowledge that. (Nora, FG2, in third-level education (university))

Care-experienced young people may sometimes confront other issues when interacting with their peers and may or may not be included in peer group activities. Subsequent social isolation at school had an adverse effect on Mila, who felt that her experiences with peers at secondary school may have contributed to some of her challenges.

When I was in secondary school, **I felt very segregated from everyone** else because it was like: 'Oh, she's different'. Nobody knew I was in care, but it was just like for other reasons. So, I feel like secondary school partially contributed to some of my issues. (Mila, FG2, in thirdlevel education (college))

Being in residential care also meant that the young person needed to abide by the rules established by the institution. While it was possible to have friends visiting, living in a different setting and following a set of rules may leave a young person feeling embarrassed.

Yeah, I was allowed to bring people over, but **how do you explain it to that person**? They might know about care, but they're coming into this kind of house and they're like: 'Oh, so you live here and you have to follow all these kind of weird rules and stuff', and that in that sense I did find it a bit embarrassing, like trying to explain home situations to people. (Hazel, FG2, in third-level education (college))

A few young people had negative experiences at school due to their care background. Information that should have been private was shared by some students, who knew about the young person's care background.

I had actually a lot of [relatives] going to the school and they basically made my life torture, and they told everyone what was going on and things that I would never have said to anyone ... it was like **everyone knew my personal business and I hadn't told them**. So that was really bad. (Ava, FG1, in third-level education (college))

In another case, the young person was made to share his care status with the rest of the school community, which resulted in him being bullied by other children. This experience made Levi very reluctant about disclosing his care status again.

... This person was putting a lot of pressure on me to share my care status with the school community ... Do you know what it did? I faced bullying within my primary school, **so I kind of had that in the back of my mind and I just didn't want that getting out** [when moving to a new school]. (Levi, FG2, in third-level education (university))

Ezra had difficulties in connecting with peers and making friends at the start of primary school, which had a negative impact on his confidence. The situation deteriorated later on with him being bullied until he finished his primary school.

You know, I've tried to make a bond, but most of the time that was, you know, just to no avail. I know I wouldn't make friends out of it. I wouldn't be invited places ... And then when fifth class hit, I was the new kid. I got bullied. And then for pretty much the whole time from fifth class to when I finished school, I was targeted. I was bullied and I was made fun of. (Ezra, FG4, interrupted third-level education (college))

For Ezra, things were made worse by the fact that he was not 'up to date' with technology or social media and found it difficult to contribute to conversations with peers.

I really struggled to know what to talk about. I was never up to date with anything online and another thing – technology killed me ... I would be behind on trends. **I'd be behind on this, that and the other**. (Ezra, FG4, interrupted third-level education (college))

Background-based bullying was also experienced by May (FG3, in third-level education (college)) after details about her circumstances were discovered by

another student and made public. In other cases, bullying was not related to the young person's care background, but other factors. Lily (FG3, in third-level education (college)), a victim of bullying, noted that while there was a lot of bullying happening at the school she attended, nothing was done about it.

### 4.6 SUPPORT IN THE SOCIAL SPHERE

Having a designated support system in place at school level was seen to offer support for a young person new to the school. Having left school for a while, on her return, Camila was paired up with another student who helped her to settle in again. The buddy system was in place for different groups of students: first- and second-year students as well as transfer students. In this instance, the buddy student also had care experience and could relate to the concerns that Camila had.

And I found that extremely helpful when I came back to my old school, because obviously I'd left for a few months and things were different from like, you know, when I left and I came back and I didn't kind of really know where to start myself. I didn't really know what to do **and** I was set up with the buddy and they were extremely lovely catching me up on, you know, all the things that are different or new or the same in the school. (Camila, FG1, in third-level education (PLC))

A similar system was in place in the school that Eliana attended. However, the buddy system was only available for first-year students, who were paired up with their sixth-year counterparts. In other cases, such support systems were not available, and young people felt that they were 'thrown into the deep end'.

Not in any of my schools anyway that I went to. There was none of that. You were kinda just like **thrown in there and hope for the best**. I mean, I think that [buddy system] would be a great idea or even like, you know, encouraging the new person to, like, take part in some sort of sport or some after-school group or something like that to get to know everybody else. (Sofia, FG5, interrupted third-level education (college))

The importance of extracurricular activities in making friends was also highlighted by Theodore, Ava and Eliana, who felt that such activities can help young people to adjust to the new school environment. However, participation in such activities depends on what is provided by the school or available in the area. While some students could avail of various activities in their locality, in other cases the options were limited, especially in more remote rural areas.

Yeah, there was a lot ... we had like debating, we had music, we had basketball. **There was a lot to do** to make different types of friends. (Eliana, FG5, interrupted third-level education (college))

I think extracurricular activities at school or externally would really be beneficial. And the extracurricular activities, table tennis or football, **definitely would have helped if that had been available**, if I had done that basically in primary school. That would be my take on that. (Theodore, FG1, in third-level education (university))

I think it's kind of important that when you're moving placements especially **that you do keep on your extracurriculars** because I wasn't able to keep up with my extracurricular activities because they were not available in the area. It wasn't a case that we didn't try. It was literally that I was gonna have to drive like 45 minutes or something like that because it was in a very rural area. So that's kind of hard as well. (Ava, FG1, in third-level education (college))

#### 4.7 CONCLUSION

This chapter has demonstrated that care-experienced young people want to be treated as 'normal students' and not be 'singled out'. Many wish to separate their care experience from their school lives. However, their personal challenges often mean that they require additional support from staff at school. Teachers can have a transformative role in the lives of students. Their support is particularly important for care-experienced young people who often have disrupted lives in terms of care placements and change of schools. Students may find it easier to ask for support if a teacher shows genuine interest in their lives. At times, a close and trusting relationship with a teacher or another staff member can make a difference in whether a student becomes an early school leaver or not. Such student-teacher bonds can sometimes last beyond the end of the second-level school. While support from individual teachers is important, participation in programmes that address early school leaving add another scaffolding around the student. On the other hand, negative student-teacher interaction, insensitivity, stereotypical views on care-experienced children and youth and lost trust can adversely impact a student. Initial teacher education and continuous professional development should provide teachers with necessary tools for supporting care-experienced students in an appropriate and sensitive way. Better understanding is needed about challenges faced by these students, many of whom may have had experienced trauma, which in turn, has impacted their school attendance and engagement.

While teachers play an important role in the school experiences of young people, the support of foster parents and care professionals cannot be underestimated. Both are instrumental in ensuring consistency and routine in education. However, inadequate care by foster families and social workers, often due to limited resources, breakdown of care placements and turnover of social workers can have an adverse impact on the outcomes of care-experienced young people.

The experiences of young people are also impacted by their relationships with peers. For some, several school changes result in disrupted friendships and difficulties in connecting with new people. In some cases, difficulties in the social sphere can last throughout their school career. The situation can be more difficult if a young person becomes a victim of bullying.

In order to support social integration of students, schools can put in place specific support structures such as a buddy system and encourage care-experienced young people to participate in extracurricular activities with other students, thus creating common experiences that are useful in building new friendships. The experiences of the two groups of students – 'in third-level education' and 'interrupted third-level education' did not vary much. Both groups had both positive and negative experiences with their teachers, social workers and peers.

# **CHAPTER 5**

## Academic experiences at school

## 5.1 INTRODUCTION

International research has shown that care-experienced children and young people often experience various disruptions and high mobility regarding care and school placements as well as adversity and trauma, which tend to adversely affect their school engagement, attendance and educational outcomes (Brownell et al., 2015; Herbers et al., 2013). Attitudes and expectations of teachers and caregivers also play a role in how these young people fare in the education system (Mannay et al., 2017a, 2017b). Similar issues are found in the Irish context (Darmody et al., 2013; McElvaney et al., 2013).

This chapter considers the academic domain in the lives of care-experienced young people, with a focus on primary and second-level schools. The chapter starts with discussing young people's attitudes towards school and school attendance. Section 5.3 focuses on expectations, programmes and academic outcomes, then moving on to identifying sources of additional help with learning in Section 5.4. The discussion then moves on to outcomes of state exams and related challenges in Section 5.5. Section 5.6 provides concluding comments on issues addressed in the chapter.

## 5.2 ATTITUDES TOWARDS SCHOOL AND ATTENDANCE

Interviews with the care-experienced young people showed that their attitude towards school varied. While most felt positive about school, for others, their attitude was shaped by several intertwined factors. For Ezra, not having sufficient support regarding study and mental health as well as the number of times he had to move, resulted in him being very unmotivated to engage with schoolwork. 'You know, I felt very unmotivated for anything at school.' (Ezra, FG4, interrupted third-level education (college))

Sofia, who had learning difficulties, found school difficult. While she was good at some subjects, she really struggled with others.

No, I hated school. I kind of classed it as a prison sentence to be honest. I really just did not like it. I suppose certain things I wanted to learn, like I loved art. Maths was an absolute nightmare. I still struggle with numbers. And Irish wasn't that great. I was good enough at French, but I think in school we were made learn a lot of things that we just didn't really need to know, because I know I learned a few things that I don't use today and in work, and that was the hardest part. (Sofia, FG5, interrupted third-level education (college))

While some young people experienced challenges in an academic sphere, they

enjoyed being involved in social activities at school. For example, Eliana, who had learning difficulties, found learning hard but liked meeting new people and taking part in after-school activities. Not being good at learning had an adverse effect on Eliana's self-esteem.

Yeah, I didn't like the academic side of it, but I did enjoy the social activities. So, I would do like debating after school. I find learning a bit challenging ... And I did enjoy like, you know, meeting people, talking to people. And there's like a science programme after school, which I absolutely loved ... but then the learning ... I just wasn't big on it ... I think that I went to a school where all the girls were like really smart and that pressure of not being good enough. I just [thought] – you guys can be the smart ones. I'll just sit back and watch. (Eliana, FG5, interrupted third-level education (college))

Some young people in this study had issues with school attendance. Discussion with them indicated that poor attendance can relate to a number of factors in young people's lives. For Ava, attendance was impacted by her health issues. However, she tried to overcome her difficulties and stay on at school.

I was in for something like 25 days of my Leaving Certificate, so that wasn't ideal. I have ADHD, so it's just like I was trying to keep myself focused. It was just so difficult and to be honest, there were points where I was just like, what is the point of doing this? (Ava, FG1, in third-level education (college))

Camila had had a number of care placements, which interrupted her attendance. Constantly moving from one school to another left her feeling frustrated and unmotivated. Camila also felt as if she had lost control over her life, as all the decisions were made about her without her input.

*I had a really big issue attending school*. Honestly, it wasn't my fault because it got to a point where, like I was moving placements so much that ... as soon as I settled, and then I started going back to school, I'd have to move again. And I got really frustrated at one point, because I thought like, I wasn't being kept in the loop of what decisions were being made about my own life. (Camila, FG1, in third-level education (PLC))

Camila felt that for care-experienced young people, school can be one constant thing in their lives and once settled in, moving to a new setting can be unsettling and disorienting.

... For a really long time, no matter where I moved, I always stressed that I wanted to stay in the same school no matter how far the placement was ... because like **consistency was really important to me**. You don't get consistency in care, so school was the only consistent thing for me since being in care, so I really didn't want to move. (Camila, FG1, in third-level education (PLC)) Another young person felt that while school attendance may be poor for some care-experienced young people due to trauma or other issues, this does not necessarily reflect their overall attitude towards education or their dedication to their studies.

... When you have young people who are care-experienced and they're experiencing trauma, maybe their **level of attendance at school is very poor, but that doesn't reflect their dedication at all to education**. I don't think anybody isn't interested in education. Everybody is interested in education. (Levi, FG2, in third-level education (university))

While some interviewees had experienced issues with attendance, which often was beyond their control, in the case of May and Levi, their school experience was impacted by being placed in a separate class.

So, I was in and out the whole time and when I did go, I used to get put into a class where **I wasn't allowed in with the other students** because they were there full time. So, I was like held back quite a lot. (May, FG3, in third-level education (college))

So, I would be concerned that maybe some teachers may **put** care-experienced students into that group of students that ... aren't interested in education. (Levi, FG2, in third-level education (university))

### 5.3 EXPECTATIONS, PROGRAMMES AND ACADEMIC OUTCOMES

## 5.3.1 Expectations

The interviewees felt that adults in their lives sometimes had low expectations of them, particularly regarding academic achievement and outcomes. However, some interviewees argued that rather than having limited interest in education, care-experienced young people may not always get the help and assistance they need.

Kids are not expected to, I guess, enjoy or engage in education. But I think for the most part, it's because **they don't get the actual assistance that they need** in regard to the education. (Camila, FG1, in third-level education (PLC))

Going into care can have an adverse effect on a young person who may have been very studious before, but who subsequently encountered various challenges, including stereotypical negative attitudes towards care-experienced young people. Luna noted that the first hurdle was getting a place in a school. Timing of the school transfer also meant that she had missed several months of going to school. Luna felt that the school staff were initially apprehensive about her, but the situation changed once they were presented with her previous grades and project work.

I was, yeah, really studious. But then I was in care for four months before I was able to go to school ... But it was at least one month before the social workers were trying to get me into school, and the **school just would not give them an appointment** ... I think from the way the interactions went during the meeting and also how long it took them beforehand, it seemed to me that **they had a stereotype of what a child in care would be like** ... Is it that they did not want to deal with a foster child because they had a prejudice or an idea of what that would mean? And then when they actually met me and saw me pulling out all my grades and projects I was doing on the side, they were really happy to have me then and very eager. I was wondering then, why would it take so long? (Luna, FG1, in third-level education (university))

This was not an isolated case. Luna had witnessed young people also going through the same process whereby they had to wait a considerable amount of time before they secured a place in the school.

And I saw it then play out with some other kids that were in the same household in another placement and going into primary school, that **they were waiting for ages as well**. And because I was friends with someone whose parent was on the Board of Management of the school, I saw it first-hand that they **didn't want to deal with foster children**, because of what they expect ... the additional needs. So therefore, they try and delay it. I don't really ... understand the reason why, unless it's that they genuinely do need extra supports and they're trying to get those in place before they go to school. But I still also think education is a right. You know, sometimes teachers may be a little bit uncomfortable with maybe a rowdy student, but they can't block their access to education because of it. I don't think it's fair. (Luna, FG1, in third-level education (university))

#### 5.3.2 Programmes

The majority of young people in this study followed the traditional Leaving Certificate (LC) programme. Only a few interviewees had followed the Leaving Certificate Vocational Programme (LCVP)<sup>21</sup>. Regarding the Leaving Certificate Applied (LCA),<sup>22</sup> in some cases teachers in the school the young people attended had advised them to take this option, instead of the regular Leaving Certificate, thinking that this programme would be more suitable to them.

My teacher said, 'I think you should do LCA because I think **you're really gonna struggle**,' and I turned around and said, 'Look, listen, I'm going to do my Leaving Cert and that's that.' (Sofia, FG5, interrupted

<sup>&</sup>lt;sup>21</sup> The Leaving Certificate Vocational Programme (LCVP) is a two-year Leaving Certificate programme with a focus on self-directed learning, enterprise and preparation for working life.

<sup>&</sup>lt;sup>22</sup> The Leaving Certificate Applied (LCA) is a two-year Leaving Certificate programme with a focus on adult and working life. The course is organised into four half-year blocks or sessions and students are assessed in each session.

#### third-level education (college))

When asked why she had not considered taking the LCA, Sofia noted that she felt that she was able to sit for the regular Leaving Certificate. She also wanted to do the course that everybody else was doing.

I suppose I felt able to do my Leaving Cert. I wanted to do the same thing, more or less, as everybody else, and I didn't want to go into LCA because of somebody else having an opinion that I wouldn't be able for it when I felt I was. (Sofia, FG5, interrupted third-level education (college))

Resistance to opting for LCA was also associated with not wanting to be different from students taking the regular Leaving Certificate course. Eliana indicated that the LCA programme was not thought of as highly as the regular Leaving Certificate.

The school I went to, LCA was kind of labelled like 'slow' or you weren't good enough to do the normal [Leaving Certificate]. And even how it is very segregated, how they would assign you to the Applied. Even when the principal [was giving a talk about LC], it was [evident that there was a divide between LC and LCA] – like you were different people – you weren't the same. (Eliana, FG5, interrupted third-level education (college))

Eliana observed that negative perception was also associated with the Post Leaving Certificate (PLC) programme.

I think [it's] the same thing with LCA. [PLC] was seen as degrading, like why would you go there? **You're kind of stupid if you do that.** (Eliana, FG5, interrupted third-level education (college))

#### 5.3.3 Academic outcomes

Much of international and Irish research has shown that care-experienced students tend to have lower academic outcomes. Many care-experienced young people in this study had strived to do well at school.

For me, when I was at home, I was pulled out of school almost as like a punishment. I was really, really eager to go to school. I was very academic. I did ten subjects for my Leaving Certificate, and I was, yeah, really studious. (Luna, FG1, in third-level education (university))

Not surprisingly, care experience – particularly regarding undergoing multiple care placements – had an impact on the learning experience of the interviewees.

I wasn't like, incredibly happy, with my Leaving Cert results. Sometimes I could get a bit tired. When I was in foster care ... and was switched to different foster homes – five in the [space of] 18 months ... like **that really did impact my grades and everything**. (Mila, FG2, in third-level education (college))

Another young person had started off secondary school with good grades, but a

change in care placement, as well as the pressure of the exams, had an adverse effect on his learning.

I was ... struggling through the whole end of secondary school and I was very lost coming out of it. In second year, I was still doing very well. Towards the end of the year, I won two awards for maths and English. I was at the top of my game educationally ... Once I started heading in towards the third year, between the move and the stress of the exams, everything started caving in. (Ezra, FG4, interrupted thirdlevel education (college))

For this young person, participating in Transition Year during the COVID-19 pandemic, with less focus on learning, meant a loss of motivation.

I moved and as you know, COVID came along somewhere down the line. So, fourth year, Transition Year, we didn't really do much at all. It was kind of a blank year. So, for that year, there weren't a lot of assignments. There wasn't a lot of work. And I think that almost allowed me to just crash. I wasn't motivated. **There wasn't a lot to do, so I just kind of let myself fall back**. (Ezra, FG4, interrupted third-level education (college))

Gradual decline in academic achievement also applied to Sofia, who had performed well at the beginning of secondary school and got a lot of support from her foster family regarding her education. However, being moved in her final year at school affected her grades and motivation.

Academically, I had done quite [well] for my Junior Cert ... it was a family I was with ... I had a set routine to study and everything. As of my last two years, I would have gone really downhill with the Leaving Cert and everything. I kind of **lost a lot of interest in education because I had been moved again in my Leaving Certificate year**, so I didn't really see the point in doing my education. (Sofia, FG5, interrupted third-level education (college))

The opposite was true for Eliana, who initially struggled at school due to her learning difficulties, but ended up doing better, when appropriate help was put into place.

From first year to third year, I kind of struggled **because I have** dyslexia and there wasn't that much support. My foster parents had to really push to get supports in place for me. But then **once I got into** fourth and sixth year, I did pretty well. I did everything higher level, got 420 points in my Leaving Cert, so that was pretty a good year, I think. (Eliana, FG5, interrupted third-level education (college))

Multiple challenges that some care-experienced children face can be best illustrated by May's experience. A combination of care placement and school changes, bullying and mental health issues resulted in the young person dropping out of school for a while. However, with help from teachers, she was eventually able to re-engage and finish school.

And then when I moved into my foster home, I went into a different school ... I went into sixth class there. So automatically **I was behind to their teaching**. Even in the classroom, like everyone noticed that the teacher would have to sit beside me the whole time and give me a lot of attention ... I was getting bullied the whole time ... towards fifth year. At this point, I dropped out completely. I went back to fifth year, and I was like, I'm just gonna stay the two years, get through, and I did that with the help of a few teachers and the school. That gave me a lot of time to catch up on everything. So, I was lucky in that aspect that they actually gave me the time to go back in. Change my failure. (May, FG3, in third-level education (college))

Lily decided not to sit for the Leaving Certificate as she struggled with some subjects and dropped out of school. However, she continued her education elsewhere and enjoyed a more flexible approach to learning.

The school was not for me in the sense that you're graded on subjects that you're not ... If you're not good at it, you're still graded and then it's taken into account ... And then when it came to dropping out, when fourth year I got asked to go back – 'Would you like to go back to school?' And I said no, I was like, I'd rather go and try study outside of school, and that's what I did. And you know what? I found it more lively, more interesting and more inclusive ... there were no sad faces because somebody didn't get something right. If you had a problem, you were able to go up to the teacher and tell that problem to that teacher, and that teacher would go one-to-one with you, until you understood it. Like, that's what I loved about it. (Lily, FG3, in third-level education (college))

Taking fewer subjects was also a preferred option for Camila, who struggled at school due to health and other issues. For her, a PLC course was seen as more suitable.

So, I originally started off doing Leaving Cert, but I never actually finished it. I dropped out in fifth year. I was still in touch with my career guidance counsellor from the school, and we were just looking at ways to get me back into education because it was extremely [difficult] at the time. I was going through an ADHD assessment, so it was extremely difficult to manage home life and school life at the same time. So, I ended up doing PLC for a year and I found that extremely helpful because I was actually doing something that I had interest in, not doing eight or nine subjects when I only had an interest in one or two. (Camila, FG1, in third-level education (PLC))

Inflexibility of the education system regarding the needs of different types of learners was also highlighted by the stakeholder consultation group. It was acknowledged that once a young person has missed school, it is very difficult to catch up due to the inflexibility of the system. Even when different learning options are available, this can be problematic for some students.

I think the word that's coming to mind for me is the systems. That **the** systems are not responsive to the needs of the young people. You know, the Leaving Cert syllabus, the way that schools are set up, it can be very inflexible. So, if you're moving from one school to another and you're missing school and have gaps in your education, there's no way of like, making that up. You know, it's very hit and miss, I suppose. You can be lucky and maybe that chapter's being covered or revised again in a few weeks, or maybe not. I think with things like maths and science where you're building kind of blocks in education – this piece informs this piece. It can be really, really difficult to play catch-up when you've, you know, been in different settings or had gaps. So, I think the education system itself is very inflexible. (Stakeholder consultation group)

It is something that's there; there's a project, I believe, being done by TESS [Tusla Education Support Service] at the moment that there's an iScoil,<sup>23</sup> but it's not the traditional Leaving Certificate. It's QQI [Quality and Qualifications Ireland]<sup>24</sup> modules that you can do. I believe it's only up to level three. So Junior Cert level. But again, you need supports with a young person to be able to help them navigate that, get them, you know, logged in, et cetera. But **those two systems then don't meet**; you know, you can't do the Junior Cert and do the QQI. There's no integration we'll say. (Stakeholder consultation group)

#### 5.3.4 Additional help with learning

The interviewees were asked whether they had received any additional help in school. Overall, the care-experienced young people said that help was available when they needed it.

All the help they gave me ... **as much help as I needed**, and I even asked for more on top of this ... I had grinds for nearly every subject in my Leaving Cert ... And my free classes, I made sure that I found a teacher who was free that could help me to learn a bit more ... So, **I had enough support**. I suppose my attitude always was like, if you don't ask for help, then nothing's gonna change. (Sofia, FG5, interrupted third-level education (college))

I took my Junior Cert, and those **teachers that did know what was going on did help me** and did at least give some advice into, you know, just the easier part of the exam to complete so that I could fulfil my

<sup>&</sup>lt;sup>23</sup> iScoil is a non-profit online learning service that offers young people a pathway to learning, accreditation and progression.

<sup>&</sup>lt;sup>24</sup> Quality and Qualifications Ireland (QQI) is the state agency responsible for providing information on the qualifications included on the National Framework of Qualifications.

points ... And that did help me do at least relatively well in my Junior Cert. (Ezra, FG4, interrupted third-level education (college))

Lily felt that she had benefitted the most from one-to-one help, although this was provided for a very limited period of time.

Like one-to-one [help], when I did request it, it was very brief. I think if I had more one-to-one with them, I would have continued with chemistry and biology ... but with the Irish as well. (Lily, FG3, in thirdlevel education (college))

When asked about getting help from other students, the interviewees felt that they did not want the other students to know that they were struggling.

No, I kept to myself. **I wouldn't go near anyone to ask for help** ... because you wouldn't want anyone knowing that you're struggling. (May, FG3, in third-level education (college))

May's response may reflect a general attitude of teenagers who, being selfconscious, prefer not to let other students know that they may be struggling. However, the situation may be more acute for care-experienced young people who may not wish to draw attention to themselves.

Another young person was less concerned about the reaction of other students, noting that the care experience had taught her to look after herself. However, she also noted that not enough help was offered.

For me, it was not being afraid to ask somebody for help because of their reaction ... we were always like fight or flight, you know? **We're always kind of looking out for ourselves**, especially from day one. We always had to do our own things, so **asking somebody for help, it feels weird, it feels odd.** You're not used to it, because you've never been actually offered help before. Like there was no help actually offered, there could have been help offered. (Lily, FG3, in third-level education (college))

In addition to help available at school, in many cases the foster parents were also a good source of additional support.

It was very, very important, and **I was kind of made to work**, like I had no choice but to do it. Which was good, you know, I'm happy about that. (Sofia, FG5, interrupted third-level education (college))

I had a good relationship with my foster parents ... **they were pushing me to do well at school**, (Willow, FG5, interrupted third-level education (college))

#### 5.3.5 Outcomes of state exams

Young people in the study had varying degrees of success in their state exams. While most had successfully sat their Leaving Certificate exams, some had only managed to sit for their Junior Certificate exams. [I did] **mostly higher-level subjects for my Leaving Certificate**, but ... I took two ordinary maybe, feeling I could do well. (Theodore, FG1, in third-level education (university))

*I never did my Leaving Cert.* I did my Junior Certificate, then I did fourth year, my Transition Year. It was compulsory in my school, so there was no skipping it. So, fourth year was a bit of messing around and then after that I went [to an education centre] (Lily, FG3, in thirdlevel education (college))

Life circumstances of another young person meant that even though she performed well in the Junior Certificate, she had decided not to sit for the Leaving Certificate, much to the disappointment of the teachers in the school.

Yeah, I feel like I did really well in my Junior Cert. **The grades that I got were really good** as well. Like I remember when I did kinda bring up at the meeting with the school and my social worker to say that I was dropping out of school, like everyone was extremely upset and disappointed ... they had this expectation, and they were like, you know, you were one of our high hopes. (Camila, FG1, in third-level education (PLC))

Young people in this study who took the Leaving Certificate exam felt the stress and anxiety associated with it. In one case, this was compounded by the approach adopted by the school in highlighting the importance of the exam in future life chances of young people. Eliana felt that the pressure from the school to do well in the exams may have been counterproductive as it increased the anxiety levels in students and may have had an adverse effect on their exam performance.

We had an assembly every Monday with our principal who would give us this lecture on how important Leaving Cert is. Just **how like brutal those conversations were** ... I think at one point it felt just very degrading. It's like if you don't work hard, you could end up in a low-grade job having a worse life. All of this, every single Monday, the same conversation. So, we kind of knew [the Leaving Cert] was important. I think [the talk] made a lot of girls very nervous. I know three girls had an anxiety attack coming out of classes and that's not OK, you know? Mental health wise, I think it really destroyed a lot of the girls who would have gotten to do something greater but that kind of 'I'm not good enough' [thinking] stopped them from getting better results in their exams. (Eliana, FG5, interrupted thirdlevel education (college))

Sofia knew that she was going to take a PLC course and was not concerned about the points in the Leaving Certificate exam.

I'll be honest, like **I just passed my Leaving Cert** and I'm happy that I passed. I wasn't actually looking for points at all. I didn't have an aim because if I did, **I just would have stressed myself out too much**, so I

was ... Great, I'm gonna pass it and all I had to do was pass, to get into my course. I was gonna do a PLC and yeah, I didn't really have to worry about point systems or anything like that. (Sofia, FG5, interrupted third-level education (college))

However, this young person did not do that well on the course, because of her personal issues that affected her progression on the course.

The PLC was really good. Unfortunately, I didn't actually do quite well in this because of **my own issues, personal issues and that affected my results.** What I had done was really, really good. Like I got some fantastic experience out of it and it's probably something that I may go back into further down the line. (Sofia, FG5, interrupted third-level education (college))

At the time of COVID-19, students preparing to leave secondary school had the option of a calculated grade or a written exam. The option of calculated grade was available in all subjects. Isabella felt that if they had sat for the Leaving Certificate exam, they might have done better.

It was the year 2020 when COVID hit. So that's when we got the predicted grades. So, we had done, you know, the mocks, I think in January ... I can't even remember – and I was close to 400. So, I was happy with that. Like my course was only 300 points, so it didn't really matter as long as I got over 300. But when the predicted grades came out, I was actually quite disappointed, because one or two of the teachers put my max low, when my average was like a grade higher than that ... It just **felt like the teachers undermined me**. (Isabella, FG1, in third-level education (college))

Several young people noted that although they had endeavoured to do well in their Leaving Certificate exams, often what was going on in their lives at the time took precedence.

To be honest, **I just did not care. I had too much on**. I turned 18 in the July, so I had really too much on my plate. But like you finished your exams in June? I literally had a month to get my act together and kind of move and stuff for that, and that was really [tough]. (Ava, FG1, in third-level education (college))

I did [consider Leaving Certificate results important] but considering what was going on around me at that time and you know, the pressure of turning 18 and whatnot, and the stress to deal with that and **your attention being divided**, you know, between studies and everything else going on ... (Theodore, FG1, in third-level education (university))

While the young people were trying to cope with the immediate challenges associated with gearing up to the exam and leaving care, later on some of them regretted that they did not do better in the exams. Not getting enough points meant that they had limited access to some courses and professions.

I'm caring about it now because **I'm basically a lost student** and I'm applying for a lot of intern programmes and every single one of them asked for your points ... I didn't realise this, but if you got below 500, it actually takes you off the application page. So, you can't apply any further. They don't want anyone below 500 points, so that was a bit of a shocker for me. I didn't realise this at all. I kind of have to reach out to the smaller firms and hope they reply, because **I actually can't apply to the bigger firms. It's not possible**. (Ava, FG1, in third-level education (college))

Similar sentiment was expressed by Theodore, who felt that while he had a general idea of the importance of Leaving Certificate exams regarding entry into third-level education, his life situation and challenges were his immediate concern.

I think with what was going on, you have enough. I started the aftercare process at 16 and I suppose you have care plans there and you're told about the aftercare allowance and the importance of going on to college. I suppose at the time I did know the importance of the Leaving Certificate for getting into college. Now, whether or not I cared, considering everything else that was going on at the time, my answer will be no. I didn't particularly care that much ... Had I been settled as such, if I had known then what I know now, it definitely would have changed things. (Theodore, FG1, in third-level education (university))

While Sofia did not think the results of the Leaving Certificate mattered to her personally, she acknowledged the importance of the results for entry into third-level education.

I didn't really see the point of the whole thing. Yeah, to be honest, because what I'm doing now, like, they didn't even ask for my Leaving Cert. But yeah, I think, you know, it is somewhat important because **it's kind of one of those things in life where you just have to get this stage done**. And you know, we're in a country where Leaving Cert is necessary to get into third-level college or university. (Sofia, FG5, interrupted third-level education (college))

#### 5.4 CONCLUSIONS

Existing research has highlighted the role of personal circumstances of careexperienced young people's lives that impact their educational outcomes. This chapter has considered factors that influence young people's attitudes to school and their experiences. The interviews conducted for this study showed that young people's attitudes to school are often influenced by changes in the care placements and school changes, learning difficulties, insufficient support with schoolwork, not feeling as smart as other students, and mental health issues. Some of these issues were also associated with poor school attendance. It is important to note that poor school attendance and lower grades do not necessarily reflect lack of interest of a young person in education but is often influenced by circumstances beyond their control.

In line with international and Irish research, the interviews in this study showed that some adults in the lives of care-experienced students had low expectations of them. While young people can be very motivated, challenges associated with being in care often impacted their academic achievement. In some cases, lower expectations can be observed when teachers guide students toward specific second-level programmes, despite the resistance by the young person. Stereotypical negative attitudes about children in care can sometimes result in difficulties finding schools, due to care changes of the young person.

Having learning difficulties, insufficient help with learning and mental health issues shaped the attitude towards school for some people. Care placement changes may result in students missing school and finding it hard to catch up afterwards. Second-level school can be a particularly stressful time for young people due to the high-stakes Leaving Certificate exams. If this coincides with personal challenges at the time, and drawing up a care leaving plan, it can be a particularly difficult time for young people in care. Young people 'in third level' and 'interrupted third level' groups shared some similar experiences. In both groups, there were young people who endeavoured to do well at school. However, their aspirations were sometimes impacted by more immediate issues such as care and school placements, which also impacted on their school attendance. In both groups, there were young people whose foster parents supported their education. However, the 'interrupted third level' group had more people with learning difficulties and who experienced the stress associated with placement and school moves, and preparation for the state exams more acutely.

# **CHAPTER 6**

## Decision-making and advice about post-school pathways

#### 6.1 INTRODUCTION

Educational attainment is a key indicator of post-school outcomes such as improved life chances and labour market success. Educational experiences and attainment are strongly influenced by the support and guidance available in the education systems and schools. Much of the existing research has focused on postschool pathways of young people in general, and their decision-making process and planning. This process is informed by a number of factors including the young person's personal background as well as support and guidance available to the young person. Earlier work from the ESRI has shown that aspirations to enter third-level education emerge as early as junior cycle (McCoy et al., 2014).

This chapter discusses the support received by care-experienced young people when they were still at school and identifies different sources of advice they received. The chapter starts with discussing young people's views about the importance of education. Section 6.3 focuses on decision-making and advice received regarding post-school pathways.

#### 6.2 PLANNING FOR POST-SCHOOL PATHWAYS

Chapter 5 highlighted various challenges that care-experienced young people experience during their educational careers. While some did well at school, others struggled, mainly because of the frequent care placement changes and changing schools. However, the interviews with care-experienced young people showed that they understood the importance of education for further life chances in terms of entering the labour market, getting a stable income and being able to support one's family.

I think that most of us would understand **the value of education in getting you out of that adversity**. And that's how I always viewed education. That's where my dedication, commitment to education came from. I viewed **education as a lifeboat** to escape that kind of sinking ship is what I'd say. That education is very good in terms of getting a new job and stable income and being able to support a family and all of that stuff. And I would definitely say that **education is a priority for nearly everyone in care**. (Levi, FG2, in third-level education (university))

Despite understanding the importance of doing well in school, care-experienced young people face various challenges in engaging with schools, as many experience disrupted education, receive limited support in settling into new schools or experience mental health issues that make it more difficult for these young people

to fully engage with education, as discussed in earlier chapters of this report. However, persisting with learning is important for improving the situation for young people as they exit the care system and start independent life.

I think there are **certain barriers that prevent young people from engaging in education**, but I think regardless whether you're maybe in a residential centre and you have grown up in a residential centre all your life, or you have come into care later in your life, I think that **the emphasis on education is so important** because you see it as a way to get out of that adversity and more importantly, you want to have your own identity. (Levi, FG2, in third-level education (university))

Post-school planning can be a stressful time for all students, but even more so for care-experienced young people who are also approaching exit from the care system. There are distinct implications both for young people who have identified a preferred occupation and those who remain undecided about them and the consequences for post-school pathways. The advice and support from families, teachers and carers are salient in this decision-making process. Care-experienced young people are likely to find the process more challenging because for many, their educational careers have been disrupted by several care placement changes and changing schools.

#### 6.2.1 Teachers and other staff members

For most young people, their teachers provided help and advice in how to organise their studies and do well in the exams to acquire a sufficient number of points to ensure they had more options available to them when finishing school.

I took my Junior Cert and those **teachers that did know what was going on did help me and did at least give some advice** ... so that I could fulfil my points. (Ezra, FG4, interrupted third-level education (college))

I had a really good **teacher who used to talk to me about twice a day** and we were just looking at stuff to do in college and stuff like that. And she always pushed me to [pursue a profession] – **she told me that I could do it.** (Willow, FG5, interrupted third-level education (college))

I found **a lot of people influential, especially some of my teachers** ... they were very, very inspirational. (Lily, FG3, in third-level education (college))

However, some interviewees noted that if they did not get along with a certain teacher, this had an impact on developing aspirations for future and their learning. Input from some teachers also shaped their understanding of what post-school education may be like.

You really have to look at who's supporting you, but I also think **you need to look at who doesn't inspire you** and what impact they have on you. So, for example, you know you may have a teacher that you

don't get along with; I think for younger people, I would hope that that is all changing now, because you have younger teachers coming in and there's a higher level of education and social awareness. I felt that some people, and it may not have been their intention to do so, **discouraged my advancement in education**, if that makes sense. (Levi, FG2, in third-level education (university))

I think it's the **people who actually showed up for you** ... Those are the people you remember. But then there's also **people you remember who weren't very helpful**. (Hazel, FG2, in third-level education (college))

I remember my [subject] teacher in particular was saying that once you go to college, the lecturers don't care about you, like you know, **you're really just a number**, hence why I picked like the institutes instead of the actual universities. (Isabella, FG1, in third-level education (college))

In order to assist students in their post-school decision-making, school-based guidance counsellors advise young people about different options available to them. Young people in this study had varied views about the support they received from their guidance counsellors. Some of the interviewees had positive experiences and discussed the additional support received by them. For May, additional support from her guidance counsellor meant that she received encouragement in continuing her education even when she had a short spell of not being at school.

When I was in hospital at the time and [the career guidance teacher] came in to actually visit me, I remember her saying: you're going to get out with this [...] she was sitting there, trying to get me through. And **she was just like, you can do this. You can go on further**. And then when I got out of hospital and went into the school a few months later, she brought me in every Thursday or Friday for a meet for an hour and go through all the colleges and stuff with me to make sure I'd get in; and she did the DARE programmes and stuff with me. So, I was lucky she was there. (May, FG3, in third-level education (college))

On the other hand, some young people felt that they did not get sufficient advice and support from their guidance counsellor. For example, Eliana felt that she was discouraged by her guidance counsellor in pursuing a specific course and subsequent profession.

I had a guidance counsellor **who was just very negative**. Like if you wanted to do something that was not popular, she would immediately shut it down, like: 'Maybe not, that probably won't work with you' ... I think a lot of girls in that year also experienced that and ended up, you know, **turning away from the person that was supposed to help us**. (Eliana, FG5, interrupted third-level education (college))

Unwillingness of guidance counsellors to consider options preferred by a young person can have a negative impact on self-confidence on all young people, but can have a disproportionate effect for care-experienced young people who may rely more on the advice and support from guidance counsellors considering their lack of family support.

In one case, insufficient advice resulted in opting for a course without realising what this would mean for getting a job after finishing the course. It was partially due to the fact that this young person seemed to be uncertain as to what they wanted to do in future.

I didn't get support in actually figuring out what I wanted to do. So, I ended up doing a whole bachelor's degree [in a subject] ... And then I didn't really have anybody to talk to about it, or have any input in what that would actually mean for careers. I didn't think about what job I would get out of [a subject] degree. So yeah, I think all students should be able to have that kind of support. Even if it seems that they're decided ... I specifically didn't get [advice] because they said, oh, you have got your head screwed on. So, you're not going to do the aptitude testing. I think it could have been helpful. (Luna, FG1, in third-level education (university))

As discussed in previous chapters in this report, many care-experienced young people have to deal with multiple challenges during the time of deciding what to do after leaving school. Some of the interviews indicated that guidance counsellors were not always aware of a young person's life challenges or mental health issues and that the advice given is often steeped in a 'standard' practice that students follow. Camila felt that her personal circumstances made it very difficult to focus on post-school decision-making. The situation became more stressful by a constant push from the guidance counsellor to sit for the Leaving Certificate exam. Camila felt that it would have been helpful if the guidance counsellor had a better understanding of her situation and suggested alternative ways in how to achieve her goal.

I had quite a few conversations with my guidance counsellor about what I wanted to do ... I guess it would have helped a lot if the people who were advising me or my career **were a little bit more open to what I needed**, instead of what was the standard or the norm. You know, believing in the norm: you do your Leaving Cert, and you go to college. Basically, your points. But there's no real actual routes for people who can't do the normal Leaving Cert or can't do the normal school life to, you know, achieve their dreams or develop their career. So, I definitely **wished that the people advising me would have been a lot more informed. It was me who found the route into becoming** a [profession] without the Leaving Cert, and it was extremely difficult to find. So, I guess I didn't feel supported. I felt like **if I wasn't doing what was expected of me that things wouldn't really work out**.

#### (Camila, FG1, in third-level education (PLC))

Ezra noted that it is a good idea to pick a course that one can get a good career out of. He felt that at the time when he was considering his post-school options, such options were limited: 'I didn't have that much to look at, I suppose'. With little advice received from the guidance counsellor, this young person was undecided about what course to opt for. Furthermore, his personal circumstances made it difficult for him to fully focus on post-school education. This young person benefitted from having some time after secondary school to fully understand what course he wanted pursue in future.

It was **difficult for me when I was coming out of secondary** school to put my head down and, you know, do the research or do the study or whatever. And it got very confusing. It was very difficult to deal with, and I ended up coming out of college and starting work. But now since I've had a bit of time, I'm looking at college options, something that I can really go for. I'm ready to be able to put my head down, to study everything that needs to be studied. Do the assignments the way they need to be done without the distractions or **confusion of not knowing what I wanted to do while doing a course**. (Ezra, FG4, interrupted third-level education (college))

Despite the input from guidance counsellors, which varied, many people in this study felt that they were undecided about what course to pursue after finishing school. Subsequently, the young people put down a number of different options on their application form. Theodore noted that he was still uncertain about the career he wanted to pursue even after pursuing an undergraduate degree in a certain field, and wished that he had received more advice while still at school.

I suppose when I was choosing mine, it was more a case of putting down as many things as you have a remote interest in as possible. And you're waiting for your points and kind of hoping something sticks. I did the bachelor's [in a certain field] and by the end of that, I was still like, do I wanna do this? You know what I mean? I had a remote interest in the area, but then I went on to do a course [in a different field] ... and now I'm in a master's [in yet another area], so **if I had got support around the time**, I probably wouldn't have had to go through five or six years of education to get where I am now. I don't regret it. However, it would have probably streamlined things a lot more if I had got a lot more support when I was at that age. I just simply didn't get it. (Theodore, FG1, in third-level education (university))

One person felt that in a way, care-experienced young people can have similar experiences as other young people, as there are always some teachers or other staff who may not be that encouraging. And this is not necessarily associated to being in care but with other factors. 'You always have those teachers or guidance counsellors who are kind of discouraging, like they might tell you: no, realistically,

you're not gonna get this choice that you want.' (Mila, FG2, in third-level education (college))

#### 6.2.2 Support from social workers

International research has highlighted the importance of care and social workers in supporting the educational careers of care-experienced children and young people (Crawford et al., 2011). Young people in this study had varied experiences with their social workers. In some cases, social workers were very supportive of the options young people wanted to pursue, giving them a sense that they are being listened to. For example, when Levi was considering continuing his education abroad, his social worker was very encouraging and supported his ideas.

So, coming up to 18, I was lucky enough to have a very good social worker. She was in my corner, and she was a huge support to me. And **she was very encouraging of my education**. So, for example, what I was looking at then, around 17, was studying abroad in Europe, because some of their degrees are done in English, they're fully funded, et cetera. Thankfully my social worker was very supportive. Now, I chose to stay on in Ireland, but she was **very supportive** of exploring that with me, so at least I had that as a backup option. (Levi, FG2, in third-level education (university))

However, Ezra felt that the support he received from his social worker regarding his education was minimal. 'I wasn't getting a lot of visits from my social worker.' (Ezra, FG4, interrupted third-level education (college))

#### 6.2.3 Support from foster parents and care staff

Previous studies have shown that support from foster parents has a positive impact on the educational careers and decision-making of young people. In the same vein, the interviews conducted for this study showed that in some cases, it was the foster family who was instrumental in supporting the young person in the decisionmaking process.

[My guidance counsellor] wasn't much help, and I suppose **my foster mum**, like she was quite good. **She'd sit down with me and talk to me about different colleges and stuff like that**. Like you know, we thought it was better that I didn't go to any big universities because they have big lecture rooms ... So, we decided that I'd go to a smaller [one]. So, my mum and myself decided what I was going to do. (Isabella, FG1, in third-level education (college))

When Camila was trying to decide what to do after finishing school, she got a lot of support from the staff of her residential care home. The staff were aware of Camila's preferences and encouraged her to pursue her dream. They also helped Camila in understanding whether her dream was realistic and achievable.

For me, it was **staff in my residential** [setting] that were really helpful because obviously they spent every day with me. So, they knew me

more than anyone. So, when I came back home and I was like, 'What career should I go for, things I should do, like health care, should I do something in business?' Like they knew that I always had a passion for being in the [specific field], so **they were extremely helpful and supporting the career that I wanted to go for** and, you know, making sure that it was really what I wanted and what would suit me going forward. (Camila, FG1, in third-level education (PLC))

Throughout the interviews, it was evident that young people wanted to be heard and have their ideas considered by the significant adults in their lives. Hazel felt that 'I think people should listen more to what the young person is saying. Rather than what they think is best, because this is happening to you, and if you're saying this, then you know yourself best.' (Hazel, FG2, in third-level education (college))

#### 6.3 CONCLUSION

For many young people, educational attainment offers them an opportunity to improve their situation and have better life chances. For care-experienced young people, continuing their studies and getting a good job enables them to exit the adversity in their lives. However, despite understanding the importance of education, many encounter a range of obstacles along the way. Support and advice from various sources are important for a young person during the decision-making process. At this juncture in their lives, young people need to be assured that they can do well and achieve their goals. While teachers can be an important support for students, some, however, may not support the ambitions young people have. Furthermore, career guidance can be a very useful source of information for students. However, negative feedback from career guidance teachers may demotivate some young people in pursuing their preferred post-school option.

In addition to teachers and career counsellors, social workers and foster families are another potential source of support and advice for young people regarding their post-school options. However, as some young people have limited contact with their social worker, there is little additional advice that a young person can receive from them. The quality of support from foster families is also important – young people with supportive foster families who recognise the importance of education can get a lot of advice and support from them. In both 'in third-level education' and 'interrupted third-level education' groups, there were students who felt that they did not get sufficient support in the decision-making process. Foster parents and social workers were mentioned more often by the first group regarding supporting their decision-making process.

It is important to note that the challenges associated with the transition from secondary to tertiary are applicable to all young people. However, the intersection of that school experience with care experience can be more limiting for care-experienced children and young people unless required support is put in place.

# **CHAPTER 7**

# Transition from care to aftercare and third-level education

## 7.1 INTRODUCTION

Throughout their educational career, children and young people experience a number of transitions: moving to primary school, moving from primary to secondary school and then to third-level education or training or, to the labour market. How well children and young people adjust to the demands of each following phase has an impact on their outcomes, as the dispositions towards learning and academic achievement are shaped by students' experiences at school early on (Bakketeig and Backe-Hansen, 2018). Most young people undergo these changes with a minimum disruption, while others may encounter considerable challenges.

As discussed in the previous chapters, care-experienced children and young people experience additional transitions throughout their lives as some have had multiple care placements and school changes. In some cases, these changes may also coincide with the major transition points in education, including entry into thirdlevel education. Care leavers are an heterogenous group in terms of their individual challenges, with many needing additional support when entering third-level education (Bunn and Fuller, 2023). In particular, having consistent support from a designated contact person throughout their third-level course can be a deciding factor for young people staying on in post-school education (Rak and Fuller, 2017). However, putting in place targeted supports can be challenging, as many young people do not wish to be identified by their care status due to the stigma they have experienced in the past (Sebba and Luke, 2019).

This chapter explores the experiences of this group of young people as they prepare for and go through a post-school transition. Leaving school and selecting their post-school trajectory happens at the time when they are also preparing for the exit from the care system. This chapter begins with discussing challenges associated with moving into aftercare. Section 7.3 focuses on the transition to college/higher education, highlighting areas where students require additional support. Section 7.4 discusses the importance of support available for young care leavers who have entered third-level education, while Section 7.5 concludes the chapter.

## 7.2 TRANSITION INTO AFTERCARE

According to the national aftercare policy in Ireland, a core eligible age range for aftercare is from 18 years up to 21 years. Aftercare services can support care leavers until the age of 23 if they are in full-time education or training. An aftercare worker can assist a care leaver in a number of areas including reviewing the

aftercare plan if they have one, giving advice on entitlements, arranging referrals to other services if needed, helping a young person to complete application forms where needed, as well as providing supporting documentation for grants, housing, social welfare or other applications.<sup>25</sup> While a young person can be referred to aftercare from age 16, once they meet the eligibility criteria, this does not always mean they are allocated an aftercare worker<sup>26</sup>. The difficulties involved in moving out of foster or residential care can be daunting for some young people. The focus group discussions with care-experienced young people in this study showed that not all were satisfied with the level of support they received during the post-school transition. For example, Ezra felt that for him the transition was difficult, given the extent of the change.

It is very difficult coming straight from foster care as a whole, from second-level education, with a family that you know you weren't born into ... To have **to jump straight from second level to third level without any additional supports in between**. So that you can come to terms with leaving foster care and entering aftercare. (Ezra, FG4, interrupted third-level education (college))

The interviewees noted that there is a need for young people to have access to supports to make the 'double transition' – moving out of care and secondary education to post-secondary education. Concern about the uncertain future ahead of them emerged from several interviews with the young people, mainly centring on not having the same family support structure that their peers do, in case something does not work out.

I feel like once you get to ... when you're near 18, when you have family and stuff, you can fall back on if stuff goes wrong. If college isn't working out or something like that. But once you're older and you're in aftercare, it's kind of ... down to you. You've to make that decision, which is really hard because I'd love to just be able to say, oh, I'm just gonna move back into my parents' house now because I can, but I can't, you know? That's difficult, that's hard. It's a difficult transition from secondary school to just being by yourself. (Hazel, FG2, in third-level education (college))

Some young people in this study felt that aftercare support should be available earlier, considering the significant change in the lives of care-experienced young people as they move out of foster care or residential care. May had asked for an aftercare worker towards the end of her secondary school in order to get the support she needed.

*I think they should start giving the aftercare worker like bang on at* **16 and not 17 – nearly 18** ... It's just the whole process of it is shocking.

<sup>&</sup>lt;sup>25</sup> See www.tusla.ie/services/alternative-care/after-care/tell-me-more-about-aftercare-workers.

<sup>&</sup>lt;sup>26</sup> Aftercare from age 16 onwards is provided for young people who have spent 12 months in the care of Tusla or the HSE, between ages 13–18.

It's like you're constantly fighting for something that you're entitled to automatically ... I had to go and actually get them on my side and **purely to fight to get an aftercare worker** ... I should have had him when I was 16. (May, FG3, in third-level education (college))

Several other people in the study were also very critical about the aftercare they received. In some cases, they felt that they did not get sufficient support from their aftercare worker. For example, Sofia felt that her aftercare worker neither had sufficient time for her, nor supported her aspirations.

Well, I found it an absolute nightmare to be honest; it didn't work well with me, and I didn't have a connection or bond with my aftercare worker. I think it would have been very important to actually get on with your aftercare worker. She didn't really have time for me to be honest. And she shot down the [type of course] studies that I wanted to do. (Sofia, FG5, interrupted third-level education (college))

Limited contact with her aftercare worker was also mentioned by Willow, who met with her aftercare worker twice a year. Willow had dropped out of third-level education due to her personal circumstances and found that her aftercare worker was not very sympathetic about the challenges she was facing at the time.

I found aftercare a nightmare and I didn't get on with my aftercare worker. **We barely talked** ... I've dropped out of college due to personal reasons. And there's just no support. I rang my aftercare worker to explain the situation and she was kind of like, 'Oh, **you're on your own now** ... you're 22 years of age now. You should know how to do this by yourself' kind of thing. (Willow, FG5, interrupted third-level education (college))

While Theodore acknowledged that aftercare workers could be a real help in practical matters, such as budgeting, limited access to this service made Theodore try and figure things out by himself.

I'd say, personally for me, I was 17 and first of all, I felt while aftercare was helpful in money management, budgeting, like I still found filling out important documents fairly difficult. **I had an aftercare worker for a very, very brief period.** And then I kind of did everything myself. (Theodore, FG1, in third-level education (university))

Limited access to an aftercare worker could be partially explained by a number of care leavers they are working with.

The aftercare worker at the time, she was new. **I'm assuming she had a lot on her plate** because I don't believe that she had just one of us. She probably had a couple of us that she had to work with. But why allocate somebody like that when you know it's needed? It's like the aftercare is something that is necessary, especially for somebody coming out of care, you know **it's a complete shock because you're left on your own. Like completely left on your own**. You don't know what to do. Like when you're 18, you're gone. (Lily, FG3, in third-level education (college))

While Sofia also noted that there was no support from aftercare, she could contact a career counsellor in her college for advice. Sofia also felt that an aftercare worker can have a considerable power to make decisions about her life.

[I had] no support from aftercare I could go to, like it was a no-go zone. I was lucky enough in college – I had this career guidance teacher, counsellor who I would see on a weekly basis every two weeks. And she helped me through a few things with regards to education and all that. To be honest, I was afraid of my aftercare worker, I actually was, because of all the power that she did have. Like, they can give me something as quick, but they can take it away just like that, you know? (Sofia, FG5, interrupted third-level education (college))

Eliana (FG5, interrupted third-level education (college)) and Theodore (FG1, in third-level education (university)) also noted that they did not have sufficient information about grants and other entitlements.

The interviews demonstrated that information about aftercare and various entitlements and supports is very fragmented, making it difficult for young people to navigate the post-care landscape. Mila, who had struggled with finding necessary information, became part of an initiative in her college that was responsible for informing young people about the supports available.

I'm a student ambassador for the Access team in my college ... We inform care-experienced students about their entitlements when they're coming into college. So, things like SUSI, because those are not clear, there are terrible websites for a start, but also, they're just not really clear. So, **if you don't have someone telling you that you should apply for these, you won't know that they exist**. So, I suppose that was kind of a thing that I struggled with at first. (Mila, FG2, in thirdlevel education (college))

Many interviewees felt that there are currently not sufficient resources to help young people leaving care, including social workers and aftercare workers. Theodore argued that considering the high turnover among the aftercare workers, it would be useful to have some standard documentation about procedures and supports that would serve as a signpost for a new person taking up the role.

There are simply not enough social workers, and definitely **not enough aftercare workers to meet the amount of people leaving care** ... And also, you're having such, you know, **turnover rates within the field**, you know ... I think that would be really, really helpful even if the aftercare worker can't be fully engaged at all times, or if there is a change you have a document there, you have a signpost basically. I think that would be really, really helpful. (Theodore, FG1, in third-

#### level education (university))

The topic of aftercare was also extensively discussed by the stakeholder consultation group. The participants noted that all eligible care leavers<sup>27</sup> should have an aftercare worker assigned to them. It was also noted that the level of engagement between the young person and the aftercare worker may vary depending on the young person's previous experience or other factors. The stakeholders argued that aftercare workers tend to offer support across a number of areas (see above), but that as a service, it is stretched.

[Eligible young people] all have an aftercare worker, but **the level of engagement changes from student to student** ... **Sometimes they don't engage**, and sometimes the reason for that can be that their previous experience hasn't been good ... yes, [aftercare workers are] really busy, they are trying to manage education, but also the housing health, all the daily kind of practical issues as well. But **I would say they're stretched**. (Stakeholder consultation group)

A call for more support for children and young people leaving care was already issued by the delegates at the Irish Aftercare Network's Annual Conference several years ago. The delegates found current service provision for care leavers insufficiently resourced to provide the level of practical support or therapeutic support for this group of young people, many of whom have suffered significant trauma during childhood.<sup>28</sup>

The stakeholders in this study noted that the provision of aftercare services can also be region dependent, with some larger regions being more stretched than others, with the number of referrals that the service receives taking longer to process.

And so, [aftercare workers are] **very, very, very stretched**. And referrals coming into them are taking a little bit longer for that reason. So, in some areas, I think they are probably sufficiently resourced. But in other areas, I know they're under-resourced. (Stakeholder consultation group)

Aftercare provision was associated with various challenges including supporting a young person who moves to another city to pursue their third-level education as well as being driven by personal circumstances of some young people.

If a young person is from [town name] and they're attending college in [town name], **the geographical spread is very hard for an aftercare worker**. They might have phone contact, but the physical contact is often missing because for the aftercare worker to go from [A] to [B], it becomes a full day's work. And then the nature of the work, sometimes they're crisis led. I'm not saying all young people in aftercare are in

<sup>&</sup>lt;sup>27</sup> For more information, see www.tusla.ie/uploads/content/4255-TUSLA\_Allocation\_Aftercare\_Worker\_v3.pdf.

<sup>&</sup>lt;sup>28</sup> See www.focusireland.ie/press-release/irish-aftercare-network-calls-for-more-support-for-young-people-leaving-care.

crisis, but it can be crisis led. So, then **the person that's in the biggest crisis gets the resources** and the young people or students that are doing quite well might be kind of siloed or become the secondary priority, if that makes sense. So, if you feel the students are doing well, they are in college, then maybe the priorities don't always go to that young person either. (Stakeholder consultation group)

## 7.3 FROM CARE TO COLLEGE

### 7.3.1 Considering post-school options

For young people who have been in care, leaving school means choosing between different options. When asked what they wanted to do after leaving school, some young people in the study said that they had their sights set on continuing their education. 'I always just thought that's what you do, that you finish school and you go to college.' (Luna, FG1, in third-level education (university)) For Isabella, going on to third level was portrayed by her social workers and foster parents as the only option available to her.

I just thought that was **something you had to do**, to be honest. You know, you just went. You finish sixth year at school and go through to college. I didn't think about not doing college ... I just always thought that you had to go to college and that was it. Like, **I was never given any other options** but to go to college, and that was from social workers. And my foster mom in particular, she said no, you go to college. So yeah, I just thought that was something that everybody did, to be honest with you. (Isabella, FG1, in third-level education (college))

In other words, for some young people, going to third-level education was something that was expected of them, something that everybody did after finishing secondary school.

However, some others would have liked to have had some time between leaving school and continuing their studies. Three interviewees had wanted to do a gap year before moving into third-level education, but were advised against it, mainly because of the rules around the aftercare allowance.<sup>29</sup> In fact, Willow noted that the only reason she considered third-level education was access to the aftercare support.

Like, nobody believed in the gap year. **I was warned about not doing the gap year**, you know? So yeah, [going to college was] just expected. (Isabella, FG1, in third-level education (college))

I didn't really want to go [to third level]. I wanted to do a gap year, and I kind of **was told pretty sharply that that wasn't really gonna be** 

<sup>&</sup>lt;sup>29</sup> Aftercare allowance of €300 per week is available for eligible care leavers attending accredited education or training course. This is available to care leavers up to the age of 23.

*a great option for me* because of the aftercare payment (Ava, FG1, in third-level education (college))

I think **the only reason I went to college was because of aftercare**. Because I needed that support. I always wanted to take the year out between doing my Leaving Cert and going to college, but that wasn't an option. (Willow, FG5, interrupted third-level education (college))

For Willow, taking a gap year would have given her more time to try and figure out what she was interested in doing after finishing school. She felt that not knowing what to do and feeling the pressure to go to college affected her mental health. The situation was not helped by the fact that the college course was chosen for her, in a field that she had no interest in. This may partly be a reason why Willow did not complete the course she was signed up to do.

I was kind of **forced into college** and **the course was picked for** me because I didn't know what I wanted to do. Yeah, **I had no choice in it**. It was decided for me. Because before I moved into aftercare, my social worker at the time asked me what I wanted to do, and I didn't know what I wanted to do ... And then I just got a call one day: you're starting [a course]. On this date. I had the biggest argument of my life that I did not want to do it, that it wasn't for me. And I'm still put into the course, and I still have to go, because my aftercare worker would ring the college and ask them if I was in. And if I wasn't in, then I'd get a call. So, I was kind of forced to resign myself to go in, which I really didn't want to. (Willow, FG5, interrupted third-level education (college))

Researcher: And so, did you have any interest in this area?

Willow: No.

Financial support is available from Tusla based on assessment of need. Eligible young people who turn 18 years of age while in care are entitled to an aftercare allowance if they are in full-time education and training. Allowance payment from Tusla is  $\notin$  300 per week. Some young people<sup>30</sup> may also be eligible for an Aftercare Grant, which is a one-off payment, aimed at supporting them with initial moving home expenses. Setting up home allowance stands at  $\notin$  300. The aftercare allowance/grant emerged as a major theme in young people's discussion about their post-school pathway. Several young people felt that having an aftercare allowance tied to entry into third-level education was too restrictive and made them sometimes choose a course they may not have otherwise considered.

The first time I went to college for the [name of the course] course, where I left two months in. The only reason I did that was because I

<sup>&</sup>lt;sup>30</sup> This refers to people who turned 18 years in the care of the Child and Family Agency having spent a period of 12 months in care of the agency between the ages of 17–18 years. Eligibility for the grant is also established by financial assessment of need.

was told that **if I did not go to college after I finished secondary school, I would not be entitled to any aftercare**. So that really kind of just forced my mind into thinking that I have to go to college or have to do something, or else I'm gonna be, you know, thrown to the curb. (Ezra, FG4, interrupted third-level education (college))

Similar sentiment was expressed by Sofia, who felt that her future was very uncertain if she did not go to college.

I think I was just going to give it a go. To be honest, I was worried. Like when I did turn 18, what was I going to do and where was I gonna go? And how am I gonna get by? And, I won't lie ... I did go to college so that I would be supported as well and to stay in aftercare. You have to be in education. You don't get support for nothing up until the age of 21. So, there's kind of like an extra pressure then. (Sofia, FG5, interrupted third-level education (college))

Both young people ended up not completing the course. A phrase 'on my own terms' frequently featured in conversation with young people in this study when discussing entry into third-level education. Some people had wanted to continue their education after finishing their second-level education but would have preferred to make up their own mind rather than having their decision-making tied to the aftercare allowance and support. Hazel felt that despite wanting to continue her education, she did not enjoy the experience as she felt she was forced to make this decision.

I was basically forced into education because of the aftercare allowance ... So, I felt that it's more like you are forced into it rather than you want to go into it. I did want to go back into education. But on my terms kind of thing. But I was told, 'If you don't go back in now, you're not getting this [allowance] and you're not staying where you are' ... So I was kind of forced into making an education choice that I didn't really want to (Hazel, FG2, in third-level education (college))

While Levi enjoyed the course at third level, he still felt that he was forced into education, rather than doing it on his own terms. Levi also noted that some flexibility is necessary when granting support for care leavers, as some of them may be struggling with their personal issues and mental health as they leave school, and an entry into a new environment may not be in the best interest for such young people.

It **should be on your own terms** and there needs to be a level of flexibility granted. So again, what I mean by this is, I felt that I was forced into taking the course and I was really lucky with the course that I got. It's a very good course at a very good university, and I'm not complaining. I'm really enjoying that course, but I didn't feel that I had an option. I'd also say that I was nearly taking a risk by accepting that offer in terms of my mental health. (Levi, FG2, in third-level education (university)) The interviews with young people demonstrated that linking aftercare support to third-level education was seen to have both positive and negative aspects. For young people who do not necessarily want to progress to third level immediately after secondary school, other options should be available. Going to college and ending up on a course that a young person does not enjoy, is not ideal for many reasons. However, the link between aftercare and college could also be positive, helping people to figure out what they want to do in future. For example, while Ezra felt very lost during the transition from secondary school to third level and feeling that he is almost doing it 'for the money', he acknowledged that linking an aftercare support to post-school education could incentivise young people to get a qualification and a career.

Obviously, you won't be able to ever get everyone under the same bracket. Some people will want to work. Some people will want to do an apprenticeship. Some people want to take time off, and other people would want to go straight into college. Obviously, you know you can't just facilitate all of that. But I do think that there definitely could be a different way around ... but I do think that it definitely still could be tied to college and apprenticeships because they do give incentive to, you know, get a course, so that you can have a career. So that you can almost plan out what you wanna do in the future. (Ezra, FG4, interrupted third-level education (college))

The views of young people participating in the study were also echoed by the stakeholders. There was a consensus among the consultation group participants regarding inflexibility of the aftercare allowance. The allowance may be discontinued if the person needs time out of their third-level course for whatever reason. These young people may not return to college later on.

Aftercare policy and the 'continuing education' clause means their allowance can be stopped if they take a year out, or in some cases have to stop due to mental health challenges. And **this can lead to young people not returning to higher education**. (Stakeholder consultation group)

Some participants argued that not all young people may be ready for continuing their education straight after finishing secondary school. Furthermore, more could be done to provide them with necessary skills that would make it easier for them to transition into young adulthood and enter third-level education.

The aftercare allowance, I suppose there is inflexibility there – a lot of young people aren't necessarily ready for further education, higher education at that time. And if there could be more room for personal development, mental health support, that could be financially supported to get them more able and ready and competent to take on the rigours of academia and college life and young adulthood. (Stakeholder consultation group)

I think the other thing I would say is that the aftercare allowance, that flexibility, I think, needs to happen, because **not everyone is mentally**, **academically ready after Leaving Cert**. Especially considering all the difficulties with the placements and the chaos. Sometimes **there needs to be room for personal development before going back into academia**. And I'm saying that as a guidance counsellor, you know. (Stakeholder consultation group)

The current aftercare allowance may not be sufficient for young people considering the cost-of-living crisis in Ireland. The situation is particularly precarious for young care leavers who do not have families to fall back on when things may get difficult.

I think another layer of the practical side of it is just the cost-of-living crisis and the fact that **an aftercare allowance hasn't changed in years**, and it doesn't replicate actually the increase. Never before have so many young people still been living at home, or older young people going back to live at home and having that base, and just those ripple effects of that impact of where that's going to lead. (Stakeholder consultation group)

## 7.3.2 Challenges related to transition to college/higher education

Moving into third level can be challenging for all young people, irrespective of their background. Moving into third-level education may be challenging for care-experienced young people for a number of reasons, as discussed in this section. While many young people in care were used to encountering new environments and new people, moving to college can still be stressful for care-experienced young people.

You think that you'd be used to moving to different places and new places ... I suppose when I moved here for college I found that very difficult ... it was like PTSD again ... having to move and having to find a new place and being surrounded by new people. Like living in an apartment with two other people and stuff like that. Just like the whole movement to college type thing, I found quite difficult, you know? (Isabella, FG1, in third-level education (college))

Furthermore, timing of the move to a third-level institution was mentioned by some young people as an additional difficulty as the move coincided with many overlapping challenges. Their views are best summed up by Sofia:

I had started college in September, and I moved out in December, that Christmas. So, **I was in the middle of everything** like, you know, I'm trying to move things and attend and then organise myself and all that. **That was a huge, huge challenge**. (Sofia, FG5, interrupted third-level education (college))

#### 7.3.2.1 Stigma attached to care experience

International research has shown that care-experienced young people can sometimes encounter stereotypical attitudes that associate being in care with poor educational outcomes, becoming young parents or turning to crime and may be in care for the reasons of their own doing (Dansey et al., 2019). Chapter 4 revealed that care-experienced young people in Ireland also feel stigma attached to being in care and sometimes encounter negative stereotypical views associated with being in care. When discussing their experiences and challenges in third-level education, some interviewees again highlighted stigma attached to care experience. Mila and Levi noted that care-experienced young people may be perceived by some as badly behaved and likely to end up in jail.

I think there's a big stigma ... like if you're in foster care ... that people in foster care are kind of different. And they're, like, **badly behaved**. And like **most people are gonna end up in jail**. Even if that's the case for some people, it's not the case for all people that are in care. So, it just kind of like impacted me in that kind of way. (Mila, FG2, in thirdlevel education (college))

Ireland does have this kind of stigma of those who are care experienced, maybe **having antisocial problems**. They do drugs, they do alcohol, et cetera, et cetera. (Levi, FG2, in third-level education (university))

Even if a young person had not encountered people with negative views of careexperienced young people, being in care is not necessarily the identity they want to be associated with. These young people felt that they wanted to develop their own identity, one that is not associated with being in care.

I find it hard to tell people even in college that I was in care. I don't want people to see me as a 'hero' ... Because I came into college when I was 17, so some people would have been like, 'Wait, so you were kept in foster care, you did your Leaving Certificate, and you enter all these foster homes and then you had this struggle and then this happened to you and you're still in college?' ... It's not how I want to be seen. (Mila, FG2, in third-level education (college))

For these reasons, about half of young people in this study did not want to reveal their care background when going to third-level education. However, not knowing about the challenges some young people experience, it can be difficult to assess the numbers or make targeted support available for these young people.

We've had an issue here as to **how to get data of the numbers of CES** [care-experienced students] in our colleges and we're not getting it. **They're not coming forward**. And I get it because it was talked about – sensitivity and identity. (Stakeholder consultation group)

#### 7.3.2.2 Social and academic sphere

Leaving secondary school and moving into third-level education is likely to cause some adjustment issues for all young people, including those in care. Adjustment challenges were also highlighted by young people in this study, particularly when the move involved moving from a quiet, rural area into a large town. 'I think the socialisation aspect was a bit different for me and trying to make friends. So, I think I was a bit unprepared in that aspect.' (Nora, FG2, in third-level education (university))

Young people who had mental health issues seemed to face additional difficulties in settling into college life and making new friends. Hazel ended up gravitating towards young people who had similar experiences and felt it easier to connect with them.

So, it is quite difficult making new relationships with people because I just think, oh, they're just gonna leave now. And what's the point in making friendships? But the majority of my friends have been in foster care, so they are my friends, my care friends, to be honest. It's people who share that experience. You're kind of just able to connect with them on a different level that you're not able to do with other friends. (Hazel, FG2, in third-level education (college))

Many young people who moved from secondary school to third level found it difficult to adjust to the new teaching and learning approaches in the latter. Instead of structured classes at school, in post-school education, students need to manage their own time. There is also more emphasis on self-directed learning. Some students that were interviewed for this study felt that they were not prepared for this shift, found study hard and struggled with attendance. The situation was even more difficult for students who had learning difficulties.

Academically speaking, it was more challenging. I didn't expect it ... I think that just could be the education system, though, like I didn't feel prepared going into college. It was a big shock. (Nora, FG2, in third-level education (university))

I think **attending, then coming home and doing it [study] by myself**, there wasn't really a lot of support in the college I was in. **I struggle academically**, so if there was stuff that they couldn't understand, then there wasn't a lot of support to explain it to me. (Willow, FG5, interrupted third-level education (college))

Sofia felt that her aftercare worker put more pressure on their attendance and study than their tutors. She found it particularly difficult due to her personal issues at the time, which were not taken into consideration. The accumulation of stress made this person drop out of college.

I suppose just **the constant pressure with the attendance**. I won't lie, it was a nightmare for me. I am trying to attend and it's just the amount of pressure ... not even tutors would put on to you. But like, again aftercare, the amount of pressure that they put on me was horrendous. And my own personal circumstances as well, [things] that were going on in my family weren't taken into consideration as to how that would have affected my performance in studying in college. So, that would have been probably another reason why if I were to drop out ... I more or less did towards the end of it. (Sofia, FG5, interrupted third-level education (college))

While already struggling academically, health issues had an impact on attendance for Willow, who subsequently dropped out of third-level education. This added additional stress to this young adult, as aftercare support was withdrawn once she was no longer in full-time education. They felt responsible for not staying on the course and reported feeling they had nowhere to turn to for help.

I did drop out this year due to health reasons and there was no support from aftercare. It was kind of like: This is all your fault. Like everything in aftercare, my money, where I'm living, all of that is just gone. It's like gone within seconds of when you drop out of college. And no, I got the letter from my doctor just to say that I wasn't in the right headspace and that I wasn't able for college this year. And then I got a letter from my doctor to say that I physically wasn't able to get into college every day. And that was just pushed aside, like it didn't matter. I asked could I change to online learning and that wasn't an option either. (Willow, FG5, interrupted third-level education (college))

Inflexibility of the system was also highlighted by the stakeholder consultation group. It was acknowledged that the current aftercare system that is based on 'continuing education' does not allow for the mental health issues or personal circumstances that many care-experienced students experience. Some may need to take a short break before returning to their studies, but the current system makes it very difficult for students to re-engage once they have dropped out of third-level education.

Mental health and gaps in education where young people may take breaks and then return to education, especially in the aftercare system, which refers to continuing education, it **makes it difficult then if young people want to re-enter third-level education** ... it's not impossible but it can be a challenge. (Stakeholder consultation group)

While many of the challenges mentioned above could also be encountered by young people with no care experience, some can be specific to those in care. For example, some young people felt that weekends and holidays were a difficult time, especially when their peers were leaving the campus to be with their families. Many felt that going to college enabled them to turn a new leaf in their lives and create a new support network around them.

When everybody's going home during the weekend, I always kind of feel jealous because it's like, oh, they have something good to go to ... So, there were a lot of traumatic events I would have experienced [at home] along the way when I was younger; it would have been very hard to go back home because it's like, **I'm going back to my trauma rather than I'm going back home**. Like when you're in foster ... when you come to college, **you're kind of trying to form your own family**, **like with the friends you have**. (Mila, FG2, in third-level education (college))

#### 7.3.2.3 Mental health issues and support with emotional wellbeing

Some young people who had previously experienced mental health issues found it challenging to start college or sit their exams. For May, who has mental health issues, in-person learning at third level would have been a challenge. For her, starting her course online at first helped to build up her confidence. Mila was also experiencing mental health issues and was worried about how this might affect her exam performance.

I struggled with college a good bit, but what helped me was the actual lockdown where it was online the first year and **I knew I was hidden** behind the camera. I was like, oh, this is fine. Like I'll get through this year. And then as I got more comfortable over the year, then I was like, OK, well if I can do it online, I can go in and do it in college, because **I've really bad anxiety** as well. So, I find it hard to go out. (May, FG3, in third-level education (college))

And like for me, **there is just so much pressure** ... I have exams next week but I still feel like, am I really ready? I'm kind of like **coming out of a bit of an episode or am kind of still in it**. I'm just not myself. I'm very tense and I just haven't been feeling well. I've just been like in bed. (Mila, FG2, in third-level education (college))

When asked about support regarding mental health and emotional wellbeing, Sofia (FG5) noted that this did not 'come into it at all'. She organised her own therapy, funded by the aftercare system. She received little support from her aftercare worker in doing so.

It didn't come into it at all. Now, **I took care of my own emotional wellbeing** with my own therapy and I made sure that aftercare funded it for me. You know, I kind of got on to them and said that you do have to actually fund me for that. So, they financially supported my emotional wellbeing through therapy, but the connection between me and the aftercare worker was like, no. (Sofia, FG5, interrupted third-level education (college))

For Willow, insensitivity from her aftercare worker regarding personal issues and related stress affected her emotional wellbeing.

Well, **I feel like there's no emotional support.** Like when I was in my first year of college, a family member of mine got cancer, so that affected me going to college because it was really upsetting me, and

I was just told, 'Oh, it's not you that has cancer. So, **you have to keep going to college**.' (Willow, FG5, interrupted third-level education (college))

The stakeholder consultation group noted that many care-experienced students go through third-level education without any notable issues. However, the lack of mental health support and increased anxiety post-COVID-19 can be a problem for some care-experienced students, as well as housing instability. This may result in some students repeating a year, sometimes several times.

So, I'm working at the [name of university] and I'm kind of full-time working with care-experienced students, which is great. A lot of the students move from year to year with no problems, but I think 39% do repeat a year, at least once. But the students that are repeating a year for three or four times, that have linked in with me, it's either lack of mental health support or extreme anxiety post-COVID-19. Some of them lived alone during the lockdowns, and then housing instability. Those tend to be the three things that kind of [impact their progression]. Being left between the transition from having CAMHS and then being now with the adult mental health, and maybe not received a needed diagnosis. The students that are really struggling tend to be around those specific areas. (Stakeholder consultation group)

#### 7.3.2.4 Financial issues

Young people in this study felt that financial issues were a notable challenge when moving into aftercare. Some felt that the financial support available from Tusla was not sufficient for their needs. The situation can be more difficult for these young people who need to pay for their therapy.

Like living in student accommodation. Much of my money is gone to live here, so I'm left with  $\leq 140$  a week;  $\leq 100$  of that has gone to counselling each week, which leaves me  $\leq 40$  to get things that I actually need, like shopping. Stuff to get me through the week. (Willow, FG5, interrupted third-level education (college))

In order to support themselves, some young people felt they needed to take up employment when in college, reflecting the situation for many young people in the general student population. This echoes the comments made by the consultation group about the financial aid to care-experienced young people having not changed for a while.

I ended up **having to get myself a full-time job**, so I was doing my bar work full time and trying to go to college. Late nights. I couldn't find the balance to put 100% of the work into college. So that was another challenge. (Sofia, FG5, interrupted third-level education (college))

While engaging in employment while in college is relatively common to all

students, the situation may be more challenging for care-experienced students as their attendance tends to be closely monitored by their aftercare worker.

Close monitoring of their financial expenditure by their aftercare worker was adding additional stress to the experience in post-school education for some young people.

Literally like giving the student financially what they need, what is considered enough nowadays. Like because you know how very expensive it is to live. But **not to be like controlling what they do with the money** that's being given to the student. Like because you know you're not gonna learn otherwise. You have to make mistakes in order to learn. You know you should not be crucified for a mistake. (Sofia, FG5, interrupted third-level education (college))

The stakeholders noted that young people should be given more guidance in how to manage their finances while in college and later on, when they leave education.

From my experience, I have seen numerous issues from a financial aspect – a lack of money management, no education/support given to the young person [about that]. Then, financial support falling off a cliff once they turn 22 or leave education. (Stakeholder consultation group)

#### 7.3.2.5 Accommodation

Accommodation was another challenge that young people in this study had to face, reflecting the difficulties experienced by all students due to the housing shortage. The options available to students include on-campus<sup>31</sup> or off-campus (rented) accommodation. Some third-level colleges do not provide on-campus accommodation<sup>32</sup>. This was seen to put an onus on young people to find a place in the current high-priced rental market. Ava and Levi felt that it is harder for care-experienced young people to find suitable accommodation within the funding available.

When I came to college, finding accommodation [was a challenge] because in my college, there's **no on-campus accommodation**, it's all private accommodation. So that's a bit of a weird one. You kinda have to go and look, and keep looking, and sometimes it might not be the most suitable accommodation and stuff like that, and also, it's really hard to budget  $\in$ 300 a week, which I know sounds like a lot, but when **my accommodation is \in650 a month, and that does not include electricity. So that's basically over half gone.** (Ava, FG1, in third-level education (college))

<sup>&</sup>lt;sup>31</sup> On-campus accommodation is currently over-subscribed (Department of Education, 2024). Students who are not successful in securing on-campus accommodation need to turn to the private rental market.

<sup>&</sup>lt;sup>32</sup> For the 2024/2025 academic year, Student Accommodation Assistance is being provided to assist students from Traveller and Roma backgrounds and care-experienced young people while attending third-level education. The latter must have experience of the care system between the ages of 16–18 and not be in receipt of an aftercare allowance.

I generally didn't know what to do, because it was either I get accommodation or I'm not going to college – like there's no way. I have no family within [name of a city] or commuting counties. It's either I get student accommodation, or I don't go to college, because the reality of me trying to find private accommodation when first of all I live in [name of a province], and I don't have a driving licence. It's just not possible, because everyone's looking for accommodation. So, you might only get a limited number of viewings, and it might be even harder for you to attend those viewings and then to show up on time and stuff like that ... students, at least from my university, which is a fairly big university, are not guaranteed student accommodation. I think that [care-experienced students] need to be included within the priority groups. (Levi, FG2, in third-level education (university))

Challenges related to accommodation of care-experienced students in college were also highlighted by the stakeholders. Once a young person turns 18, in most cases they need to move from a foster home/residential care into an alternative accommodation. Where the foster parents still provide accommodation, they face financial challenges associated with reduced allowance.

Accommodation after turning 18 is a challenge as **the young person may not have the option to stay on in the foster home** and foster carers who do offer accommodation ongoing have reported it to be financially difficult due to a drop in the allowance. (Stakeholder consultation group)

Securing priority accommodation for care-experienced young people is not common in Ireland. However, one institution has a small number of rooms for first year care-experienced students. The representative of this institution argued that unstable accommodation tends to be one of the biggest sources of stress for the students. The stakeholders suggested that the provision of accommodation for care-experienced young people could be modelled following an example from the UK, where such students can stay in the college-provided accommodation for the whole year and throughout their studies. It was felt that third-level institutions should reserve some accommodation for care-experienced students.

This year is the first year that we have kind of set aside [accommodation], we got eight rooms for first year care-experienced students. So, if they didn't get an allocation in the general lottery, I have a waiting list of students who've contacted us. And that has been another effective way of getting aftercare workers linked in, and students linked in as well. But I think also that's one of the biggest stressors for students is unstable accommodation. Like, there's a young girl who's living in a house with eight other girls in her first year at university, who previous to this had had multiple placements. So, it is something that I'm trying to advocate to the university – to expand [accommodation provision] from first year, because the UK models,

they will have accommodation that can go throughout the year, and you can be in that for all four years. Whereas our accommodation, because it's so limited, is very much reserved for first year students. So, there's more work to be done there, but I think it's made a difference. (Stakeholder consultation group)

A representative of another institution noted that while they do not currently have priority accommodation for care-experienced students, this is something that they would like to provide – indicating that there is an awareness of need and willingness to provide such support for care-experienced young people attending third-level education courses.

We don't own our student accommodation, but **we'd love to be in a position of prioritising ten beds a year or 15 beds a year**. And maybe that's something that we can work in conjunction with Tusla on securing. I'm not sure of how to go about it, but I think if every university could do 20 beds. (Stakeholder consultation group)

### 7.4 THE NEED FOR FLEXIBILITY IN SUPPORTING YOUNG CARE LEAVERS

The interviewees in this study felt that the support available to care leavers should be more flexible and take into account their various circumstances and challenges.

Levi felt that it is important for young care leavers to know what kind of educational pathway they want to pursue. Flexibility of support is needed as the pathway may not necessarily be an entry into third-level education. Some may be passionate about going into trade that is not accredited by Quality and Qualifications Ireland (QQI) but would allow the person to build a career and stable life for themselves. In addition to having access to different opportunities and courses, some flexibility for taking a break and taking stock may also be beneficial for some young people.

Maybe when you're in **some of the trades** that may not be QQI accredited, and **that might be what that young person's really passionate about** and could make a really good job and career and life ... So, I definitely think there needs to be support and help in that for a young person, the different opportunities and courses available to them. But **also allow a young person to take a breather**, I think is very important. (Levi, FG2, in third-level education (university))

Flexible entry into third-level education was highlighted by another young person. Ezra felt that he was not given an opportunity to say that he was not ready to continue his education straight after finishing school as the aftercare support is tied to being in full-time accredited education or taking up an apprenticeship. He argued that the current aftercare system provides young people with a direction, but this is not always enough if the person is unsure what option they would like to pursue.

I did visit a couple of different counsellors, and this was in between times while I was in school ... It was as if **I couldn't even tell them that**  *I'm* not ready for this right now, because if I did, it would be: 'OK that's not the way our system works' or ... 'You must advance into aftercare or, you know, get an apprenticeship or do a course or you have to be doing something.' It was very difficult to even to know what to do or where to go. It was almost like giving me a direction, but not having a roll-out, you know? There was nothing in front of me, and I didn't know where to go and ... I felt pressured. I felt like I had to make a decision now or else I would lose the extra benefits that aftercare does allow. (Ezra, FG4, interrupted third-level education (college))

The need for additional and targeted support for care-experienced students in third-level education was also acknowledged by the stakeholders. The appointment of a dedicated person with a responsibility of working with such young people was one approach discussed by the group, although across the third-level institutions, it is still 'unusual' to have such a person in place, as it is difficult to identify care-experienced students if they have not made themselves known. 'If we were to work solely with care-experienced students, we wouldn't know who they are. We actually don't know who the students are.' (Stakeholder consultation group) The institution where a dedicated person was in place, students were identified by a number of different means, including a laptop loan scheme on financial aid services. Most of the students that have linked in with the dedicated person came through referrals from aftercare services.

The stakeholders highlighted the need to have a dedicated person, such as a guidance counsellor, in place that works specially with care-experienced students and is funded by the state across all higher education institutions.

I think the other thing that is really important, in terms of the good supports, **having that one good adult**, and the extra support that could be available through a guidance counsellor in Tusla or other educational settings that could be that one good adult in the personal and academic space. There's a role there. I think it's really important to get more allocations for that. (Stakeholder consultation group)

It was also noted that data on care-experienced students could be collected centrally, at the CAO level. However, it was acknowledged that students need an incentive to tick the care-experienced box on their application form.

But if you could make [identification as care-experienced on an application form], but the reason they tick the care-experienced box there is **because they get supports that are very clear and very evident**. I'm not going to tick a care-experienced box unless I know what I'm gonna get out of it. I think, there has to be a carrot. Like there has to be a reason that somebody would benefit from doing that in the first instance. And so therefore, it has to be a national piece – that you get supported accommodation, or you get whatever. (Stakeholder consultation group)

Overall, it was noted by the stakeholders that to effectively support care-

experienced young people throughout their educational career and build up their ambition regarding post-secondary education, early intervention is a must, reflecting international research that suggest supporting care-experienced children's education and ambitions in primary school. The support systems must also acknowledge that direct entry into third-level education should not be the only option for care-experienced young people. The support available should be personalised, taking into account the wishes of a young person.

And early intervention ... the importance of that at all the key transition stages. And particularly that at those early intervention points, that there is, I suppose, **more aspirational work done** on what the potential of children is to achieve and clearer pathways on how to achieve their goals, that **there could be more work done in those areas**. (Stakeholder consultation group)

Guidance counts, and even earlier intervention ... because **the earlier** you can get in as a guidance counsellor, and then other support workers, that I think is really key to young people reaching their potential. And it needs to be person-centred as well, not just you're going into higher education. You have to get so many points. There are so many options now in the education system, and you can get into higher education in a roundabout way if you want, but it's just working with the person and what they want to do. And kind of building the support around that. (Stakeholder consultation group)

Being **able to avail of other options, other than college**, such as apprenticeships or going into a work environment. The aftercare payment is linked solely to education, which doesn't suit everyone, doesn't meet the needs of all young people. (Stakeholder consultation group)

Previous research in Ireland has highlighted the importance of interagency work in supporting young people while noting a lack of co-operation by agencies (McElvaney et al., 2013). In the same vein, the stakeholders in this study highlighted the importance of interagency collaboration involving guidance counsellors, Access Officers, and care workers in supporting care-experienced young people, considering the variety of challenges these young people experience. While there are different agencies represented on steering committees working with Tusla<sup>33</sup> at the moment, it was felt that the makeup of these may vary and may not always include input from some essential stakeholders that work closely with care-experienced young people.

I think **interagency collaboration is key**. We all agreed on that point, whether we're in further education and training or higher education, and we have specific roles, as guidance counsellors, Access Officers

<sup>&</sup>lt;sup>33</sup> For example, an aftercare steering committee includes representatives from a number of government agencies. For further information, see www.tusla.ie/uploads/content/4250-TUSLA\_Steering\_Committee\_v3.pdf.

and dedicated care experience support roles. I think the interagency collaboration, because the challenges are so varied, just having that as a framework rather than something that would happen organically where the professional might know this person, or that ... just to have that tightened, uniform framework of support, I think would be really important. (Stakeholder consultation group)

We're coming back to that, the level of interagency work, which is out there. I know from my own experience that there are interagency steering committee meetings where you'll have a number of agencies working with Tusla. But again, **the makeup of those in committees can vary as well** ... Guidance counselling, I think is key. (Stakeholder consultation group)

In addition to early interventions and interagency collaboration in supporting care-experienced young people, consistency of support was found to be essential. 'Support structures (educational and other) set up throughout the educational system for all transitions and ongoing throughout school and college.' (Stakeholder consultation group) Overall, the stakeholders called for a change in supporting care-experienced young people. 'We need systemic change, things are very siloed, we need more interagency, we need more referral systems and dedicated supports.' (Stakeholder consultation group) In order to bring about change, various organisations need to operate in tandem in providing consistent support to care-experienced young people.

## 7.5 CONCLUSION

This chapter has focused on the transition from the care system and second-level education to aftercare and third-level education. In recognition of the importance of post-care transition, supports such as an aftercare worker, have been put in place to assist young care leavers. However, the relationship between the young person and their aftercare worker is not always positive as demonstrated by the narratives of young people in this study.

Making the transition from second-level school to third-level education can be challenging for young people due to the differences in teaching and learning approaches between the two sectors. Many young people may also be uncertain about whether this pathway is right for them, or even if it is, what course they should apply for. Furthermore, for some, the decision to enter third-level education tends to be linked to access to aftercare allowance rather than a desire to continue their education. While providing a useful roadmap for many young people who are not certain what to do after finishing school, a requirement for being in full-time education or training in order to avail of aftercare support may put some people under pressure, especially the ones who may not have chosen this pathway, or who may have wished to pursue this pathway later on.

Once in third-level education, young care leavers tend to encounter a number

of overlapping challenges, including inadequate finances, issues around accommodation, adjusting to more self-directed learning, to name a few. Young people with learning difficulties or who experience mental health issues tend to find the transition to third-level education even more challenging. Some young people were critical about the level of assistance they received, particularly regarding their mental health and emotional wellbeing.

In order to assist young people, changes are needed, whereby children and young people in care receive consistent and sufficient support throughout their educational career. Interventions need to be introduced early, and the support needs to be embedded in an interagency collaboration and be sufficiently flexible, reflecting the varied, and sometimes complex needs of young care leavers. Designated personnel, or 'one good adult' continues to be important in the lives of young care leavers as they enter into independent living and third-level education.

## **CHAPTER 8**

## **Conclusions and policy implications**

#### 8.1 INTRODUCTION

In cases where the birth parents of children and young people are not able to offer adequate care for them, they are provided with alternative care arrangements by the state. Although being in the care of the state is something that is common among these children and young people, they are not a homogenous group as their personal experiences and circumstances vary. International research across different jurisdictions (Evans et al., 2017; Gypen et al., 2017; Haddow et al., 2021; Bakketeig et al., 2020) and Irish research (Darmody et al., 2013; Brady and Gilligan, 2019; Brady et al., 2019a, 2019b) has consistently shown that, on average, these young people encounter a range of challenges throughout their lives and educational careers. Systematic reviews have shown that these children tend to have poorer educational outcomes and attainment level, often associated with high mobility in terms of multiple care placements and school changes (Gypen et al., 2017; O'Higgins et al., 2015; Townsend et al., 2020). A scoping review has indicated that the situation tends to be more difficult for young people who have learning difficulties or experience issues with mental health (Hammond et al., 2020). Furthermore, some care-experienced children and young people may encounter lower expectations from their teachers, foster parents and social workers and, at times, may not receive sufficient and consistent support from them (Mannay et al., 2017a; Harvey et al., 2015). In turn, poorer educational outcomes may lead to poorer life chances of these young people, as tends to be the case with all young people (Duncan and Murnane, 2011; Smyth and Hannan, 2000), but may be more acute for care-experienced young people.

According to the data from Tusla, there were 5,801 children (0–17 years) in care at the end of February 2025 (Tusla, 2025), mostly in general or relative foster care (CSO, 2024a). The CSO data show that while many of them had just one care placement, a small minority experienced several disruptions to their care placements. Compared to their non-care peers, care-experienced children were also more likely to be enrolled at more than one primary or secondary school (ibid.). In line with international studies (see Klein et al., 2022), care-experienced children and young people in Ireland also tend to have poorer levels of school attendance than their non-care peers (CSO, 2024a). Being in foster care, compared to other types of care, seems to act as a protective factor against poor school attendance (ibid.). Reflecting various challenges these young people encounter, some tend to leave school without a Leaving Certificate (28% vs 8% of all students) (CSO, 2024a) – the results of which determine their entry into third-level education. Compared to all children in Ireland, those with care experience are less likely to attend higher education courses.

In Ireland, total enrolments in higher education have increased over time, with a proportion of individuals with tertiary degrees now higher than the EU average (CSO, 2024b). To widen participation in higher education, efforts have been made to facilitate access to higher education among underrepresented groups such as those from lower socio-economic backgrounds, ethnic minorities (e.g. Traveller and Roma communities), and students with disabilities. It has been increasingly acknowledged by studies from different jurisdictions and Ireland that although care-experienced young people is a highly heterogenous group, these young people face a number of challenges when accessing third-level education, including lower levels of educational attainment, low expectations from teachers and carers, low aspirations and concerns about the affordability of continuing their education (Abbott-Chapman, 2011; Bayfield, 2023; Colvin and Knight, 2021; Harvey et al., 2015; McNamara et al., 2017). Reflecting this, the National Access Plan: A Strategic Action Plan for Equity of Access, Participation and Success in Higher Education 2022-2028 (Government of Ireland, 2022) has included care-experienced young people in the categories that need additional support in continuing their education in this sector for the first time.

Previous international research has shown that a lower level of participation in tertiary education among this group can be associated with a number of factors, including prior academic outcomes and low aspirations, but also insufficient information and support in planning their post-secondary school trajectories (McNamara et al., 2017; Sellar and Gale, 2011; Williams et al., 2020). Finishing school and mapping out their next steps coincides with leaving the care system and entry into independent living and post-school education, in other words, a 'double transition'. This small-scale exploratory study has explored the factors that had influenced educational experiences of care-experienced young people while still in school and while preparing to leave school, to understand the challenges associated with the 'double transition'. In particular, the report aims to shed light on how the care experience impacts on academic and social spheres at school, what factors inform the decision-making process regarding post-school education, and the main challenges care-experienced young people face after leaving school and entering third-level education.

In addressing these aims, this report draws on the views and experiences of 16 care-experienced young people who had left school at the time of data collection as well as the views of key stakeholders representing a range of organisations. All young people in this study had entered third-level education. At the time of the interview, most of them were attending a course; however, four individuals had dropped out. To understand the progression of young people in this study to post-secondary education, the study delves deeper into their experiences when still in school and subsequent transition, seeking to highlight the factors that impacted on their aspirations and post-school trajectories. As the study is based on the views of a small group of care-experienced young people, the findings cannot be generalised to the population of care-experienced young people.

However, the findings provide useful insights for policy development and further research in this area.

## 8.2 MAIN FINDINGS

#### 8.2.1 Care-experienced children and young people at school

The results of the study show that how students fare in academically and socially is strongly influenced by their care experiences, particularly for these young people who had had multiple care placements and school changes, in line with international (Haddow et al., 2021; O'Higgins et al., 2015; Bakketeig et al., 2020) and Irish research (Darmody et al., 2013; Brady and Gilligan, 2019; Daly, 2012; McElvaney et al., 2013). Discussions with young people in this study indicated that having less challenges in school was associated with being placed into care at an early age, and having more stable and nurturing care placements, while high mobility in terms of care placements and school changes seemed to make their school engagement more challenging due to the unsettling nature of constant change. Care placement changes may result in students missing school and finding it hard to catch up afterwards. Multiple moves may have an adverse impact on a young person's ability to adjust to new routines and rules, their educational outcomes, ability to form interpersonal relationships and their wellbeing. Proactive and supportive foster parents play a key role in supporting the education of young people in their care and helping them in developing aspirations for their future. In addition to the foster parents, social workers can play an important role in supporting care-experienced young people, including help with adjusting to a new school. However, communication between social workers, foster parents and school personnel may sometimes be insufficient, as demonstrated by this study. Additional challenges were experienced by those young people who had mental health issues or learning difficulties, in line with international (Hammond et al., 2020) and Irish research (McElvaney et al., 2023). The findings of this study also show that some young people felt that some school staff and teachers had low expectations of them, which in some cases were linked to stereotypical views of care-experienced children as low achievers or ones who behave badly. However, most young people noted that they generally got on well with their teachers, and some had at least one teacher or another member of staff that they could approach for support. Teachers can have a transformative role in the lives of students. Their support is particularly important for care-experienced young people who often have disrupted lives in terms of care placements and change of schools. Students may find it easier to ask for support if a teacher shows genuine interest in their lives. The findings also showed that support from foster families and social workers was important for young people in this study in negotiating their way through school and supporting their education. Quality of support from foster families is important as young people with supportive foster families who recognise the importance of education can get a lot of advice and support from them. While many young people in this study had performed well at school, some struggled with attendance and juggling schoolwork alongside their everyday personal difficulties. Students who had learning difficulties or were experiencing issues around their mental health and wellbeing tended to find study harder, especially during the preparation for state exams with associated added stress.

The interviews conducted for this study showed that young people's attitude to school is often influenced by changes in the care placements and school changes, learning difficulties, insufficient support with schoolwork, not feeling as smart as other students, and mental health issues. Some of these issues were also associated with poor school attendance. It is important to note that poor school attendance and lower grades do not necessarily reflect lack of interest of a young person in education but is often influenced by circumstances beyond their control. Better understanding is needed about challenges faced by these students, many of whom may have had experienced trauma, which in turn has impacted their school attendance and engagement. Recently launched Take Care resources (EPIC, 2025), developed for schools, provide invaluable information for teachers and principals on how to best support care-experienced children and young people.

International (Schwartz-Mette et al., 2020) and Irish research (Darmody et al., 2013) has highlighted the importance of friendships in academic settings that can act as a support network for young people. Positive interaction between students has been found to enhance their social skills, support their learning and may contribute to their academic outcomes (Haddow et al., 2021). The results of this study showed that making friends was harder for young people who had experienced several school changes. Interrupted friendships made it more difficult for these young people to form new interpersonal connections, as their experiences had shown that once they needed to move schools, these friendships would, once again, cease. In some cases, care-experienced young people found it difficult to form close friendships as it was difficult to explain their living arrangements, especially when in residential care. A few young people had also experienced bullying at school. However, this was not always linked to their care background.

Interpersonal relationships with their social workers was another challenge that some young people had experienced. While in many cases the social worker was the 'one constant' in young people's lives, in other cases such support was found to be wanting, especially when there had been a change in a social worker or when the interaction between the young person and the social worker was limited. These young people often felt unsupported. Young care leavers also had varying support from their aftercare workers. While in some cases an aftercare worker was considered to be an asset in exiting the care system and supporting their transition to third-level education, in other cases this support was negligible, leaving some young people to be 'their own support'.

#### 8.2.2 Leaving school and leaving care

In line with previous research from Ireland and elsewhere, the results of this study

show that care-experienced young people who 'age out' of the care system need support in a range of areas (Mullan, 2018; Atkinson and Hyde, 2019; Ellis and Johnston, 2022). Information gathered from young people in this study showed that preparation for the 'double transition' of leaving care and leaving school was a challenging time for many. While some young people experienced difficulties in the academic sphere throughout the second-level schooling, preparation for the Leaving Certificate exams had left some of them additionally anxious and stressed, particularly those young people who were experiencing personal challenges such as mental health issues, learning difficulties and/or care changes. While all students may find this period in their schooling stressful, the situation is likely to be more challenging for care-experienced young people who often lack the supportive family structure, as has highlighted by international literature (Storø, 2018; Paulsen and Berg, 2016).

International research has shown that care-experienced young people may not have sufficient information about the possibilities and opportunities beyond secondary school (Muir and Hand, 2018). Many young people in this study had been undecided about what to do after finishing school. While all young people had access to a guidance counsellor, for many, guidance about their post-school options was seen as insufficient. For some, their personal ambitions for a course to pursue in third-level education were not supported by their guidance counsellors or their social workers. At this juncture in their lives, young people need to be assured that they can do well and achieve their goals. Negative feedback from teachers and career guidance may demotivate some young people. For others, their foster families and social workers were instrumental in supporting their ambition for accessing third-level education. Entry into third-level education was considered by some young people to be an important milestone, as it provided hope for a new start, an exit from the care system and to be 'like everybody else'.

Some young people perceived entry into third-level education after finishing school as something that 'you do'. Typically, these young people had a more stable care placement, had experienced fewer school changes and had high expectations from their foster parents and social workers. These students also tended to have more positive experiences at school. However, others were critical about this being the only pathway tied to their aftercare support. At present, aftercare support is available to young people in accredited full-time education or training. Several young people did not wish to pursue third-level education and felt pressurised to do so, as they felt there was no other support available for them once they leave the care system. Some had wanted to delay their entry into third-level education, but this option was not available to them. For some, having to pick a course in third-level education without really knowing what they wanted to do, resulted in young people picking a course that they did not really enjoy. Young people who had not wished to enter third-level education straight after secondary school, those who were undecided what to study, and those experiencing mental health challenges and learning difficulties, were also more likely to have an interrupted third-level experience. Greater flexibility in supports was also endorsed by the stakeholders in this study, echoing the views of young people, some of whom were uncertain about the next step, or were not ready to transition into third-level education.

## 8.2.3 Challenges faced by care-experienced young people in college/ higher education

Previous research from Ireland has demonstrated that moving from a secondary school to third-level education involves multiple challenges for young people (McCoy et al., 2014; Denny, 2020). Typically, this transition involves moving away from home and starting an independent life in college, needing to adapt to a new environment, forming interpersonal relationships with academic staff and peers and experiencing different approaches to teaching and learning. For careexperienced students, the challenges may be more acute, as they need to make this transition largely without family support or continued support of their foster parents, as highlighted by international research (Atkinson and Hyde, 2019; Ellis and Johnston, 2022). Based on the assessment of need, young care leavers in Ireland can be allocated an aftercare worker, who is responsible for assisting young people across a range of areas including support with all relevant documentation, information about their entitlements and making relevant referrals, where and if required (Tusla, 2019). The results of this study have shown that while some people found their aftercare worker very helpful during and after the transition from the state care system, others had minimal contact with their aftercare worker or experienced tension when interacting with them. Since young care leavers do not have similar support structures in place that are available to students who live with their birth families, having one consistent adult that they could turn to was considered essential, in line with international research (Storø, 2018). The intersection of school experience with care experience can be more limiting for care-experienced children and young people unless required support is put in place.

Several young people in this study felt largely unsupported in making the transition into third-level education and had experienced various challenges ranging from the practicalities of moving to a new location to figuring out their accommodation and finances, as well as managing their workload at college. Furthermore, it was felt that the finances were not sufficient considering the cost of living in Ireland, with some students filling the gap by engaging in term-time employment. While the cost of living is likely to have an impact on all students, it may be felt more acutely by care-experienced young people who often do not have access to family support. Others felt that monitoring their expenditure and college attendance by their aftercare worker was another source of stress and left them without independence or agency. While some had access to student accommodation, in some institutions there was no on-campus accommodation, only private lodgings that proved to be hard to find and expensive, especially in larger cities. As many young people that participated in this study were no longer in touch with their foster families, holidays and weekends were often difficult, especially when other students returned to their homes.

A few young people in this study had considered dropping out of their course or had, in fact, done so. The reasons ranged from mental health issues and inflexibility of the study at third level, to struggling with their workload and having learning difficulties. Many felt that there was very little support available or were unsure who to turn to or how to access help.

### 8.3 Directions for policy development and further research

Much of international and Irish research has highlighted the importance of students' experiences in school in developing their aspirations (Billet et al., 2024; Belando-Montoro et al., 2024; McCoy et al., 2014). Schools, teachers and social workers need to prioritise the needs of care-experienced young people to support their school engagement and ensure they have the best possible opportunity to continue their education after finishing school, as a higher level of educational attainment is associated with better life chances later on. While care-experienced children and young people have their unique life histories, as a group they often experience similar challenges. At school level, teachers and other staff members should be sufficiently informed about the challenges associated with being in care, to adequately support care-experienced children and young people. By adopting a whole-school approach, schools can create an environment where all students, including those with care-experience can thrive (EPIC, 2025). Furthermore, teachers need to be 'care aware' (EPIC, 2025), in other words, having sufficient understanding of challenges that many care-experienced children and young people experience. Initial teacher education (ITE) programmes and continuous professional development (CPD) should provide teachers with necessary knowledge and skills regarding how best to support care-experienced children and young people, many of whom have experienced trauma in their lives. The Take Care resource and accompanying guide developed for schools (see EPIC, 2025) are valuable resources for schools in informing teachers and principals about the challenges faced by care-experienced children and countering negative stereotypes. Furthermore, future research could explore how different stakeholders (schools, social workers and other agencies) interact in providing a wrap-around support to young people as they move through the care and education system. Care leavers is not a homogenous group, and support structures need to be sufficiently flexible in meeting the varied needs of these young people. To do that, it is important to fully understand the experiences of care-experienced young people as they move through the care and education system and beyond. It is hoped that the Care Experiences project led by the Department of Children, Equality, Disability, Integration and Youth (DCEDIY) with Tusla will provide valuable information on the long-term outcomes for young care leavers.

Discussions with young people in this study indicated that more support may be needed for young people who are experiencing issues with wellbeing and mental health. Policy interest in promoting wellbeing of all young people is demonstrated by the development of the *Wellbeing Policy Statement and Framework for Practice* 

2018–2023 (Government of Ireland, 2019). This important resource is aimed at 'schools, centres for education, children and young people, parents/carers, support services, government departments, non-government organisations and all those with an interest in wellbeing promotion in education' (p.5) and provides guidelines for effective practice. However, there may be need for free-of-charge, tailored ongoing supports that a young person can access when in third-level education/training. This could potentially be an 'incentive' for young people to disclose their care experience on a CAO form, an approach that could facilitate third-level institutions in identifying students for support. To provide support for care-experienced young people with mental health issues, more research is needed on how this may impact their access and success in third-level education.

The results of this study showed that some care-experienced young people may need additional support with their study and school engagement. Some may need additional tuition, especially when their schooling career has been fragmented. The School Completion Programme (SCP) is one way to provide additional support for these students. However, SCP is only available in DEIS schools and thus not accessible to students attending non-DEIS schools. Schools should also ensure that care-experienced students have access to high-quality and suitably informed pastoral and emotional support as well as learning support, to help them successfully complete their studies and develop aspirations for future. Close collaboration between schools, foster families, social and care workers is important to ensure that students' needs are met and that they have adequate support throughout their educational career. Interagency collaboration can make it easier for care-experienced young people to negotiate different transition points during their educational career.

Overall, an exit from the care system and second-level education followed by an entry into independent living and post-school education – a 'double transition' – can be challenging for many students. While still in school and in care, they can have 'many pairs of eyes' on them in terms of sources of support, the extent of support largely diminishing when they turn 18. Examples from other jurisdictions indicate some flexibility in age limits and eligibility conditions for care (Gilligan, 2018). If eligible, Irish young people in third-level education or training can avail of aftercare service by Tusla up until the age of 21. This can be extended to the age of 23 if a young person is still in full-time accredited education or training. However, a requirement for being in full-time education or training in order to avail of aftercare support may put some people under pressure, especially the ones who may not have chosen this pathway, or who may have wished to pursue this pathway later on. Advice and support from guidance counsellors is particularly important for care-experienced young people who lack such input from their families. Future research could also explore the decision-making of careexperienced young people after graduating regarding the entry into the labour market, particularly focusing on guidance and information received from college/higher education institutions in planning their post-graduation options.

Despite the aftercare support that is available for those who qualify for it, some young people may still struggle in post-school education. A national  $\leq$ 300 weekly aftercare allowance was introduced in 2015. As highlighted by the students and stakeholders participating in this study, financial support available to these young people has not kept pace with the cost of living in Ireland. Not indexing this to inflation/wage increases means an effective cut in support. Limited funding is likely to necessitate taking up term-time employment. This, however, can distract young people from fully focusing on their studies. While this may apply to all young people trying to combine study with employment, the situation can be more acute for care-experienced young people who tend to lack family support.

Like many other young people, care-experienced students often struggle in finding suitable accommodation. At institutional level, care-experienced young people should be prioritised when allocating accommodation and this accommodation should be available throughout the year for the duration of the course the student is attending. Because this will be a 'home' for the young person during their studies, it should be suitable for accommodating all the belongings of the student (similar to what is provided by the University of York). These students may also need additional support during weekends and holidays, especially when other students return to their homes. One possible solution could be building linkages with local families, which may provide an opportunity for young people to have a place to stay during this time (similar to the Home Comforts programme by the Sussex Learning Network).

Third-level education providers need to put adequate and targeted supports in place to assist care-experienced young people. However, they may not be aware of students' care backgrounds, particularly if a young person is reluctant to reveal this information. Availability and validity of data on care-experienced young people is needed, in order to put in place necessary supports for these students. In the UK, the Universities and Colleges Admissions Service (UCAS) has introduced a self-disclosure field for care leavers in their application form. Adopting a similar approach could be explored in Ireland regarding the CAO application form. Considering the unwillingness of some young people to disclose this information, self-identification needs to involve an incentive for a young person to do so.

The results of this study highlighted the need to ensure that young people experience a smooth and seamless transition from second-level education and the care system, and entry into independent living and post-school education. Failure to do so may result in an interrupted post-school education for some care-experienced students. Aftercare planning that starts early and involves input from young people, and consistency in support will go a long way towards ensuring that a young person has a seamless transition experience. Some young people in this study who had been allocated an aftercare worker had received very little support from them and felt that they were largely 'left to their own devices'. This could reflect the fact that the service is 'stretched', as noted by the stakeholders. At present, Tusla employs 71 aftercare workers, indicating that these individuals provide support to a number of young people in care.

# REFERENCES

- Abbott-Chapman, J. (2011). 'Making the most of the mosaic: Facilitating post-school transitions to higher education of disadvantaged students', *The Australian Educational Researcher*, Vol. 38, No. 1, pp.57–71.
- Alderson, H., Smart, D., Kerridge, G., Currie, G., Jonson, R., Kaner, E., Lynch, A., Munro, E., Swan, J., McGovern, R. (2023). 'Moving from "what we know works" to "what we do in practice": An evidence overview of implementation and diffusion of innovation in transition to adulthood for care experienced young people', *Child and Family Social* Work, https://doi.org/10.1111/cfs.12998.
- Atkinson, C., and Hyde, R. (2019). 'Care leavers' views about transition: A literature review', Journal of Children's Services, https://doi.org/10.1108/JCS-05-2018-0013.
- Baker, Z., Ellis, K., and Harrison, N. (2023). 'Theorising educational engagement, transitions and outcomes for care-experienced people: Introduction to the special issue', *British Educational Research Journal*, Vol. 50, No. 2, https://doi.org/10.1002/berj.3942.
- Bakketeig, E., Boddy, J., Gundersen, T., Østergaard, J., Hanrahan, F. (2020). 'Deconstructing doing well. What can we learn from care experienced young people in England, Denmark and Norway?', *Children and Youth Services Review*, Vol. 118, 105333, https://doi.org/10.1016/j.childyouth.2020.105333.
- Bakketeig, E., and Backe-Hansen, E. (2018). 'Agency and flexible support in transition from care: Learning from the experiences of a Norwegian Sample of Care Leavers doing well', Nordic Social Work Research, Vol. 8, pp.30–42, https://doi.org/10.1080/2156857X.2018.1518815.
- Bayfield, H. (2023). 'More of a kind of patchy transition into university as opposed to the kind of smooth story that people expect. Transitions to higher education for careexperienced students', *International Journal of Educational and Life Transitions*, Vol. 2, No. 1: 12, pp.1–15, https://doi.org/10.5334/ijelt.50.
- Belando-Montoro, M.R., Fernández-Salinero, C., Virgós-Sánchez, M., and Naranjo-Crespo, M. (2024). 'The decision-making of students in post-compulsory education: Influence of personal, academic, family, and socioeconomic dimensions on the choice of Vocational Education and Training', *Quality & Quantity*, https://doi.org/10.1007/s11135-024-02026-1.
- Berger, L.M., Cancian, M., Han, E., Noyes, J., and Rios-Salas, V. (2015). 'Children's academic achievement and foster care', *Pediatrics*, Vol. 135, No. 1, 109, https://doi.org/10.1542/peds.2014-2448.
- Berrington, A., Roberts, S., Tammes, P. (2016). 'Educational aspirations among UK young teenagers: Exploring the role of gender, class and ethnicity', *British Educational Research Journal*, Vol. 42, No. 5, pp.729–755, https://doi.org/10.1002/berj.3235.
- Billett, S., Choy, S., Gibbs, K., Le, A.H., McKay, L., and Hay, S. (2024). 'Initiating, enacting and sustaining partnerships to inform post-school pathways', *Journal of Education and Work*, Vol. 37, No. 1–4, pp.149–164, https://doi.org/10.1080/13639080.2024.2383564.

- Boddy, J., Lausten, M., Backe-Hansen, E., and Gundersen, T. (2019). 'Understanding the lives of care-experienced young people in Denmark, England and Norway: A cross-national documentary review', København: Det Nationale Forsknings- og Analysecenter for Velfærd.
- Boguslaw, J. (2017). The aspirations-attainment paradox of immigrant children: A social networks approach, www.sofi.su.se/polopoly\_fs/1.343152.1503478458!/ menu/standard/file/Aspirations\_200817\_JB.pdf.
- Brady, B., Dolan, P., and McGregor, C. (2019b). *Mentoring for Young People in Care and Leaving Care: Theory, Policy and Practice*, London: Routledge.
- Brady, E., and Gilligan, R. (2019). 'Exploring diversity in the educational pathways of care-experienced adults: Findings from a life course study of education and care', *Children and Youth Services Review*, Vol. 104, Article 104379, https://doi.org/10.1016/j.childyouth.2019.06.014.
- Brady, E., Gilligan, R., and Nic Fhlannchadha, S. (2019a). 'Care-experienced young people accessing higher education in Ireland', *Irish Journal of Applied Soc*ial Studies, Vol. 19, No. 1, Article 5, https://arrow.tudublin.ie/ijass/vol19/iss1/5.
- Brännström, L., Forsman, H., Vinnerljung, B., and Almquist, Y.B. (2020). 'Inequalities in educational outcomes in individuals with childhood experience of out-of-home care: What are driving the differences?' *PLOS One*, Vol. 15, No. 4, e0232061, https://doi.org/10.1371/journal.pone.0232061.
- Brownell, M.D., Chartier, M., Au, W., MacWilliam, L., Schultz, J., Guenette, W., and Valdivia, J. (2015). *The educational outcomes of children in care in Manitoba*, Manitoba: Manitoba Centre for Health Policy, University of Manitoba. Retrieved from http://mchp-appserv.cpe.umanitoba.ca/reference/CIC\_report\_web.pdf.
- Bunn, M., and Fuller, E. (2023). 'Categorisations of care: Exploring representations of care leavers in higher education through critical praxis', *British Educational Research Journal*, https://doi.org/10.1002/berj.3912.
- Cameron, C., Hollingworth, K., Schoon, I., van Santen, E., Schröer, W., Ristikari, T., Heino, T., and Pekkarinen, E. (2018). 'Care leavers in early adulthood: How do they fare in Britain, Finland and Germany?', *Children and Youth Services Review*, Vol. 87, pp.163–172, https://doi.org/10.1016/j.childyouth.2018.02.031.
- Cameron, C., Jackson, S., Hauari, H., and Hollingworth, K. (2012). 'Continuing educational participation among children in care in five countries: Some issues of social class', *Journal of Education Policy*, Vol. 27, No. 3, pp.387–399, https://doi.org/10.1080/02680939.2011.644811.
- Carroll, E., McCoy, S., and Ye, K. (2024). 'Embracing diversity in all its forms': The voluntary secondary sector in Irish education, ESRI Research Series No 182, Dublin: ESRI, https://doi.org/10.26504/rs182.
- Central Statistics Office (CSO) (2024a). *Characteristics of children in care January 2024 and children who left care since April 2018,* www.cso.ie/en/releasesandpublications/ fp/fp-eaacc/educationalattendanceandattainmentofchildrenincare2018-2024/ characteristicsofchildrenincareinjanuary2024andchildrenwholeftcaresinceapril201 8.

- Central Statistics Office (CSO) (2024b). *Educational attainment thematic report 2024,* www.cso.ie/en/releasesandpublications/ep/p-eda/ educationalattainmentthematicreport2024.
- Cheruvallil, S., and Anand, K. (2023). *Expressions of self: Race, religion and representation* of minoritised children and young people in the British Care System, Research Report. Coventry University.
- Child, S., and Marvell, R. (2023). 'Care-experienced students in higher education: A case for re-figuring higher education worlds to widen access and further social justice', *British Educational Research Journal*, Vol. 50, No. 2, https://doi.org/10.1002/berj.3905.
- Children's Commissioner (2024). Supporting care experienced students: A handbook for professionals working in higher education, London: Children's Commissioner.
- Clemens, E.V., Klopfenstein, K., Lalonde, T.L., and Tis, M. (2018). 'The effects of placement and school stability on academic growth trajectories of students in foster care', *Children and Youth Services Review*, Vol. 87, pp.86–94, https://doi.org/10.1016/j.childyouth.2018.02.015.
- Collins, M.E., Augsberger, A., and Howard, R. (2023). 'Post-secondary vocational education for youth leaving care: Examining a potential pathway to successful outcomes', *British Educational Research Journal*, 1–21, https://doi.org/10.1002/berj.3933.
- Colvin, E.J., and Knight, E.B. (2021). Supporting care experienced young people into higher education, Collier Fund 4 Report, Charles Sturt University.
- Corbett, M., and Coulter, C. (2024). 'Falling through the cracks. The child law project. An Analysis of Child Care Proceedings from 2021 to 2024', Dublin: Child Law Project (CCLRP CLG), www.childlawproject.ie/wp-content/uploads/2024/11/Child-Law-Project-Falling-Through-the-Cracks-2024.pdf.
- Cosgrove, J., McKeown, C., Travers, J., Lysaght, Z., Ní Bhroin, Ó., and Archer, P. (2018). Educational Experiences and Outcomes for Children with Special Educational Needs. Dublin: National Council for Special Needs (NCSE).
- Cotton, D.R., Nash, P., and Kneale, P.E. (2014). 'The experience of care leavers in UK higher education', *Widening Participation and Lifelong Learning*, Vol. 16, No. 3, pp.5–21, https://doi.org/10.5456/WPLL.16.3.5.
- Crawford, M., Tilbury, C., Creed, P., and Buys, N. (2011). 'The role of carers and caseworkers in the school-to-work transition of young people in care', *Australian Social Work*, Vol. 64, No. 4, pp.459–474, https://doi.org/10.1080/0312407x.2010.537353.
- Creed, P., Tilbury C., Buys N., and Crawford, M. (2011). 'The career aspirations and action behaviours of Australian adolescents in out-of-home care', *Children and Youth Services Review*, Vol. 33, No. 9, pp.1720–1729, https://doi.org/10.1016/j.childyouth.2011.04.033.
- Cuervo, H., Chesters, J., and Aberdeen, L. (2019). 'Post-school aspirations in regional Australia: An examination of the role of cultural and social capital', *Australian Educational Research*, Vol. 46, pp.843–861, https://doi.org/10.1007/s13384-019-00305-7.
- Daly, F. (2012). 'What do young people need when they leave care? Views of care leavers and aftercare workers in North Dublin', *Child Care in Practice*, Vol. 18, No. 4, pp.309–324, https://doi.org/10.1080/13575279.2012.713852.

- Dansey, D., Shbero, D., and John, M. (2019). 'Keeping secrets: How children in foster care manage stigma', *Adoption and Fostering*, Vol. 43, No. 1, pp.35–45, https://doi.org/10.1177/0308575918823436.
- Darmody, M., McMahon, L., Banks, J., and Gilligan, R. (2013). *Education of children in care in Ireland: An exploratory study*, Dublin, Ombudsman for Children's Office, www.oco.ie/app/uploads/2013/05/11873\_Education\_Care\_SP1.pdf.
- Davey, C., Elsby, A., Erskine, C., Hill-Newell, M., Monk, L., Palmer, L., Smith, R., Whitley, J., Williams, M., Stanworth, N., and Baker, C. (2023). *Evaluation of the care leavers social impact bond (SIB) programme: Final evaluation report*, Department of Education.
- De La Fosse, L., Parsons, S., and Kovshoff, H. (2023). 'They are always in the top of our mind: Designated teachers' views on supporting care experienced children in England', *Children and Society*, Vol. 37, No. 6, pp.1897–1914, https://doi.org/10.1111/chso.12737.
- Denny, E. (2020). 'Student views on transition to higher education in Ireland: Challenges, impacts and suggestions', *Higher Education Quarterly*, Vol. 75, No. 1, pp.113–145, https://doi.org/10.1111/hequ.12273.
- Department of Education (2024). *Funding the future: An annual options paper on reducing the cost of education*, Dublin: Department of Education.
- Department of Education (2018). The designated teacher for looked-after and previously looked-after children: Statutory guidance on their roles and responsibilities, London: Department of Education.
- Department of Further and Higher Education, Research, Innovation and Science (2025). 'Student accommodation assistance – frequently asked questions for students', www.gov.ie/en/publication/84b27-student-accommodation-assistancefrequently-asked-questions-for-students.
- Devine, D., and Cockburn, T. (2018). 'Theorizing children's social citizenship: New welfare states and inter-generational justice', *Sage Journals*, Vol. 25, No. 2, pp.142–157, https://doi.org/10.1177/0907568218759787.
- Dixon, J., Ward, J., and Stein, M. (2018). Brighter futures for care leavers: A consultation on outcomes and aftercare for young people leaving care in Ireland, Focus Ireland, www.focusireland.ie/wp-content/uploads/2021/09/Dixon-et-al-2019-Brighter-Futures-for-Careleavers-Report-Full-PDF.pdf.
- Downes, P. (2022). 'Opening Statement to the Oireachtas Joint Committee on Education, Further and Higher Education, Research, Innovation and Science: *Mental Health Supports* in Schools and Tertiary Education', https://data.oireachtas.ie/ie/oireachtas/committee/dail/33/joint\_committee\_on\_ education\_further\_and\_higher\_education\_research\_innovation\_and\_science/sub missions/2022/2022-11-08\_opening-statement-dr-paul-downes-head-ofeducational-disadvantage-centre-dcu\_en.pdf.
- Duncan, G.J., and Murnane, R.J. (2011). Whither Opportunity?: Rising Inequality, Schools, and Children's Life Chances, Russell Sage Foundation.

- Edgerton, J.D., Roberts, L.W., and von Below, S. (2012). 'Education and Quality of Life', Handbook of Social Indicators and Quality of Life Research, Eds: Kenneth C. Land, Alex C. Michalos, M. Joseph Sirgy, Springer, pp.265–296, https://doi.org/10.1007/978-94-007-2421-1\_12.
- Ellis, K., and Johnston, C. (2022). 'Resilience, higher education and widening participation: Generating change for care experienced students', *Higher Education*, Vol. 87, pp.1–16, https://doi.org/10.1007/s10734-022-00985-8.
- Ellis, K., and Johnston, C. (2020). *Pathways to University The journey through care: Findings Report Two*, https://doi.org/10.15131/shef.data.13247639.
- EPIC (2025). *Take care: The school experiences of children and young people who have been in care*, Dublin: EPIC (Empowering People in Care).
- Erskine, S., and Harmon, C. (2023). *Eurostudent Survey VII*. Dublin: Higher Education Authority (HEA), https://hea.ie/assets/uploads/2021/02/HEA-Eurostudent-Survey-Report-FINAL.pdf.
- Evans, R., Brown, R., Rees, G., and Smith, P. (2017). 'Systematic review of educational interventions for looked-after children and young people: Recommendations for intervention development and evaluation', *British Educational Research Journal*, Vol. 43, No. 1, pp.68–94, https://doi.org/10.1002/berj.3252.
- Faas, D., Darmody, M., and Foley, H. (2019). 'Post-school transitions in Ireland: A case study of Russian-speaking students', *Comparative Education*, Vol. 55, No. 1, pp.1–21, https://doi.org/10.1080/03050068.2019.1619329.
- Fenech, C. (2025). 'Well-being of students enrolled in the field of education: A predictor for student drop-out?', EUROSTUDENT 8 Intelligence Brief.
- Flynn, R.J., Marquis, R.A., Paquet, M., and Peeke, L.M. (2011). *Effects of tutoring by foster* parents on foster children's academic skills in reading and math: A randomised effectiveness trial, Final report of the RESPs for Kids in Care project, Ottawa, Centre for Research on Educational and Community Services.
- Flynn, R.J., Marquis, R.A., Paquet, M., Peeke, L.M., and Aubry, T.D. (2012). 'Effects of individual direct-instruction tutoring on foster children's academic skills: A randomised trial', *Child and Youth Services Review*, Vol. 34, pp.1183–1189, https://doi.org/10.1016/j.childyouth.2012.01.036.
- Frederick, J., Spratt, T., and Devaney, J. (2023). 'Supportive relationships with trusted adults for children and young people who have experienced adversities: Implications for social work service provision, *The British Journal of Social Work*, Vol. 53, No. 6, pp.3129–3145, https://doi.org/10.1093/bjsw/bcad107.
- Frostick, C., Phillips, G., Renton, A., and Moore, D. (2015). 'The educational and employment aspirations of adolescents from areas of high deprivation in London', *Journal of Youth and Adolescence*, Vol. 45, No. 6, pp.1126–1140, https://doi.org/10.1007/s10964-015-0347-4.
- Furlong, A., and Biggart, A. (1999). 'Framing "choices": A longitudinal study of occupational aspirations among 13- to 16-year-olds', *Journal of Education and Work*, Vol. 12, No. 1, pp.21–35, https://doi.org/10.1080/1363908990120102.

- Gale, T., Sellar, S., Parker, S., Hattam, R., Comber, B., Tranter, D., Bills, D., and Sellar, S. (2010). Interventions early in school as a means to improve higher education outcomes for disadvantaged students: A design and evaluation matrix for university outreach in schools, South Australia: Department of Education, Employment and Workplace Relations.
- Gazeley, L., and Hinton-Smith, T. (2018). 'The "success" of looked after children in higher education in England: Near peer coaching, "small steps" and future thinking', *Higher Education Research and Development*, Vol. 37, No. 5, pp.953–965, https://doi.org/10.1080/07294360.2018.1467384.
- Gillborn, D., and Mirza, H.S. (2000). *Educational inequality: Mapping race, class and gender: A synthesis of research evidence,* Office for Standards in Education, https://dera.ioe.ac.uk/id/eprint/4428.
- Gilligan, R., Brady, E., and Cullen, L. (2022). One more adversity: The lived experience of care leavers in Ireland during the Covid-19 pandemic, Dublin: School of Social Work & Social Policy, Trinity College Dublin.
- Gilligan, R. (2018). 'Age limits and eligibility conditions for care, extended care and leaving care support for young people in care and care leavers: The case for cross-national analysis', *Scottish Journal of Residential Child Care*, Vol. 17, No. 2.
- Gilligan, R. (2007). 'Adversity, resilience and the educational progress of young people in public care', *Emotional and Behavioural Difficulties*, Vol. 12, No. 2, pp.135–145, https://doi.org/10.1080/13632750701315631.
- Glynn, N., and Mayock, P. (2018). "I've changed so much within a year": Care leavers' perspectives on the aftercare planning process', *Child Care in Practice*, Vol. 25, No. 1, https://doi.org/10.1080/13575279.2018.1521378.
- Gorard, S., See, B.H., and Davies, P. (2012). *The impact of attitudes and aspirations on educational attainment and participation*, York, Joseph Rowntree Foundation.
- Government of Ireland (2022). National Access Plan: A Strategic Action Plan for Equity of Access, Participation and Success in Higher Education 2022–2028, Dublin: Government of Ireland.
- Government of Ireland (2019). *Wellbeing Policy Statement and Framework for Practice* 2018–2023, Dublin: Government of Ireland.
- Green, R., Collingwood, A., and Ross, A. (2010). *Characteristics of bullying victims in schools*, National Centre for Social Research.
- Greenaway, R., and Treton, U. (2017). 'Awakening aspirations of primary school students: Where will your dreams take you?', *International Journal of Higher Education*, Vol. 6, No. 3, https://doi.org/10.5430/ijhe.v6n3p116.
- Gypen, L., Vanderfaeillie, J., De Maeyer, S., Belenger, L., and Van Holen, F. (2017). 'Outcomes of children who grew up in foster care: Systematic-review', *Children and Youth Services Review*, Vol. 76, pp.74–83, https://doi.org/10.1016/j.childyouth.2017.02.035.
- Haddow, S., Taylor, E.P., and Schwannauer, M. (2021). 'Positive peer relationships, coping and resilience in young people in alternative care: A systematic review', *Children and Youth Services Review*, https://doi.org/10.1016/j.childyouth.2020.105861.

- Hammond, S.P., Young, J., and Duddy, C. (2020). 'Life story work for children and young people with care experience: A scoping review', *Developmental Child Welfare*, Vol. 2, No. 4, pp.293–315, https://doi.org/10.1177/2516103220985872.
- Harder, A.T., Mann-Feder, V., Oterholm, I., and Refaeli, T. (2020). 'Supporting transitions to adulthood for youth leaving care: Consensus based principles', *Children and Youth Services Review*, Vol. 116, 105260, https://doi.org/10.1016/j.childyouth.2020.105260.
- Harrison, N. (2020). 'Patterns of participation in higher education for care-experienced England: Why has there not been students in more progress?' Studies Higher in Education, Vol. 45, No. 9, pp.1986-2000, https://doi.org/10.1080/03075079.2019.1582014.
- Harrison, N. (2017). *Moving on up: Pathways of care leavers and care-experienced students into and through higher education*, Department of Education and Childhood University of the West of England.
- Harvey, A., McNamara, P., Andrewartha, L., and Luckman, M. (2015). *Out of care, into university: Raising higher education access and achievement of care leavers,* National Centre for Student Equity in Higher Education (NCSEHE), Curtin University.
- Hauschildt, K., Gwosć, C., Schirmer, H., Mandl, S., and Menz, C. (2024). 'Social and economic conditions of student life in Europe', EUROSTUDENT 8 Synopsis of Indicators 2021–2024, German Centre for Higher Education Research and Science Studies.
- Healy, K. (2019). 'Adolescents' views on the provision of guidance counselling in an Irish post primary school', University of Limerick. Thesis. https://hdl.handle.net/10344/8386.
- Hearne, L., and Galvin, J. (2014). 'The role of the regular teacher in a whole school approach to guidance counselling in Ireland', *British Journal of Guidance & Counselling*, Vol. 43, No. 2, pp.229–240, https://doi.org/10.1080/03069885.2014.952622.
- Henderson, M., Scourfield, J., Cheung, S. Y., and Sharland, E. (2016). 'Predictors of social service contact among teenagers in England', *The British Journal of Social Work*, Vol. 46, No. 6, pp.1485–1501, https://doi.org/10.1093/bjsw/bcv081.
- Herbers, J.E., Reynolds, A.J., and Chen, C.C. (2013). 'School mobility and developmental outcomes in young adulthood', *Development and Psychopathology*, Vol. 25, No. 2, pp.501–515, https://doi.org/10.1017/S0954579412001204.
- Higher Education Authority (2022a). Widening participation for equity of access, https://hea.ie/statistics/data-for-download-and-visualisations/students/wideningparticipation-for-equity-of-access.
- Higher Education Authority (2022b). National access plan: A strategic action plan for equity of access, participation and success in higher education 2022–2028, Dublin: HEA, https://hea.ie/assets/uploads/2024/07/National-Access-Plan-2022-2028-FINAL.pdf.
- Higher Education Authority (2024a). *Exploring student progression in higher education*, Dublin: HEA.
- Higher Education Authority (2024b). *Report on student accommodation: demand and supply*, Dublin: HEA.

- Johnston, O., Wildy, H., and Shand, J. (2021). "Believe in me, and I will too": A study of how teachers' expectations instilled confidence in Grade 10 students', *Social Psychology of Education*, Vol. 24, pp.1535–1556, https://doi.org/10.1007/s11218-021-09668-1.
- Jones, C.I. (2002). 'Sources of US economic growth in a world of ideas', *The American Economic Review*, Vol. 92, No. 1, pp.220–239, https://web.stanford.edu/~chadj/ SourcesAER2002.pdf.
- Khattab, N. (2014). 'How and when do educational aspirations, expectations and achievement align?' *Sociological Research Online*, Vol. 19, No. 4, pp.61–73, https://doi.org/10.5153/sro.3508.
- Klein, M., Sosu, E.M., and Dare, S. (2022). 'School absenteeism and academic achievement: Does the reason for absence matter?' AERA Open, 8, https://doi.org/10.1177/23328584211071115.
- Lewis, E., Williams, E., Lewis, P., and Allison, D. (2015). 'Success4Life: An aspirational programme for looked after children', *Widening Participation and Lifelong Learning*, Vol. 17, No. 4, https://doi.org/10.5456/WPLL.17.4.116.
- Liebenberg, L., Ungar, M., and Ikeda, J. (2015). 'Neo-liberalism and responsibilisation in the discourse of social service workers', *British Journal of Social Work*, Vol. 45, No. 3, pp.1006–1021, https://doi.org/10.1093/bjsw/bct172.
- Lohmeyer, B.A., McGregor, J.R., Crittenden, Z., and Hartung, C. (2024). 'Mentoring for care-experienced young people: A rapid review of program design', *Children and Youth Services Review*, Vol. 156, 107350, https://doi.org/10.1016/j.childyouth.2023.107350.
- López López, M., Martínez-Jothar, G., ten Brummelaar, M., Parra, L.A., San Román Sobrino, B., and Mallon, G.P. (2024). "They told me that you can be with whomever you want, be who you are": Perceptions of LGBTQ+ youth in residential care regarding the social support provided by child welfare professionals', *Children and Youth Services Review*, Vol. 159, 107498, https://doi.org/10.1016/j.childyouth.2024.107498.
- Mannay, D., Staples, E., Hallet, S., Roberts, L., Rees, A., Evans, R., and Andrews, D. (2017a). 'Understanding the educational experiences and opinions, attainment, achievement and aspirations of looked after children in Wales', *Social Research*, Number: 62/2015, Cardiff: Welsh Government.
- Mannay, D., Evans, R., Staples, E., Hallett, S., Roberts, L., Rees, A., and Andrews, D. (2017b). 'The consequences of being labelled "looked-after": Exploring the educational experiences of looked-after children and young people in Wales', *British Educational Research Journal*, Vol. 43, No. 4, pp.683–699, https://doi.org/10.1002/berj.3283.
- McCoy, S., Smyth, E., Watson, D., and Darmody, M. (2014). *Leaving school in Ireland: A longitudinal study of post school transitions*, ESRI Research Series No. 36, Dublin: ESRI, www.esri.ie/system/files/publications/RS36.pdf.
- McCoy, S., Smyth, E., Darmody, M., and Dunne, A. (2006). *Guidance for all? Guidance provision in second-level schools*, Dublin: The Liffey Press.

- McElvaney, R., Tatlow-Golden, M., Webb, R., Lawlor, E., and Merriman, B. (2013). Someone to care: The mental needs of children and young people with experience of the care and youth justice systems, Dublin: Children's Mental Health Coalition (CMHC).
- McNamara, D.A. (2020). 'Developing a right of aftercare for young people transitioning out of state care', *Irish Journal of Family Law 39*, https://ssrn.com/abstract=3595874.
- McNamara, P., Harvey, A., and Andrewartha, L. (2017). 'Passports out of poverty: Raising access to higher education for care leavers in Australia', *Children and Youth Services Review*, https://doi.org/10.1016/j.childyouth.2017.07.015.
- McSherry, D., Fargas Malet, M., and Weatherall, K. (2013). *Comparing long-term placements for young children in care: The Care Pathways and Outcomes Study, Northern Ireland*, British Association for Adoption and Fostering.
- Mercieca, D., Mercieca, D.P., and Randall, L. (2021). "What about me?" Stories of the educational experiences of care-experienced children and young people in a Scottish local authority', *Adoption & Fostering*, Vol. 45, No. 2, pp.173–190, https://doi.org/10.1177/03085759211011736.
- Moran, L., McGregor, C., and Devaney, C. (2017). *Outcomes for permanence and stability for children in long-term care*, The UNESCO Child and Family Research Centre, the National University of Ireland, Galway.
- Moran, L., Devaney, C., McGregor, C., and Reddy, J. (2016). *Scoping review of international and Irish literature on outcomes for permanence and stability for children in care*, The UNESCO Child and Family Research Centre, The National University of Ireland, Galway.
- Mølholt, A.K., Bengtsson, T.T., and Frederiksen, M. (2022). 'Navigating educational success: Modes of expectation among care-experienced young people', *British Educational Research Journal*, Vol. 50, No. 2, pp.513–528, https://doi.org/10.1002/berj.3908.
- Muir, S., and Hand, K. (2018). Beyond 18: The longitudinal study on leaving care. Wave 1 Research Report: Transition planning and preparation, Commonwealth of Australia.
- Mullan, P. (2018). *Outcomes of young people leaving care in Ireland*, PhD thesis, Maynooth University.
- Mullan, P. (2022). *Outcomes for young people leaving care in Ireland*, thesis, Maynooth University.
- Munford, R. (2022). 'Children and young people in the care system: Relational practice in working with transitions and challenges', *Australian Social Work*, Vol. 75, No. 1, https://doi.org/10.1080/0312407X.2021.1989003.
- Munro, E.R., Lushey, C., Ward, H., and the National Care Advisory Service (2011). *Evaluation of the Right2BCared4 Pilots: Final report,* www.gov.uk/government/ uploads/system/uploads/attachment\_data/file/182483/DFE-RR106.pdf.
- Naeem, M., Ozuem, W., Howell, K., and Ranfagni, S. (2023). 'A step-by-step process of thematic analysis to develop a conceptual model in qualitative research', *International Journal of Qualitative Methods*, Vol. 22, pp.1–18, https://doi.org/10.1177/16094069231205789.

- Nelis, S., Gilleece, L., Fitzgerald, C., and Cosgrove, J. (2021). Beyond achievement: Home, school and wellbeing findings from PISA 2018 for students in DEIS and non-DEIS schools, Dublin: ERC. www.erc.ie/wp-content/uploads/2021/06/ FINAL Web version ERC-PISA-DEIS-Report-II May-2021.pdf.
- NICE (2021). Looked-after children and young people, National Institute for Health and Care Excellence, www.nice.org.uk/guidance/ng205.
- O'Donnell, R., Hatzikiriakidis, K., Mendes, P., Savaglio, M., Green, R., Kerridge, G., Currie, G., and Skouteris, H. (2020). 'The impact of transition interventions for young people leaving care: A review of the Australian evidence', *International Journal of Adolescence and Youth*, Vol. 25, No. 1, pp.1076–1088, https://doi.org/10.1080/02673843.2020.1842216.
- Office for Students (2021). 'Consistency needed: Care experienced students and higher education', *Insight 8*, www.officeforstudents.org.uk/media/5919/insight-8-april-2021-finalforweb.pdf.
- Ogunshakin, O., Hodgson, E., Ndejembi C., and Coulter, A. (2021). *Care experienced young people's views on national standards for unregulated provision*, research report, Department of Education.
- O'Higgins, A., Sebba, J., and Luke, M. (2015). What is the relationship between being in care and the educational outcomes of children? An international systematic review, Rees Centre, University of Oxford, www.education.ox.ac.uk/wpcontent/uploads/2019/06/What-is-the-Relationship-Between-Being-in-Care-andthe-Educational-Outcomes-of-Children-An-International-Systematic-Review.pdf.
- O'Neill, L., Harrison, N., Fowler, N., and Connelly, G. (2019). 'Being a student with care experience is very daunting': Findings from a survey of care experienced students in Scottish colleges and university, Centre for Excellence for Children's Care and Protection, https://strathprints.strath.ac.uk/68655/1/ ONeill\_etal\_CELCIS\_2019\_Being\_a\_student\_with\_care\_experience\_is\_very\_daunt ing\_findings\_from\_a\_survey.pdf.
- Osterman, K.F. (2000). 'Students' need for belonging in the school community', *Review of Educational Research*, Vol. 70, pp.323–367.
- Ó Súilleabháin, F., Brennan, R., Dorney, L., and McGovern, S. (2024). *Students with Care Experience and Higher Education*, Dublin: Technological University Dublin.
- Palmer, A., Norris, M., and Kelleher, J. (2022). 'Accelerated adulthood, extended adolescence and the care cliff: Supporting care leavers' transition from care to independent living', *Child and Family Social Work*, Vol. 27, No. 4, pp.1–12, https://doi.org/10.1111/cfs.12922.
- Parsons, S., Fitzsimons, E., and Schoon, I. (2023). 'Intergenerational transmission of educational disadvantage: Education progression of children of care leavers compared to a general population sample', *British Educational Research Journal*, https://doi.org/10.1002/berj.3872.
- Parsons, S., Schoon, I., and Fitzsimons, E. (2024). Long-term outcomes for care leavers who became parents and experiences of their children. Evidence on the intergenerational transmission of disadvantage in two British cohort studies, Research Report, UCL, Centre of Longitudinal Studies.

- Paulsen, V., and Berg, B. (2016). 'Social support and interdependency in transition to adulthood from child welfare services', *Child and Youth Services Review*, Vol. 68, pp.125–131, https://doi.org/10.1016/j.childyouth.2016.07.006.
- Pinkney, S., and Walker, G. (2020). "It was me, but it was them that helped me": Exploring the issues for care experienced young people within higher education', *Children and Youth Services Review*, Vol. 108, 104576, https://doi.org/10.1016/j.childyouth.2019.104576.
- Powell, F., Scanlon, M., and Jenkinson, H. (2020). *Widening Participation in Irish Higher Education: Report 2020*, Cork: University College Cork, ISBN: 978-1-5272-5692-7.
- Purtell, J., Mendes, P., and Baidawi, S. (with Inder, B.) (2016). *Evaluation of the Berry Street Stand by Me Program: Wraparound support during the transition from out-ofhome care,* Monash University, Melbourne.
- Raffe, D. (2011). Policy borrowing or policy learning? How (not) to improve education systems. Briefing based on a presentation to a Parliamentary Research Enquiry in the Houses of Parliament, London, on 28 June 2011, organised by Cambridge Assessment, www.ces.ed.ac.uk/PDF%20Files/Brief057.pdf.
- Rak, L., and Fuller, E. (2017). 'Live, Learn, Grow: Supporting young people with a care experience through higher education', *Student Success*, Vol. 8, No. 2, pp.95–99, https://doi.org/10.5204/ssj.v8i2.385.
- Robinson, D., and Salvestrini, V. (2020). *The impact of interventions for widening access to higher education: A review of the evidence*, Education Policy Institute, https://epi.org.uk/wp-content/uploads/2020/03/Widening\_participationreview\_EPI-TASO\_2020-1.pdf.
- Roesch-Marsh, A., and Emond, R. (2020). 'Care experience and friendship: Theory and international evidence to improve practice and future research', *The British Journal of Social Work*, Vol. 51, No. 1, pp.132–149, https://doi.org/10.1093/bjsw/bcaa131.
- Sanders, M., Briggs, E., Clayton, V., Copperthwaite, K., Hennessey, A., Eden, A., and Kozman, E. (2021). Springboard. Supporting care experienced young people with their next steps in further and higher education. What Works for Children's Social Care (WWCSC).
- Schoon, I., and Eccles, J.S. (2014). Gender Differences in Aspirations and Attainment: A Life Course Perspective, Cambridge, Cambridge University Press.
- Schwartz-Mette R.A., Shankman J., Dueweke A.R., Borowski S., and Rose A.J. (2020). 'Relations of friendship experiences with depressive symptoms and loneliness in childhood and adolescence: A meta-analytic review', *Psychological Bulletin*, Vol. 146, pp.664–700, https://doi.org/10.1037/bul0000239.
- Sebba, J., and Luke, N. (2019). 'The educational progress and outcomes of children in care: Editorial', Oxford Review of Education, Vol. 45, No. 4, pp.435–442, https://doi.org/10.1080/03054985.2019.1622281.
- Sebba, J., Berridge, D., Luke, N., Fletcher, J., Bell, K., Strand, S., Thomas, S., Sinclair, I., and O'Higgins, A. (2015). *The educational progress of looked after children in England: Linking care and educational data*, Rees Centre/University of Bristol.
- Sellar, S., and Gale, T. (2011). 'Mobility, aspiration, voice: A new structure of feeling for student equity in higher education', *Critical Studies in Education*, Vol. 52, No. 2, pp.115–134, https://doi.org/10.1080/17508487.2011.572826.

- Smyth, E., and Hannan, D.F. (2000). 'Education and inequality', *Bust to Boom? The Irish Experience of Growth and Inequality*, B. Nolan, P.J. O'Connell, and C.T. Whelan (Eds.), pp.109–126, Institute of Public Administration.
- Sprecher, E.A., Tuitt, I., Hill, D., Midgley, N., and Sleed, M. (2021). 'No typical care story: How do care-experienced young people and foster carers understand fostering relationships?', *Adoption & Fostering*, Vol. 45, No. 3, pp.248–264, https://doi.org/10.1177/03085759211037156.
- Stein, M. (2012). *Young People Leaving Care: Supporting Pathways to Adulthood*, London; Philadelphia: Jessica Kingsley Publishers.
- Storø, J. (2018). 'To manage on one's own after leaving care? A discussion of the concepts independence versus interdependence', *Nordic Social Work Research*, Vol. 8, No. 1, pp.104–115, https://doi.org/10.1080/2156857x.2018.1463282.
- Suárez-Orozco C., Bang H.J., and Kim H.Y. (2011). 'I felt like my heart was staying behind: Psychological implications of family separations and reunifications for immigrant youth', *Journal of Adolescent Research*, Vol. 26, No. 2, pp.222–257, https://doi.org/10.1177/0743558410376830.
- Taylor, D., Albers, B., Mann, G., Chakraborty, S., Lewis, J., Mendes, P., Macdonald, G., Williams, K., and Shlonsky, A. (2021). Systematic review and meta-analysis of policies, programmes and intervention that improve outcomes for young people leaving the out-of-home care system, What Works for Children's Social Care/Centre for Evidence and Implementation, Technical Report, https://doi.org/10.13140/RG.2.2.24525.56809.
- The Smith Family (2022). *Pathways, engagement and transitions: Initial post-school transitions among young people experiencing disadvantage,* Melbourne: The Smith Family.
- Thomas, J., McGinty, S., te Riele, K., and Wilson, K. (2017). 'Distance travelled: Outcomes and evidence in flexible learning options', *The Australian Educational Researcher*, Vol. 44, 43224, pp.443–460, https://doi.org/10.1007/s13384-017-0239-6.
- Thornton, M., Darmody, M., and McCoy, S. (2013). 'Persistent absenteeism among Irish primary school pupils', *Educational Review*, Vol. 65, No. 4, pp.488–501, https://doi.org/10.1080/00131911.2013.768599.
- Townsend, I., Berger, E., and Reupert, A. (2020). 'Systematic review of the educational experiences of children in care: Children's perspectives', *Children and Youth Services Review*, Vol. 111, 104835, https://doi.org/10.1016/j.childyouth.2020.104835.
- Tusla (2017). Financial support in aftercare, Dublin: Tusla.
- Tusla (2019). National aftercare policy for alternative care, Dublin: Tusla.
- Tusla (2023). Annual report and financial statements. Dublin: Tusla.
- Tusla (2025). *Monthly service performance and activity report*, February 2025. Dublin: Tusla, www.tusla.ie/uploads/content/Monthly\_Service\_Performance\_and\_Activity\_Rep ort\_Feb\_2025\_\_V1.0.pdf.
- Vinnerljung, B., and Sallnäs, M. (2008). 'Into adulthood: a follow-up study of 718 young people who were placed in out-of-home care during their teens', *Child and Family Social Work*, Vol. 13, No. 2, pp.144–155, https://doi.org/10.1111/j.1365-2206.2007.00527.x.

Walsh, P., McHugh, M., Blunden, H., and Katz, I. (2018). *Pathways of care longitudinal study, Research Report No. 6, Literature Review: Factors influencing the outcomes of children and young people in out-of-home care,* New South Wales Department of Family and Community Services.

West Yorkshire (2024). Go Higher West Yorkshire annual report 2023–2024, West Yorkshire.

- Williams, A., Edwards, V., Doherty, E., Allnatt, G., Bayfield, H., Lyttleton-Smith, J., and Warner, N. (2020). *Care-experienced young people and higher education*, Cardiff University.
- Xu, W., and Zammit, K. (2020). 'Applying thematic analysis to education: A hybrid approach to interpreting data in practitioner research, *International Journal of Qualitative Methods*, Vol. 19, https://doi.org/10.1177/1609406920918810 (original work published 2020).

# CARE TO COLLEGE: A STUDY ON CARE-EXPERIENCED STUDENTS IN HIGHER EDUCATION

## **Interview schedule**

### Care placements and settling in at school

- Did you experience care placement changes when in school? How many care placements did you have? If yes, to what extent do you think it influenced your settling in at school?
- How often did you move schools? How did you find this? What could have been done to help?
- If you could cast your mind back to when you began secondary school (or when you were moved to your final secondary school) did you have any settling in difficulties? What helped you settle into school (support from carers, support from teachers, any other)?

#### Social experiences at school

- Can you tell me about your time in secondary school (experience of and attitudes towards education, schoolwork, the school system, teachers)?
- What did you like/dislike about school?
- Did you ever get into trouble at school?
- Did you ever have any problems with attendance?
- How did you get on with your teachers? Did they know you were in care?
   Did you want them to know?
- Were there any particular teachers or other professionals that you felt easy to talk to? How did it shape your experiences at school? If you had to change schools, what impact did this have?
- Do you feel that teachers provided you with additional academic help, emotional support, encouragement, and promoted your self-belief?
- How easy was it for you to make friends at school? How did this shape your experiences at school? If you had to change schools, what impact did this have?
- Did you feel your school understood what 'being in care' meant and supported you?
- Did you ever have any negative experiences with your peers or teachers at school as a result of being in care (peers and teacher perception, stigma)?

### Academic experiences at school

- When in school, what programme did you follow (Leaving Certificate, LCA, LCVP)?
- What did you like most/least about the programme/subjects at school (subjects available, favourite subjects)?
- How did you do in your state exams (Junior Certificate, Leaving Certificate)?
- Was doing well in your exams important to you? Why? Do you feel you received sufficient help at school?
- Did you have a special educational need in school, or did you ever receive any extra supports for your work in school? What did you find most helpful? Why?
- Do you think that your care experience impacted on how well you did at school (emotional issues, care placements, etc.)?
- Do you think that your school(s) and teachers were supportive in meeting your learning and emotional needs? What could be done differently?
- Did you ever receive additional help from your classmates with your school work?

## Decision-making and progression

- While you were in school, did you get any advice on what you might do after school? Who gave you this advice? What kinds of things were you told (career guidance, options available)? Was this advice helpful?
- Do you think that your experiences in care influenced your goals and aspirations regarding what to do after you finished school? How?
- Thinking back, what do you think was most influential in deciding what to do after you finished school (influence of particular teachers, peers, how well you did at school)?
- What about care professionals and their support? Foster families?

## Leaving school

- What did you do when you finished school? (Details on the nature of the job or jobs, courses, etc.)
- [If entry into labour market] What kinds of employment did you work in (employment, unemployed)?
- **[If PLC/training/apprenticeship]** What type of course did you do? Why did you choose to go on to further education (mobility, expectations, job/career choices, knowledge available)?
- How much support and information did you get in school regarding entry to higher education (HE)? Was this enough?

- [If HE] Why did you choose to go on to higher education?
- [If not progressing to HE], what were the main reasons for this (encouragement from teachers, carers)?
- [If dropping out of HE], what were the main reasons for this?
- Did the necessity to be in full-time education to receive an aftercare package have any influence on you going to university/college?
- **[If HE]**, who helped you to decide which course or higher education institution to choose? Who helped you to fill out the CAO form?
- What were the main challenges entering HE (geographic location, support network, curriculum, teaching and learning, accommodation, finances)?

## **General reflections**

- What advice would you give to children in care at different stages of education?
- How could schools and teachers best support children with care experiences?
- Do you think there should be specific supports available in schools for children and young people in care?
- How do you think your experiences in care have shaped the way you see yourself and your personal growth?

# CARE TO COLLEGE: A STUDY ON CARE-EXPERIENCED STUDENTS IN HIGHER EDUCATION

### Interview consent form

#### *Purpose of the study*

You are being asked to take part in a study on **Care-Experienced Students in Higher Education**, conducted by the Economic and Social Research Institute (ESRI) and funded by EPIC. The study involves online focus group interviews with young people with care experience who have progressed to higher education (HE), whose participation in HE has been interrupted, or who have not considered progression to HE.

### What we are asking of you

We are asking you to take part in an interview on your experiences at school and how you decided what to do after finishing school. This interview will take up to 60 minutes and will be facilitated by EPIC. The interview will be conducted as a group online (by Microsoft Teams) or, if you prefer, by telephone. We would like to record the interview so that we can analyse the transcript. The research report on postschool decision-making of care-experienced young people will not name or identify any participants. The purpose is to learn from current experiences of careexperienced youth who have accessed HE or have not considered this option for various reasons, to improve access to HE for young people like you.

#### What you are agreeing to

- The interview will be recorded and transcribed. We will not ask you to share any personal information as part of the interview.
- The transcript of the interview will not be shared with anyone outside the research team.
- What you share with us is confidential unless you tell us something that makes us concerned about your wellbeing.
- The recordings will be destroyed once the transcriptions are finalised. The anonymised transcripts will be used for research purposes only and deleted after five years.
- Participation is entirely voluntary. You have the right to stop the interview and withdraw from the study.

#### Data protection

This study is carried out in accordance with Data Protection legislation. You can find detailed information about privacy and data protection for research conducted by the ESRI at www.esri.ie/esri-privacy-notice-for-research.

If you agree to participate in the study, please sign here:

Signature:



empowering people in care

7 Red Cow Lane, Smithfield, Dublin 7 Phone: 01 8727661 Web: www.epiconline.ie Email: info@epiconline.ie

😢 @epiconline.ie	
------------------	--

- @epicempoweringpeopleincare
- in @epicireland
- **F** EPIC Empowering People in Care
- EPIC Empowering People in Care



Whitaker Square, Sir John Rogerson's Quay, Dublin 2 **Phone:** +35318632000 **Web:** www.esri.ie **Email:** admin@esri.ie

@esri.ie
 Economic and Social Research Institute
 @esridublin